UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

FORM 8-K

CURRENT REPORT

PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

May 17, 2024 (Date of earliest event reported)

LABCORP HOLDINGS INC.

(Exact Name of Registrant as Specified in its Charter)

Delaware	1-11353	99-2588107
(State or other jurisdiction of Incorporation)	(Commission File Number)	(I.R.S. Employer Identification No.)
358 South Main Street		
Burlington,	North Carolina	27215
(Address of principal executive offices)		(Zip Code)

(Registrant's telephone number including area code) **336-229-1127**

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

□ Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

□ Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

□ Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

□ Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Securities registered pursuant to Section 12(b) of the Exchange Act.

Title of Each Class	Trading Symbol	Name of exchange on which registered
Common Stock, \$0.10 par value	LH	New York Stock Exchange

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter).

Emerging growth company \Box

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. \Box

EXPLANATORY NOTE

As previously disclosed, on April 25, 2024, Laboratory Corporation of America Holdings, a Delaware corporation ("Labcorp"), announced plans to implement a holding company reorganization (the "Reorganization"). On May 17, 2024, Labcorp implemented the Reorganization pursuant to an Agreement and Plan of Merger (the "Merger Agreement") dated as of May 17, 2024, among Labcorp, Labcorp Holdings Inc., a Delaware corporation ("Labcorp Holdings") and Radiance Merger Sub Inc., a Delaware corporation and direct, wholly owned subsidiary of Labcorp Holdings ("Merger Sub"). Pursuant to the terms of the Merger Agreement, Merger Sub merged with and into Labcorp, with Labcorp continuing as the surviving corporation and a wholly owned subsidiary of Labcorp Holdings (the "Merger"). Following the Merger, Labcorp Holdings became the successor issuer to Labcorp. This Current Report on Form 8-K is being filed for the purpose of establishing Labcorp Holdings as the successor issuer pursuant to Rule 12g-3(a) promulgated under the Securities Exchange Act of 1934, as amended (the "Exchange Act") and to disclose certain related matters. Pursuant to Rule 12g-3(a) promulgated under the Exchange Act, shares of Labcorp Holdings common stock, par value \$0.10 per share ("Labcorp Holdings Common Stock"), issued in connection with the Merger are deemed registered under Section 12(b) of the Exchange Act as the common stock of the successor issuer.

Item 1.01. Entry into a Material Definitive Agreement.

The information set forth above under Explanatory Note is incorporated hereunder by reference.

Adoption of Agreement and Plan of Merger and Consummation of Reorganization

On May 17, 2024, Labcorp completed the Reorganization by implementing the Merger pursuant to the terms of the Merger Agreement. The Merger was completed pursuant to Section 251(g) of the General Corporation Law of the State of Delaware (the "DGCL"), which provides for the formation of a holding company (i.e., Labcorp Holdings) without a vote of the shareholders of the constituent corporation (i.e., Labcorp). At the Effective Time (as defined in the Merger Agreement), (i) the separate existence of Merger Sub ceased and (ii) each share of Labcorp common stock, par value \$0.10 per share ("Labcorp Common Stock"), issued and outstanding immediately prior to the Effective Time was automatically converted into one share of Labcorp Holdings Common Stock, having the same designation, rights, powers, and preferences, and qualifications, limitations, and restrictions as a share of Labcorp Common Stock immediately prior to consummation of the Reorganization. The conversion of stock occurred automatically without an exchange of stock certificates. Accordingly, each shareholder of Labcorp immediately before the Effective Time owned, immediately after the Effective Time, shares of Labcorp Holdings Common Stock in the same amounts and percentages as such shareholder owned in Labcorp immediately before the Effective Time. The Reorganization is intended to be a tax-free transaction, such that Labcorp shareholder should not recognize gain or loss for U.S. federal income tax purposes upon the conversion of their shares of Labcorp Common Stock pursuant to the Reorganization.

Following the consummation of the Reorganization, Labcorp Holdings Common Stock continues to trade on the New York Stock Exchange (the "NYSE") on an uninterrupted basis under the ticker symbol "LH" with a new CUSIP number (504922 105). As a result of the Reorganization, Labcorp Holdings became the successor issuer to Labcorp pursuant to Rule 12g-3(a) promulgated underthe Exchange Act, and as a result, shares of Labcorp Holdings Common Stock are deemed registered under Section 12(b) of the Exchange Act as the common stock of the successor issuer. Labcorp will continue to be the issuer of its senior unsecured notes (which will have the benefit of the Labcorp Holdings guarantee described below), and there will be no change to the CUSIP numbers of such notes.

Immediately following the consummation of the Reorganization, on a consolidated basis, the assets, businesses, and operations of Labcorp Holdings are not materially different than the corresponding assets, business, and operations of Labcorp immediately prior to the consummation of the Reorganization.

The foregoing descriptions of the Merger Agreement and the Reorganization do not constitute complete descriptions of, and are qualified in their entirety by reference to, the full text of the Merger Agreement, which is attached hereto as Exhibit 2.1 and incorporated by reference herein.

Seventeenth Supplemental Indenture and the Labcorp Holdings Guarantee

In connection with the Merger, on May 17, 2024, Labcorp, Labcorp Holdings and U.S. Bank Trust Company, National Association (the "Trustee") entered into a seventeenth supplemental indenture (the "Seventeenth Supplemental Indenture") to the indenture, dated as of November 19, 2010 between Labcorp and the Trustee (the "Indenture"), in order to (i) provide for the full and unconditional

guarantee by Labcorp Holdings of Labcorp's obligations under the Indenture and each series of senior unsecured notes issued and outstanding thereunder (the "Labcorp Holdings Guarantee"), (ii) permit Labcorp Holdings to satisfy Labcorp's reporting obligations under the Indenture for so long as the Labcorp Holdings Guarantee remains in place and Labcorp Holdings' financial statements and other information comply with the requirements of Rule 3-10 of Regulation S-X and the Exchange Act and (iii) make certain other changes to the Indenture consistent with the foregoing.

The foregoing description of the Seventeenth Supplemental Indenture does not constitute a complete description of, and is qualified in its entirety by reference to, the full text of each of the Seventeenth Supplemental Indenture and the Labcorp Holdings Guarantee, which are filed herewith as Exhibits 4.2 and 4.3, respectively, and are each incorporated by reference herein.

Guarantor Joinder Agreement

On May 17, 2024, Labcorp Holdings entered into the Guarantor Joinder Agreement (the "Guarantor Joinder Agreement"), pursuant to which, among other things, Labcorp Holdings provided a full and unconditional guarantee of Labcorp's obligations under the Third Amended and Restated Credit Agreement, dated as of April 30, 2021 (the "Guarantor Joinder Agreement"), by and among Labcorp, as borrower, the lenders from time to time party thereto and Bank of America, N.A., as administrative agent.

The foregoing description of the Guarantor Joinder Agreement does not constitute a complete description of, and is qualified in its entirety by reference to, the full text of the Guarantor Joinder Agreement, which is attached hereto as Exhibit 10.1 and incorporated by reference herein.

Item 2.03 Creation of a Direct Financial Obligation or an Obligation under an Off-Balance Sheet Arrangement of a Registrant.

The information set forth under and/or incorporated by reference into the headings "Guarantor Joinder Agreement" and "Seventeenth Supplemental Indenture and the Labcorp Holdings Guarantee" in Item 1.01 is incorporated hereunder by reference.

Item 3.01 Notice of Delisting or Failure to Satisfy a Continued Listing Rule or Standard; Transfer of Listing.

The information set forth above under Explanatory Note is incorporated hereunder by reference.

In connection with consummation of the Reorganization, Labcorp notified the NYSE that the Merger had been completed. As noted above, Labcorp Holdings Common Stock continues to trade on the NYSE on an uninterrupted basis under the ticker symbol "LH," which was the same symbol formerly used for Labcorp Common Stock, with the new CUSIP number (504922 105). The NYSE is expected to file with the Securities and Exchange Commission (the "SEC") an application on Form 25 to delist Labcorp Common Stock from the NYSE and to deregister Labcorp Common Stock under Section 12(b) of the Exchange Act. Labcorp intends to file with the SEC a certificate on Form 15 requesting that the Labcorp Common Stock be deregistered under the Exchange Act, and that Labcorp's reporting obligations under Section 15(d) of the Exchange Act with respect to Labcorp Common Stock be suspended. Following the Reorganization, Labcorp Holdings and Labcorp will make filings with the SEC separately. Labcorp Holdings will make filings with the SEC under a new CIK (0002022473). Pursuant to the terms of the Seventeenth Supplemental Indenture, and following the execution of the Labcorp Holdings Guarantee on May 17, 2024, Labcorp Holdings' filings with the SEC will satisfy the statutory (under Section 15(d) of the Exchange Act) and contractual reporting requirements with respect to Labcorp's issued and outstanding senior unsecured notes.

The information set forth under and/or incorporated by reference into Items 1.01 and 5.03 is incorporated hereunder by reference.

Item 3.03 Material Modification of Rights of Securityholders.

The information set forth under and/or incorporated by reference into Items 1.01, 3.01 and 5.03 is incorporated hereunder by reference.

Item 5.01 Changes in Control of the Registrant.

The information set forth under and/or incorporated by reference into Items 1.01, 3.01 and 8.01 is incorporated hereunder by reference.

Item 5.02 Departure of Directors or Certain Officers; Election of Directors; Appointment of Certain Officers; Compensatory Arrangements of Certain Officers.

Appointment of Certain Officers of Labcorp Holdings; Election of New Directors of Labcorp Holdings

The directors of Labcorp Holdings and their committee memberships and titles, which are listed below, are the same as the directors of Labcorp immediately prior to consummation of the Reorganization. Mr. Schechter will continue to serve as Chairman of the Board of Directors of Labcorp Holdings.

Directors

Name	Age	Audit Committee	Compensation and Human Capital Committee	Nominating and Corporate Governance Committee	Quality and Compliance Committee
Adam H. Schechter	59				
Kerrii B. Anderson	66	•		•	
Jeffrey A. Davis	60	•			•
D. Gary Gilliland, M.D., Ph.D.	69	•			•
Kirsten M. Kliphouse	57	•			
Garheng Kong, M.D., Ph.D.*	48		•	•	
Peter M. Neupert	69	•		•	
Richelle P. Parham	56		•	•	
Paul B. Rothman, M.D.	65				•
Kathryn E. Wengel	58		•		•

 \bullet = Committee Member

* = Lead Independent Director

From and after the Effective Time, the executive officers of Labcorp Holdings, who are listed below, were all executive officers of Labcorp immediately prior to consummation of the Reorganization.

Executive Officers

Name	Age	Position
Adam H. Schechter	59	President and Chief Executive Officer
Glenn A. Eisenberg	62	Executive Vice President, Chief Financial Officer
Mark S. Schroeder	63	Executive Vice President and President, Diagnostics Laboratories and Chief Operations Officer
Peter J. Wilkinson	53	Senior Vice President, Chief Accounting Officer

Biographical information about Labcorp Holdings' directors and executive officers is included in <u>Labcorp's definitive proxy statement</u> for the 2024 Annual Meeting of Shareholders under "Election of Directors" and "Executive Officers," respectively, and incorporated by reference herein. There are no arrangements or understandings with any person pursuant to which the directors and the executive officers were appointed. There are no family relationships amongst any of the directors or any of the executive officers of Labcorp Holdings.

Information regarding the compensation arrangements of Labcorp Holdings' named executive officers, including Messrs. Schechter, Eisenberg and Schroeder, and regarding related party transactions pursuant to Item 404(a) of Regulation S-K is included in <u>Labcorp's definitive proxy statement</u> for the 2024 Annual Meeting of Shareholders under "Compensation Discussion & Analysis" and "Executive Compensation" and "Corporate Governance— Corporate Governance Policies and Procedures—Related Party Transactions" and each of these sections is incorporated by reference herein. In addition, Mr. Wilkinson will receive an annual base salary of \$404,000 and will be eligible to participate in the Labcorp Holdings' annual cash incentive plan, initially at a target level of 50 percent of his base salary. Subject to future Compensation and Human Capital Committee approval, Mr. Wilkinson is expected to receive an equity grant consisting of a combination of Restricted Stock Units, Stock Options, and Performance Shares with an aggregate value of \$450,000.

Outstanding Equity Plans, Awards and Related Arrangements

In connection with consummation of the Reorganization, on May 17, 2024, (i) Labcorp, Labcorp Holdings and Adam H. Schechter entered into an Assignment and Assumption Agreement (the "Schechter Assignment and Assumption Agreement"), pursuant to which, at the Effective Time, Labcorp transferred and assigned to Labcorp Holdings, and Labcorp Holdings assumed, all of Labcorp's rights and obligations under an Executive Employment Agreement, dated as of June 4, 2019, between Labcorp and Adam H. Schechter, and (ii) Labcorp, Labcorp Holdings and Merger Sub entered into an Assignment and Assumption Agreement (the "Equity Compensation Plans Assignment and Assumption Agreement"), pursuant to which, at the Effective Time, Labcorp assigned (including sponsorship of) to Labcorp Holdings, and Labcorp Holdings assumed (including sponsorship of) from Labcorp, all of Labcorp's rights and obligations under (a) the Laboratory Corporation of America Holdings 2016 Omnibus Incentive Plan and all outstanding awards and award agreements thereunder and (b) the Laboratory Corporation of America Holdings Amended and Restated 2016 Employee Stock Purchase Plan and all outstanding awards, award agreements and rights thereunder.

On May 17, 2024, Labcorp Holdings amended and restated the equity compensation plans assumed pursuant to the Equity Plans Compensation Plans Assignment Agreement — (i) the Labcorp Holdings Inc. Amended and Restated 2016 Omnibus Incentive Plan (the "Labcorp Holdings LTIP") and (ii) the Labcorp Holdings Inc. Amended and Restated 2016 Employee Stock Purchase Plan (the "Labcorp Holdings ESPP"). The share reserve of the Labcorp Holdings LTIP is 9,044,725 shares and the share reserve of the Labcorp Holdings ESPP is 447,462 shares. Outstanding awards granted under Labcorp's equity compensation plans will be satisfied by the share reserves established under the Labcorp Holdings LTIP and the Labcorp Holdings ESPP, as the case may be.

The foregoing descriptions of the Schechter Assignment and Assumption Agreement, the Equity Compensation Plans Assignment and Assumption Agreement, the Labcorp Holdings LTIP and the Labcorp Holdings ESPP do not constitute complete descriptions of, and are qualified in their entirety by reference to, the full text of each of the Schechter Assignment and Assumption Agreement, the Equity Compensation Plans Assignment and Assumption Agreement, the Labcorp Holdings LTIP and the Labcorp Holdings ESPP, which are attached hereto as Exhibits 10.2, 10.3, 10.4 and 10.5, respectively, and each incorporated by reference herein.

Item 5.03 Amendments to Articles of Incorporation or Bylaws; Change in Fiscal Year.

Upon consummation of the Reorganization, the Amended and Restated Certificate of Incorporation of Labcorp Holdings (the "Amended and Restated Certificate of Incorporation") and the Amended and Restated By-laws of Labcorp Holdings (the "Amended and Restated By-laws") are the same as the certificate of incorporation and by-laws of Labcorp in effect immediately prior to consummation of the Reorganization, respectively, other than changes permitted by Section 251(g) of the DGCL. The Amended and Restated Certificate of Incorporation was filed with the Secretary of State of the State of Delaware on May 17, 2024.

In addition, upon consummation of the Reorganization, Labcorp amended and restated its Amended and Restated Certificate of Incorporation (as so amended and restated the "Labcorp Amended and Restated Charter") by filing the Labcorp Amended and Restated Charter as an exhibit to the Certificate of Merger filed with the Secretary of State of the State of Delaware on May 17, 2024 in connection with the Merger (the "Certificate of Merger"), in order to (i) decrease the authorized number of shares of Labcorp Common Stock from 265,000,000 shares to one thousand (1,000) shares, (ii) decrease the authorized number of shares of Labcorp preferred stock having a par value of \$0.10 from 30,000,000 shares to two hundred fifty (250) shares and (ii) add a provision, which is required by Section 251(g) of the DGCL, that provides that any act or transaction by or involving Labcorp, other than the election or removal of directors, that requires for its adoption under the DGCL or the Labcorp Amended and Restated Charter the approval of the stockholders of Labcorp Holdings by the same vote as is required by the DGCL and/or the Labcorp Amended and Restated Charter.

The foregoing descriptions of the Amended and Restated Certificate of Incorporation, the Amended and Restated By-laws, and the Labcorp Amended and Restated Charter do not constitute complete descriptions of, and are qualified in their entirety by reference to, the full text of each of the Amended and Restated Certificate of Incorporation and the Amended and Restated By-laws, and the Labcorp Amended and Restated Charter, respectively, which are attached hereto as Exhibits 3.1, 3.2, and 3.3 respectively, and each incorporated by reference herein.

Item 8.01 Other Events.

The information set forth above under Explanatory Note is incorporated hereunder by reference.

Description of Securities Registered Pursuant to Section 12 of the Exchange Act

The description of Labcorp Holdings' securities registered pursuant to Section 12 of the Exchange Act provided in Exhibit 4.1, which is incorporated by reference herein, modifies and supersedes any prior description of Labcorp's capital stock in any registration statement or report filed with the SEC and will be available for incorporation by reference into certain of Labcorp Holdings' filings with the SEC pursuant to the Securities Act of 1933, as amended, the Exchange Act, and the rules and forms promulgated thereunder.

Forward-Looking Statements

This Current Report on Form 8-K contains forward-looking statements. Some of these forward-looking statements relate to future events and expectations and can be identified by the use of forward-looking words such as "believes," "expects," "may," "will," "should," "seeks," "approximately," "intends," "plans," "estimates," or "anticipates" or the negative of those words or other comparable terminology. Such forward-looking statements speak only as of the time they are made and are subject to various risks and uncertainties and Labcorp Holdings claims the protection afforded by the safe harbor for forward-looking statements contained in the Private Securities Litigation Reform Act of 1995. Forward-looking statements, including, but not limited to, statements regarding Labcorp Holdings' ability to complete the Reorganization, the impacts of the Reorganization, Labcorp Holdings' ability to realize the expected benefits of Reorganization and Labcorp's obligations pursuant to agreements related to the consummation of the Reorganization, are not guarantees of future performance and involve risks and uncertainties that may cause Labcorp Holdings' actual results to differ materially from Labcorp Holdings' expectations discussed in the forward-looking statements. Each of the forward-looking statements is subject to change based on various important factors, many of which are beyond Labcorp Holdings' control, including without limitation: the effect of the announcement of the Reorganization on Labcorp Holdings' business generally, unexpected issues that arise following completion of the Reorganization, market reaction to the announcement, updates on and completion of the Reorganization, and those factors discussed in Labcorp Holdings' reports filed from time to time with the SEC. Except as may be required by applicable law, Labcorp Holdings undertakes no obligation to publicly update or revise any forward-looking statements.

Item 9.01	Financial Statements and Exhibits.
<u>Exhibit</u>	Exhibit Name
Exhibit 2.1	Agreement and Plan of Merger, dated May 17, 2024, by and among Laboratory Corporation of America Holdings, Labcorp Holdings Inc. and Radiance Merger Sub Inc.
Exhibit 3.1	Amended and Restated Certificate of Incorporation of Labcorp Holdings Inc.
Exhibit 3.2	Amended and Restated By-laws of Labcorp Holdings Inc.
Exhibit 3.3	Certificate Merger.
Exhibit 4.1	Description of registrant's securities registered pursuant to Section 12 of the Securities and Exchange Act of 1934, as amended.
Exhibit 4.2	Seventeenth Supplemental Indenture, dated as of May 17, 2024, by and among Laboratory Corporation of America Holdings, as issuer, Labcorp Holdings Inc., as guarantor, and U.S. Bank National Trust Company, Association, as trustee.
Exhibit 4.3	New Holding Company Guarantee, dated May 17, 2024, by Labcorp Holdings Inc.
Exhibit 10.1	Guarantor Joinder Agreement, dated May 17, 2024, by Labcorp Holdings Inc.
Exhibit 10.2	Assignment and Assumption Agreement, dated as of May 17, 2024, by and among Laboratory Corporation of America Holdings, Labcorp Holdings Inc. and Adam H. Schechter.
Exhibit 10.3	Assignment and Assumption Agreement, dated as of May 17, 2024, by and among Laboratory Corporation of America Holdings, Labcorp Holdings Inc. and Radiance Merger Sub Inc.
Exhibit 10.4	Labcorp Holdings Inc. Amended and Restated 2016 Omnibus Incentive Plan.
Exhibit 10.5	Labcorp Holdings Inc. Amended and Restated 2016 Employee Stock Purchase Plan.
Exhibit 104	Cover Page Interactive Data File (embedded within the Inline XBRL document).

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

LABCORP HOLDINGS INC.

By: /s/ Sandra D. van der Vaart

Sandra D. van der Vaart

Executive Vice President, Chief Legal Officer and Secretary

May 17, 2024

AGREEMENT AND PLAN OF MERGER

This AGREEMENT AND PLAN OF MERGER (the "<u>Agreement</u>"), entered into as of May 17, 2024, by and among Laboratory Corporation of America Holdings, a Delaware corporation (the "<u>Company</u>"), Laborp Holdings Inc., a Delaware corporation ("<u>New Parent</u>") and a direct, wholly owned subsidiary of the Company, and Radiance Merger Sub Inc., a Delaware corporation ("<u>Merger Sub</u>") and a direct, wholly owned subsidiary of New Parent. The Company, New Parent and Merger Sub are each referred to herein individually as a "<u>Party</u>" and collectively as the "<u>Parties</u>."

RECITALS

WHEREAS, on the date hereof, the Company has the authority to issue 295,000,000 shares, consisting of: (i) 265,000,000 shares of Common Stock, par value \$0.10 per share (the "<u>Company Common Stock</u>"), of which 84,293,628 shares were issued and outstanding as of April 29, 2024; and (ii) 30,000,000 shares of Preferred Stock, par value \$0.10 per share (the "<u>Company Preferred Stock</u>"), of which no shares are issued and outstanding;

WHEREAS, as of the Effective Time (as defined below), New Parent will have the authority to issue 295,000,000 shares, consisting of: (i) 265,000,000 shares of Common Stock, par value \$0.10 per share (the "<u>New Parent Common Stock</u>"); and (ii) 30,000,000 shares of Preferred Stock, par value \$0.10 per share (the "<u>New Parent Preferred Stock</u>");

WHEREAS, as of the date hereof, Merger Sub has the authority to issue 1,250 shares, consisting of: (i) 1,000 shares of common stock, par value \$0.10 per share (the "<u>Merger Sub Common Stock</u>"), of which 1,000 shares are issued and outstanding on the date hereof and owned by New Parent; and (ii) 250 shares of preferred stock, par value \$0.10 per share ("<u>Merger Sub Preferred Stock</u>"), of which no shares are issued and outstanding;

WHEREAS, as of the Effective Time, the designations, rights, powers and preferences, and the qualifications, limitations and restrictions of the New Parent Common Stock and New Parent Preferred Stock will be the same as those of the Company Common Stock and Company Preferred Stock, respectively;

WHEREAS, the Amended and Restated Certificate of Incorporation of New Parent (the "<u>New Parent Charter</u>") and the By-laws of New Parent (the "<u>New Parent By-laws</u>"), which will be in effect from and after the Effective Time, contain provisions identical to the Amended and Restated Certificate of Incorporation of the Company (the "<u>Company Charter</u>") and the Amended and Restated By-laws of the Company (the "<u>Company By-laws</u>"), in effect as of the date hereof and that will be in effect immediately prior to the Effective Time, respectively (other than as permitted by Section 251(g) of the General Corporation Law of the State of Delaware (the "<u>DGCL</u>"));

WHEREAS, New Parent and Merger Sub are newly formed corporations organized for the sole purpose of participating in the transactions herein contemplated and actions related thereto, own no assets (other than New Parent's ownership of Merger Sub and nominal capital) and have taken no actions other than those necessary or advisable to organize the corporations and to effect the transactions herein contemplated and actions related thereto;

WHEREAS, the Company desires to reorganize into a holding company structure pursuant to Section 251(g) of the DGCL, under which New Parent would become a holding company, by the merger

of Merger Sub with and into the Company, and with each share of Company Common Stock and Company Preferred Stock being converted in the Merger (as defined below) into a share of New Parent Common Stock and New Parent Preferred Stock, respectively;

WHEREAS, at the Effective Time, the Company and New Parent will enter or have entered into an Assignment and Assumption Agreement (as defined below), pursuant to which, among other things, the Company will, at the Effective Time, transfer to New Parent, and New Parent will assume, sponsorship of all of the Company's Equity Plans and the Other Agreements (each as defined below) and all of the Company's rights and obligations thereunder;

WHEREAS, the boards of directors of New Parent and the Company have approved and declared advisable this Agreement and the transactions contemplated hereby, including, without limitation, the Merger;

WHEREAS, the board of directors of Merger Sub has (i) approved and declared advisable this Agreement and the transactions contemplated hereby, including, without limitation, the Merger, (ii) resolved to submit the approval of the adoption of this Agreement and the transactions contemplated hereby, including, without limitation, the Merger, to its sole stockholder, and (iii) resolved to recommend to its sole stockholder that it approve the adoption of this Agreement and the transactions contemplated hereby, including, without limitation, the Merger, to its sole stockholder, and (iii) resolved to recommend to its sole stockholder that it approve the adoption of this Agreement and the transactions contemplated hereby, including, without limitation, the Merger; and

WHEREAS, for U.S. federal income tax purposes, the Parties intend that the Merger shall qualify as a "reorganization" within the meaning of Section 368(a) of the Code, and this Agreement is intended to be and is adopted as a "plan of reorganization" for the Merger for purposes of Sections 354 and 361 of the Code.

NOW, THEREFORE, in consideration of the premises and the covenants and agreements contained in this Agreement, and intending to be legally bound hereby, the Company, New Parent and Merger Sub hereby agree as follows:

- 1. THE MERGER. In accordance with Section 251(g) of the DGCL and subject to, and upon the terms and conditions of, this Agreement, Merger Sub shall be merged with and into the Company (the "<u>Merger</u>"), the separate corporate existence of Merger Sub shall cease, and the Company shall continue as the surviving corporation of the Merger (the "<u>Surviving Corporation</u>"). At the Effective Time, the effects of the Merger shall be as provided in this Agreement and in Section 259 of the DGCL.
- 2. EFFECTIVE TIME. As soon as practicable on or after the date hereof, the Surviving Corporation shall file a certificate of merger executed in accordance with the relevant provisions of the DGCL, (the "<u>Certificate of Merger</u>") in the form attached hereto as <u>Exhibit A</u> with the Secretary of State of the State of Delaware (the "<u>Secretary of State</u>") and shall make all other filings or recordings required under the DGCL to effectuate the Merger. The Merger shall become effective at such time as the Certificate of Merger is duly filed with the Secretary of State or at such later date and time as the Parties shall agree and specify in the Certificate of Merger (the date and time the Merger becomes effective being referred to herein as the "<u>Effective Time</u>").
- 3. SURVIVING CORPORATION CERTIFICATE OF INCORPORATION. From and after the Effective Time, the Company Charter shall be amended and restated in the Merger by filing the Certificate of Merger with the Secretary of the State, and as so amended and restated, shall be

the certificate as set forth below, and as so amended, shall be the certificate of incorporation of the Surviving Corporation (the "Surviving Corporation Charter") until thereafter amended as provided therein or by the DGCL.

4. SURVIVING CORPORATION BY-LAWS. From and after the Effective Time, the Company By-laws shall be amended and restated in the Merger in the form attached hereto as <u>Exhibit B</u> and, as so amended and restated, shall constitute the by-laws of the Surviving Corporation (the "<u>Surviving Corporation By-laws</u>") until thereafter amended as provided therein or by applicable law.

5. DIRECTORS.

- a. <u>Surviving Corporation</u>. The directors of Merger Sub in office immediately prior to the Effective Time shall be the directors of the Surviving Corporation upon the Effective Time and will continue to hold office from the Effective Time until the earlier of their resignation or removal or until their successors are duly elected or appointed and qualified in the manner provided in the Surviving Corporation Charter and Surviving Corporation By-laws, or as otherwise provided by law.
- b. <u>New Parent</u>. Prior to the Effective Time, the Company, as sole stockholder of New Parent, and New Parent, shall take all action necessary to elect as directors of New Parent effective as of the Effective Time the persons who are the directors of the Company immediately prior to the Effective Time, each to hold office from the Effective Time until the earlier of their resignation or removal or until their successors are duly elected or appointed and qualified in the manner provided in the New Parent Charter and New Parent By-laws, or as otherwise provided by law.

6. OFFICERS.

- a. <u>Surviving Corporation</u>. The officers of the Company in office immediately prior to the Effective Time shall be the officers of the Surviving Corporation and will continue to hold office from the Effective Time until the earlier of their resignation or removal or until their successors are duly elected or appointed and qualified in the manner provided in the Surviving Corporation Charter and Surviving Corporation By-laws, or as otherwise provided by law.
- b. <u>New Parent</u>. The officers of New Parent in office immediately prior to the Effective Time shall be the officers of New Parent and will continue to hold office from the Effective Time until the earlier of their resignation or removal or until their successors are duly elected or appointed and qualified in the manner provided in the New Parent Charter and New Parent By-laws, or as otherwise provided by law.
- 7. ADDITIONAL ACTIONS. If, at any time after the Effective Time, the Surviving Corporation shall consider or be advised that any deeds, bills of sale, assignments, assurances or any other actions or things are necessary or desirable to vest, perfect or confirm, of record or otherwise, in the Surviving Corporation its right, title or interest in, to or under any of the rights, properties or assets of either Merger Sub or the Company acquired or to be acquired by the Surviving Corporation as a result of, or in connection with, the Merger or otherwise to carry out this Agreement, the officers and directors of the Surviving Corporation shall be authorized to execute and deliver, in the name and on behalf of each of Merger Sub and the Company, all such deeds, bills of sale, assignments and assurances and to take and do, in the name and on behalf of each of



Merger Sub and the Company or otherwise, all such other actions and things as may be necessary or desirable to vest, perfect or confirm any and all right, title and interest in, to and under such rights, properties or assets in the Surviving Corporation or otherwise to carry out this Agreement.

- **8. CONVERSION OF SECURITIES.** At the Effective Time, by virtue of the Merger and without any action on the part of New Parent, Merger Sub, the Company or any holder of any securities thereof:
 - a. <u>Conversion of Company Common Stock and Company Preferred Stock</u>. Each share of Company Common Stock and Company Preferred Stock (or fraction of any such share) issued and outstanding immediately prior to the Effective Time shall be converted into one validly issued, fully paid and nonassessable share (or equal fraction of a share) of New Parent Common Stock and New Parent Preferred Stock, respectively.
 - b. <u>Conversion of Company Stock Held as Treasury Stock</u>. Each share of Company Common Stock, held in the Company's treasury shall be converted into one validly issued, fully paid and nonassessable share (or equal fraction of a share) of New Parent Common Stock, to be held immediately after completion of the Merger in the treasury of New Parent.
 - c. <u>Conversion of Merger Sub Common Stock</u>. Each share of Merger Sub Common Stock issued and outstanding immediately prior to the Effective Time shall be converted into one validly issued, fully paid and nonassessable share of common stock, par value \$0.10 per share, of the Surviving Corporation.
 - d. <u>Rights of Certificate Holders</u>. Upon conversion thereof in accordance with this <u>Section 8</u>, all shares of Company Common Stock or Company Preferred Stock shall no longer be outstanding and shall cease to exist, and each holder of a certificate representing any such shares of Company Common Stock or Company Preferred Stock shall cease to have any rights with respect to such shares of Company Common Stock or Company Preferred Stock, respectively, except, in all cases, as set forth in <u>Section 9</u> herein. In addition, each outstanding book-entry that, immediately prior to the Effective Time, evidenced shares of Company Common Stock or Company Preferred Stock shall, from and after the Effective Time, be deemed and treated for all corporate purposes to evidence the ownership of the same number of shares of New Parent Common Stock or New Parent Preferred Stock, respectively.
- 9. CERTIFICATES. At and after the Effective Time until thereafter surrendered for transfer or exchange in the ordinary course, each outstanding certificate which immediately prior thereto represented shares of Company Common Stock or Company Preferred Stock shall be deemed for all purposes to evidence ownership of and to represent the shares of New Parent Common Stock or New Parent Preferred Stock, as applicable, into which the shares of Company Common Stock or Company Preferred Stock represented by such certificate have been converted as herein provided and shall be so registered on the books and records of New Parent and its transfer agent. At and after the Effective Time, the shares of New Parent Common Stock shall be uncertificated; provided, that, any shares of New Parent Common Stock that are represented by outstanding certificates of the Company pursuant to the immediately preceding sentence shall continue to be represented by certificates as provided therein and shall not be uncertificated unless and until a valid certificate representing such shares pursuant to the immediately preceding sentence is delivered to New Parent at its registered office in the State of Delaware, its principal place of business, or an officer or agent of New Parent having custody of books and records of New Parent, at which time such certificate shall be canceled and in lieu of the delivery of a certificate

representing the applicable shares of New Parent Common Stock, New Parent shall (i) issue to such holder the applicable uncertificated shares of New Parent Common Stock by registering such shares in New Parent's books and records as book-entry shares, upon which such shares shall thereafter be uncertificated and (ii) take all action necessary to provide such holder with evidence of the uncertificated book-entry shares, including any action necessary under applicable law in accordance therewith, including in accordance with Sections 151(f) and 202 of the DGCL. If any certificate that prior to the Effective Time represented shares of Company Common Stock or Company Preferred Stock shall have been lost, stolen or destroyed, then, upon the making of an affidavit of such fact by the person or entity claiming such certificate to be lost, stolen or destroyed and the providing of an indemnity by such person or entity to New Parent, in form and substance reasonably satisfactory to New Parent, against any claim that may be made against it with respect to such certificate, New Parent shall issue to such person or entity, in exchange for such lost, stolen or destroyed certificate, uncertificated shares representing the applicable shares of New Parent Common Stock or Company Preferred Stock in accordance with the procedures set forth in the preceding sentence.

10. ASSUMPTION OF EQUITY PLANS AND AWARDS AND OTHER AGREEMENTS. At the Effective Time, pursuant to this Agreement, New Parent and Company agree that they will execute, acknowledge and deliver an assignment and assumption agreement (the "Assignment and Assumption Agreement"), pursuant to which, effective as of the Effective Time, the Company will transfer to New Parent, and New Parent will (i) assume sponsorship of all of the Company's Equity Plans (as defined below), along with all of the Company's rights and obligations under the Equity Plans, and (ii) assume and agree to perform all obligations of the Company pursuant to each of the other agreements (the "Other Agreements") listed on Exhibit C hereto.

At the Effective Time, pursuant to this Agreement and the Assignment and Assumption Agreement, the Company will transfer to New Parent, and New Parent will assume the Equity Plans and the Company's rights and obligations under (i) each stock option to purchase a share of Company Common Stock (each, a "Stock Option"), (ii) each right to acquire or vest in a share of Company Common Stock pursuant to awards of restricted stock units (each, an "<u>RSU</u>") and pursuant to awards of performance shares (each, a "<u>Performance Share</u>"), and (iii) each right to purchase a share of Company Common Stock under the 2016 Employee Stock Purchase Plan (each, an "<u>ESPP Right</u>") (the Stock Options, RSUs, Performance Shares and ESPP Rights, collectively, the "<u>Awards</u>") issued under the Equity Plans that is outstanding and unexercised, unvested and not yet paid immediately prior to the Effective Time, which Awards shall be converted into a Stock Option or ESPP Right or an RSU or a Performance Share pursuant to which the holder may acquire a share of New Parent Common Stock with the same rights and privileges relative to New Parent that such Stock Option, ESPP Right, RSU or Performance Share had relative to the Company immediately prior to the Effective Time on otherwise the same terms and conditions as were applicable immediately prior to the Effective Time, including, for Stock Options, at an exercise price per share equal to the exercise price per share for the applicable share of Company Common Stock and for ESPP Rights at a purchase price determined in accordance with the terms of the 2016 Employee Stock Purchase Plan. For purposes of this Agreement, "<u>Equity Plans</u>" shall mean, collectively, the 2016 Omnibus Incentive Plan and the 2016 Employee Stock Purchase Plan and any and all subplans, appendices or addendums thereto, and any and all agreements evidencing Awards.

11. NEW PARENT SHARES. Prior to the Effective Time, the Company and New Parent shall take any and all actions as are necessary to ensure that each share of New Parent Common Stock that

is owned by the Company immediately prior to the Effective Time shall be cancelled and cease to be outstanding at the Effective Time, and no payment shall be made therefor, and the Company, by execution of this Agreement, agrees to forfeit such shares and relinquish any rights to such shares.

- 12. RESERVATION OF SHARES. At or prior to the Effective Time, New Parent will reserve sufficient authorized but unissued shares of New Parent Common Stock to provide for the issuance of New Parent Common Stock upon the exercise or satisfaction of rights and obligations under outstanding Awards under the Equity Plans.
- **13. NEW YORK STOCK EXCHANGE LISTING.** Prior to the Effective Time, the New Parent Common Stock to be issued pursuant to the Merger shall have been approved for listing, upon official notice of issuance, by the New York Stock Exchange, to the extent such approval is required.
- 14. TAX OPINION. The board of directors of the Company has determined, based on the advice of Hogan Lovells US LLP, special tax counsel to the Company, that, the Merger, when effective, will qualify as a "reorganization" within the meaning of Section 368(a) of the Code, <u>provided</u>, that, the Company shall have received the opinion of Hogan Lovells US LLP dated the Effective Time to that effect, substantially in the form previously provided to the board of directors of the Company.
- **15. NO APPRAISAL RIGHTS.** In accordance with the DGCL, no appraisal rights shall be available to any holder of shares of Company Common Stock in connection with the Merger.
- 16. TERMINATION. This Agreement may be terminated, and the Merger and the other transactions provided for herein may be abandoned, whether before or after the adoption of this Agreement by the mutual written consent of the Parties. In the event of termination of this Agreement, this Agreement shall forthwith become void and have no effect, and neither the Company, New Parent, Merger Sub nor their respective stockholders, directors or officers shall have any liability with respect to such termination or abandonment.
- 17. AMENDMENTS. At any time prior to the Effective Time, this Agreement may be supplemented, amended or modified, whether before or after the adoption of this Agreement by the mutual written consent of the Parties; <u>provided</u>, <u>however</u>, that, no amendment shall be effected subsequent to the adoption of this Agreement that by law requires further approval or authorization by the sole stockholder of Merger Sub or the stockholders of the Company without such further approval or authorization. No amendment of any provision of this Agreement shall be in writing and signed by all of the Parties hereto.
- **18. GOVERNING LAW.** This Agreement shall be governed by, and construed in accordance with, the laws of the State of Delaware, regardless of the laws that might otherwise govern under applicable principles of conflicts of laws.
- **19. COUNTERPARTS.** This Agreement may be executed in one or more counterparts, each of which when executed shall be deemed to be an original but all of which shall constitute one and the same agreement.
- **20. ENTIRE AGREEMENT.** This Agreement, including the documents and instruments referred to herein, constitutes the entire agreement and supersedes all other prior agreements and undertakings, both written and oral, among the Parties, or any of them, with respect to the subject matter hereof.

21. SEVERABILITY. The provisions of this Agreement are severable, and in the event any provision hereof is determined to be invalid or unenforceable, such invalidity or unenforceability shall not in any way affect the validity or enforceability of the remaining provisions hereof.

[Signature Page Follows]

IN WITNESS WHEREOF, the Company, New Parent and Merger Sub have caused this Agreement to be executed as of the date first written above by their respective officers thereunto duly authorized.

LABORATORY CORPORATION OF AMERICA HOLDINGS

By: <u>/s/ Sandra D. van der Vaart</u>

Name: Sandra D. van der Vaart Title: Executive Vice President, Chief Legal Officer, Chief Compliance Officer, and Secretary

LABCORP HOLDINGS INC.

By: <u>/s/ Sandra D. Van der Vaart</u> Name: Sandra D. van der Vaart Title: President and Secretary

RADIANCE MERGER SUB INC.

By: <u>/s/ Sandra D. van der Vaart</u> Name: Sandra D. van der Vaart Title: President and Secretary

Exhibit A – Certificate of Merger

Certificate of Merger. See attached.

CERTIFICATE OF MERGER

OF

RADIANCE MERGER SUB INC. (a Delaware corporation)

WITH AND INTO

LABORATORY CORPORATION OF AMERICA HOLDINGS (a Delaware corporation)

Pursuant to Section 251 of the General Corporation Law of the State of Delaware (the "DGCL"), the undersigned corporation does hereby certify the following information in connection with the merger (the "Merger") of Radiance Merger Sub Inc., a Delaware corporation ("Merger Sub"), with and into Laboratory Corporation of America Holdings, a Delaware corporation ("Corporation"):

FIRST: The name and state of incorporation of each of the constituent corporations in the Merger (the "Constituent Corporations") are as follows:

Name	State of Incorporation
Radiance Merger Sub Inc.	Delaware
Laboratory Corporation of America Holdings	Delaware

- **SECOND:** The Agreement and Plan of Merger, dated as of May 17, 2024 (the "Merger Agreement"), by and among Labcorp Holdings Inc., a corporation organized under the laws of the State of Delaware, Merger Sub and the Corporation, which expressly provides for the Merger to be effected under Section 251(g) of the DGCL, has been approved, adopted, executed and acknowledged by each of the constituent corporations in accordance with Section 251 of the DGCL.
- **THIRD:** The Corporation shall be the surviving corporation in the Merger (the "Surviving Corporation"), and the name of the Surviving Corporation shall be Laboratory Corporation of America Holdings.
- **FOURTH:** The certificate of incorporation of the Corporation, as in effect immediately prior to the Merger, shall be amended and restated in its entirety to read as set forth on <u>Exhibit A</u> attached hereto and, as so amended, shall be the amended and restated certificate of incorporation of the Surviving Corporation until further amended in accordance with its terms and the provisions of the DGCL.
- FIFTH: The executed Merger Agreement is on file at an office of the Surviving Corporation at 531 South Spring Street, Burlington, North Carolina, 27215.
- **SIXTH:** A copy of the Merger Agreement will be furnished by the Surviving Corporation, on request and without cost, to any stockholder of any Constituent Corporation.
- SEVENTH: This Certificate of Merger shall become effective at 4:05 p.m., Eastern Time, on May 17, 2024.

IN WITNESS WHEREOF, the Surviving Corporation has caused this Certificate of Merger to be signed by an authorized officer, this 17th day of May, 2024.

LABORATORY CORPORATION OF AMERICA HOLDINGS

By: _____ Name: Sandra D. van der Vaart Title: Executive Vice President, Chief Legal

Officer, Chief Compliance Officer and Secretary

Exhibit A

AMENDED AND RESTATED CERTIFICATE OF INCORPORATION

<u>OF</u>

LABORATORY CORPORATION OF AMERICA HOLDINGS

FIRST: The name of the Corporation is Laboratory Corporation of America Holdings (hereinafter the "Corporation").

SECOND: The address of the registered office of the Corporation in the State of Delaware is Corporation Trust Center, 1209 Orange Street, in the City of Wilmington, County of New Castle. The name of its registered agent at that address is The Corporation Trust Company.

THIRD: The purpose of the Corporation is to engage in any lawful act or activity for which a corporation may be organized under the General Corporation Law of the State of Delaware as set forth in Title 8 of the Delaware Code (the "GCL").

FOURTH: The total number of shares of stock which the Corporation has authority to issue is one thousand two hundred fifty (1,250) shares of which one thousand (1,000) will be shares of Common Stock, each having a par value of \$0.10, and two hundred fifty (250) will be shares of Preferred Stock, each having a par value of \$0.10.

The Board of Directors is expressly authorized to provide for the issuance of all or any shares of the Preferred Stock in one or more classes or series, and to fix for each such class or series such voting powers, full or limited, or no voting powers, and such distinctive designations, preferences and relative, participating, optional or other special rights and such qualifications, limitations or restrictions thereof, as shall be stated and expressed in the resolution or resolutions adopted by the Board of Directors providing for the issuance of such class or series and as may be permitted by the GCL, including without limitation, the authority to provide that any such class or series may be (i) subject to redemption at such time or times and at such price or prices; (ii) entitled to receive dividends (which may be cumulative or non-cumulative at such rates, on such conditions, and at such times, and payable in preference to or in such relation to, the dividends payable on any other class or classes or to any other series); (iii) entitled to such rights upon the dissolution of, or upon any distribution of the assets of, the Corporation; or (iv) other class or classes of stock, or of any other series of the same or any other class or classes of stock, of the Corporation at such price or prices or at such rates of exchange and with such adjustments; all as may be stated in such resolution or resolutions.

FIFTH: The following provisions are inserted for the management of the business and the conduct of the affairs of the Corporation, and for further definition, limitation and regulation of the powers of the Corporation and of its directors and stockholders:

(1) The business and affairs of the Corporation shall be managed by or under the direction of the Board of Directors.

A - 1

(2) The directors shall have concurrent power with the stockholders to make, alter, amend, change, add to or repeal the By-Laws of the Corporation.

(3) The number of directors of the corporation shall be as from time to time fixed by, or in the manner provided in, the By-Laws of the Corporation. Election of directors need not be by written ballot unless the By-Laws so provide.

(4) No director shall be personally liable to the Corporation or any of its stockholders for monetary damages for breach of fiduciary duty as a director, except for liability (i) for any breach of the director's duty of loyalty to the Corporation or its stockholders, (ii) for acts or omissions not in good faith or which involve intentional misconduct or a knowing violation of law, (iii) pursuant to Section 174 of the Delaware General Corporation Law or (iv) for any transaction from which the director derived an improper personal benefit. Any repeal or modification of this Article Sixth by the stockholders of the Corporation shall not adversely affect any right or protection of a director of the corporation existing at the time of such repeal or modification with respect to acts or omissions occurring prior to such repeal or modification.

(5) In addition to the powers and authority hereinbefore or by statute expressly conferred upon them, the directors are hereby empowered to exercise all such powers and do all such acts and things as may be exercised or done by the Corporation, subject, nevertheless, to the provisions of the GCL, this Certificate of Incorporation, and any By-Laws adopted by the stockholders; provided, however, that no By-Laws hereafter adopted by the stockholders shall invalidate any prior act of the directors which would have been valid if such By-Laws had not been adopted.

SIXTH: Meetings of stockholders may be held within or without the State of Delaware, as the By-Laws may provide. The books of the Corporation may be kept (subject to any provision contained in the GCL) outside the State of Delaware at such place or places as may be designated from time to time by the Board of Directors or in the By-Laws of the Corporation.

SEVENTH: The Corporation reserves the right to amend, alter, change or repeal any provision contained in this Certificate of Incorporation, in the manner now or hereafter prescribed by statute, and all rights conferred upon stockholders herein are granted subject to this reservation.

EIGHTH: Any act or transaction by or involving the Corporation, other than the election or removal of directors of the Corporation, that requires for its adoption under the GCL or this Certificate of Incorporation the approval of the stockholders of the Corporation shall, in accordance with Section 251(g) of the GCL, require, in addition, the approval of the stockholders of Laboratory Holdings Inc. (or any successor thereto by merger), by the same vote as is required by the GCL and/or this Certificate.

A - 1

Exhibit B – Surviving Corporation By-Laws

Surviving Corporation By-Laws See attached.

AMENDED AND RESTATED BY-LAWS OF

LABORATORY CORPORATION OF AMERICA HOLDINGS

(hereinafter called the "Corporation") (as amended and restated as of May 17, 2024)

1. OFFICES

1.1 Registered Office

The initial registered office of the Corporation in the State of Delaware shall be at 251 Little Falls Drive in the City of Wilmington, County of New Castle, Delaware 19808. The name of the initial registered agent of the Corporation at such address is the Corporation Service Company.

1.2 Other Offices

The Corporation may also have offices at such other places, both within and without the State of Delaware, as the Board of Directors may from time to time determine or as may be necessary or useful in connection with the business of the Corporation.

2. STOCKHOLDERS

2.1 Place of Meetings

All meetings of the stockholders shall be held at such place as may be fixed from time to time by the Board of Directors, the Chair or the President. Notwithstanding the foregoing, the Board of Directors may determine that the meeting shall not be held at any place, but may instead be held by means of remote communication.

2.2 Annual Meetings

Unless directors are elected by written consent in lieu of an annual meeting, the Corporation shall hold annual meetings of stockholders on such date and at such time as shall be designated from time to time by the Board of Directors, the Chair or the President, at which stockholders shall elect a Board of Directors and transact such other business as may properly be brought before the meeting. If a written consent electing directors is less than unanimous, such action by written consent may be in lieu of holding an annual meeting only if all of the directorships to which directors could be elected at an annual meeting held at the effective time of such action are vacant and are filled by such action.

2.3 Special Meetings

Special meetings of the stockholders, for any purpose or purposes, unless otherwise prescribed by statute, may be called by the Board of Directors or by the record holders of at least a majority of the shares of stock of the Corporation issued and outstanding and entitled to vote thereof.

2.4 Notice of Meetings

Notice of any meeting of stockholders, stating the place, if any, date and hour of the meeting, the means of remote communication, if any, by which stockholders and proxyholders may be deemed to be

present in person and vote at such meeting, and (if it is a special meeting) the purpose or purposes for which the meeting is called, shall be given to each stockholder entitled to vote at such meeting not less than 10 nor more than 60 days before the date of the meeting (except to the extent that such notice is waived or is not required as provided in the General Corporation Law of the State of Delaware (the "Delaware General Corporation Law") or these By-Laws). Such notice shall be given in accordance with, and shall be deemed effective as set forth in, Sections 222 and 232 (or any successor section or sections) of the Delaware General Corporation Law.

2.5 Waivers of Notice

Whenever the giving of any notice is required by statute, the Certificate of Incorporation or these By-Laws, a written waiver thereof signed by the person or persons entitled to said notice, or a waiver thereof by electronic transmission by the person entitled to said notice, delivered to the Corporation, whether before or after the event as to which such notice is required, shall be deemed equivalent to notice. Attendance of a stockholder at a meeting shall constitute a waiver of notice (1) of such meeting, except when the stockholder at the beginning of the meeting objects to holding the meeting or transacting business at the meeting, and (2) (if it is a special meeting) of consideration of a particular matter at the meeting that is not within the purpose or purposes described in the meeting notice, unless the stockholder objects to considering the matter at the beginning of the meeting.

2.6 Business at Special Meetings

Business transacted at any special meeting of stockholders shall be limited to the purposes stated in the notice (except to the extent that such notice is waived or is not required as provided in the Delaware General Corporation Law or these By-Laws).

2.7 Quorum at Meetings

Stockholders may take action on a matter at a meeting only if a quorum exists with respect to that matter. Except as otherwise provided by statute or by the Certificate of Incorporation, the holders of a majority of the shares entitled to vote at the meeting, and who are present in person or represented by proxy, shall constitute a quorum at all meetings of the stockholders for the transaction of business. Where a separate vote by a class or series or classes or series is required, a majority of the outstanding shares of such class or series or classes or series, present in person or represented by proxy, shall constitute a quorum entitled to take action with respect to that vote on that matter. Once a share is represented for any purpose at a meeting (other than solely to object (1) to holding the meeting or transacting business at the meeting, or (2) (if it is a special meeting) to consideration of a particular matter at the meeting that is not within the purpose or purposes described in the meeting notice), it is deemed present for quorum purposes for the remainder of the meeting and for any adjournment of that meeting unless a new record date is or must be set for the adjourned meeting. The holders of a majority of the voting shares represented at a meeting, whether or not a quorum is present, may adjourn such meeting from time to time.

2.8 Voting and Proxies

Unless otherwise provided in the Delaware General Corporation Law or in the Corporation's Certificate of Incorporation, and subject to the other provisions of these By-Laws, each stockholder shall be entitled to one vote on each matter, in person or by proxy, for each share of the Corporation's capital stock that has voting power and that is held by such stockholder. No proxy shall be voted or acted upon after three years from its date, unless the proxy provides for a longer period. A duly executed appointment of proxy shall be irrevocable if the appointment form states that it is irrevocable and if, and only as long as, it is coupled with an interest sufficient in law to support an irrevocable power. If authorized by the Board of Directors, and subject to such guidelines as the Board of Directors may adopt, stockholders and proxyholders not physically present at a meeting of stockholders may, by means of remote communication, participate in a meeting of stockholders

and be deemed present in person and vote at such meeting whether such meeting is held at a designated place or solely by means of remote communication, provided that (1) the Corporation implements reasonable measures to verify that each person deemed present and permitted to vote at the meeting by means of remote communication is a stockholder or proxyholder, (2) the Corporation implements reasonable measures to provide such stockholders and proxyholders a reasonable opportunity to participate in the meeting and to vote on matters submitted to the stockholders, including an opportunity to read or hear the proceedings of the meeting substantially concurrently with such proceedings, and (3) if any stockholder or proxyholder votes or takes other action at the meeting by means of remote communication, a record of such vote or other action is maintained by the Corporation.

2.9 Required Vote

When a quorum is present at any meeting of stockholders, all matters shall be determined, adopted and approved by the affirmative vote (which need not be by ballot) of the holders of a majority of the shares present in person or represented by proxy at the meeting and entitled to vote with respect to the matter, unless the proposed action is one upon which, by express provision of statutes or of the Certificate of Incorporation, a different vote is specified and required, in which case such express provision shall govern and control with respect to that vote on that matter. If the Certificate of Incorporation provides for more or less than one vote for any share, on any matter, every reference in these By-Laws to a majority or other proportion of stock, voting stock or shares shall refer to a majority or other proportion of the votes of such stock, voting stock or shares. Where a separate vote by a class or classes is required, the affirmative vote of the holders of a majority of the shares of such class or classes present in person or represented by proxy at the meeting shall be the act of such class. Notwithstanding the foregoing, directors shall be elected by a plurality of the votes of the shares present in person or represented by proxy at the meeting and entitled to vote on the election of directors.

2.10 Action Without a Meeting

Any action required or permitted to be taken at a stockholders' meeting may be taken without a meeting, without prior notice and without a vote, if the action is taken by persons who would be entitled to vote at a meeting and who hold shares having voting power equal to not less than the minimum number of votes that would be necessary to authorize or take the action at a meeting at which all shares entitled to vote were present and voted. The action must be evidenced by one or more written consents describing the action taken, signed by the stockholders entitled to take action without a meeting, and delivered to the Corporation in the manner prescribed by the Delaware General Corporation Law for inclusion in the minute book. No consent shall be effective to take the corporate action specified unless the number of consents required to take such action are delivered to the Corporation vithin 60 days of the delivery of the earliest-dated consent. An electronic transmission consenting to such action and transmitted by a stockholder or proxyholder, or by a person or persons authorized to act for a stockholder or proxyholder, shall be deemed to be written, signed and dated for the purposes of this Section 2.10. No consent given by electronic transmission shall be deemed to have been delivered until such consent is delivered to the Corporation in accordance with Section 228(d)(1) (or any successor section) of the Delaware General Corporation Law to all stockholders who do not participate in taking the action who would have been entitled to notice if such action had been taken at a meeting having a record date on the date that written consents signed by a sufficient number of holders to take the action were delivered to the Corporation.

1. DIRECTORS

3.1 Powers

The business and affairs of the Corporation shall be managed by or under the direction of the Board of Directors, which may exercise all such powers of the Corporation and do all such lawful acts and things, subject to any limitation set forth in the Certificate of Incorporation or as otherwise may be provided in the Delaware General Corporation Law.

3.2 Number

The number of directors which shall constitute the whole Board of Directors shall not be fewer than one member nor more than nine and shall be that number of members that the Board of Directors or stockholders determines shall be serving on the Board of Directors from time to time.

3.3 Election of Directors; Qualification

Except as provided in Section 3.4 hereof, directors shall be elected at the annual meeting of stockholders, and may be elected at any special meeting of stockholders for such purpose or by written consent in lieu of meeting pursuant to Section 2.10. Each director elected shall hold office until such director's successor is elected and qualified or until the director's earlier death, resignation or removal. Directors need not be stockholders.

3.4 Vacancies

Vacancies and newly created directorships resulting from any increase in the authorized number of directors elected by all of the stockholders having the right to vote as a single class may be filled by the affirmative vote of a majority of the directors then in office, although fewer than a quorum, or by a sole remaining director. Whenever the holders of any class or classes of stock or series thereof are entitled to elect one or more directors by the provisions of the Certificate of Incorporation, vacancies and newly created directorships of such class or classes or series may be filled by the affirmative vote of a majority of the director selected by such class or classes or series thereof then in office, or by a sole remaining director so elected. Each director so chosen shall hold office until the next election of directors of the class to which such director was appointed, and until such director's successor is elected and qualified, or until the director's earlier death, resignation or removal. In the event that one or more directors resign from the Board of Directors, effective at a future date, a majority of the directors then in office, including those who have so resigned, shall have power to fill such vacancy or vacancies, the vote thereon to take effect when such resignation or resignations shall become effective, and each director so chosen shall hold office until the next election of directors, and until such director's successor is elected and qualified, or until the director's and until such director's successor is elected and qualified of the event that one or more director so chosen shall hold office until the next election of directors then in office, including those who have so resigned, shall have power to fill such vacancy or vacancies, the vote thereon to take effect when such resignation or resignations shall become effective, and each director's earlier death, resignation or removal.

3.5 Meetings

3.5.1 Regular Meetings

Regular meetings of the Board of Directors may be held without notice at such time and at such place as shall from time to time be determined by the Board of Directors.

3.5.2 Special Meetings

Special meetings of the Board of Directors may be called by the Chair or President on one day's notice to each director, either personally or by telephone, express delivery service (so that the scheduled delivery date of the notice is at least one day in advance of the meeting), facsimile transmission, electronic mail (effective when directed to an electronic mail address of the director), or other electronic transmission, as defined in Section 232(c) (or any successor section) of the Delaware General Corporation Law (effective when directed to the director), and on five days' notice by mail (effective upon deposit of such notice in the mail). The notice need not describe the purpose of a special meeting.

3.5.3 Telephone Meetings

Members of the Board of Directors may participate in a meeting of the Board of Directors by means of telephone conference, videoconference or other communications equipment by means of which all participating directors can hear each other. A director participating in a meeting by this means is deemed to be present in person at the meeting.

3.5.4 Action Without Meeting

Any action required or permitted to be taken at any meeting of the Board of Directors may be taken without a meeting if the action is taken by all members of the Board of Directors. The action must be evidenced by one or more consents in writing or by electronic transmission describing the action taken, signed by each director, and delivered to the Corporation for inclusion in the minute book.

3.5.5 Waiver of Notice of Meeting

A director may waive any notice required by statute, the Certificate of Incorporation or these By-Laws before or after the date and time stated in the notice. Except as set forth below, the waiver must be in writing, signed by the director entitled to the notice, or made by electronic transmission by the director entitled to the notice, and delivered to the Corporation for inclusion in the minute book. Notwithstanding the foregoing, a director's attendance at or participation in a meeting waives any required notice to the director of the meeting unless the director at the beginning of the meeting objects to holding the meeting or transacting business at the meeting and does not thereafter vote for or assent to action taken at the meeting.

3.6 Quorum and Vote at Meetings

At all meetings of the Board of Directors, a quorum of the Board of Directors consists of a majority of the total number of directors prescribed pursuant to Section 3.2 of these By-Laws. The vote of a majority of the directors present at any meeting at which there is a quorum shall be the act of the Board of Directors, except as may be otherwise specifically provided by statute or by the Certificate of Incorporation or by these By-Laws.

3.7 Committees of Directors

The Board of Directors may designate one or more committees, each committee to consist of one or more directors. The Board of Directors may designate one or more directors as alternate members of any committee, who may replace any absent or disqualified member at any meeting of the committee. If a member of a committee shall be absent from any meeting, or disqualified from voting thereat, the remaining member or members present and not disqualified from voting, whether or not such member or members constitute a quorum, may, by unanimous vote, appoint another member of the Board of Directors to act at the meeting in the place of such absent or disqualified member. Any such committee, to the extent provided in the resolution of the Board of Directors, shall have and may exercise all the powers and authority of the Board of Directors in the management of the business and affairs of the Corporation, and may authorize the seal of the Corporation to be affixed to all papers which may require it; but no such committee shall have the power or authority in reference to approving or adopting, or recommending to the stockholders, any action or matter (other than the election or removal of directors) expressly required by the Delaware General Corporation Law to be submitted to stockholders for approval or adopting, amending or repealing any By-Law of the Corporation; and unless the resolution designating the committee, these By-Laws or the Certificate of Incorporation expressly so provide, no such committee shall have the power or authority to declare a dividend, to authorize the issuance of stock, or to adopt a certificate of ownership and merger pursuant to Section 253 (or any successor section) of the Delaware General Corporation Law. Such committee or committees shall have such name or names as may be determined from time to time by resolution adopted by the Board of Directors.

Each committee shall keep regular minutes of its meetings and report the same to the Board of Directors, when required. Unless otherwise specified in the Board of Directors' resolution appointing the committee, all provisions of the Delaware General Corporation Law and these By-Laws relating to meetings, action without meetings, notice (and waiver thereof), and quorum and voting requirements of the Board of Directors apply, as well, to such committees and their members. Unless otherwise provided in the Certificate of Incorporation, these By-Laws, or the resolution of the Board of Directors designating the committee, a committee may create one or more subcommittees, each subcommittee to consist of one or more members of the committee, and delegate to a subcommittee any or all of the powers and authority of the committee.

3.8 Compensation of Directors

The Board of Directors shall have the authority to fix the compensation of directors. No such payment shall preclude any director from serving the Corporation in any other capacity and receiving compensation therefor.

2. OFFICERS

4.1 Positions

The officers of the Corporation shall be a President, a Secretary and a Treasurer, and such other officers as the Board of Directors (or an officer authorized by the Board of Directors) from time to time may appoint. Each such officer shall exercise such powers and perform such duties as shall be set forth below and such other powers and duties as from time to time may be specified by the Board of Directors or by any officer(s) authorized by the Board of Directors to prescribe the duties of such other officers. Any number of offices may be held by the same person. As set forth below, the President and any other officer authorized by the Board of Directors may execute contracts under the seal of the Corporation, if required, except where required or permitted by law to be otherwise executed and except where the execution thereof shall be expressly delegated by the Board of Directors to some other officer or agent of the Corporation.

4.2 President

The President shall be the chief executive officer of the Corporation and shall have full responsibility and authority for management of the operations of the Corporation, subject to the authority of the Board of Directors. The President may execute contracts, under the seal of the Corporation, if required, except where required or permitted by law to be otherwise signed and executed and except where the signing and execution thereof shall be expressly delegated by the Board of Directors to some other officer or agent of the Corporation.

4.3 Secretary

The Secretary shall have responsibility for preparation of minutes of meetings of the Board of Directors and of the stockholders and for authenticating records of the Corporation. The Secretary shall give, or cause to be given, notice of all meetings of the stockholders and special meetings of the Board of Directors. The Secretary or an Assistant Secretary may also attest all instruments signed by any other officer of the Corporation.

4.4 Assistant Secretary

The Assistant Secretary, or if there be more than one, the Assistant Secretaries in the order determined by the Board of Directors (or if there shall have been no such determination, then in the order of their election), shall, in the absence of the Secretary or in the event of the Secretary's inability or refusal to act, perform the duties and exercise the powers of the Secretary.

4.5 Treasurer

The Treasurer shall be the chief financial officer of the Corporation and shall have responsibility for the custody of the corporate funds and securities and shall see to it that full and accurate accounts of receipts and disbursements are kept in books belonging to the Corporation.

The Treasurer shall render to the President, and the Board of Directors, upon request, an account of all financial transactions and of the financial condition of the Corporation.

4.6 Assistant Treasurer

The Assistant Treasurer, or if there shall be more than one, the Assistant Treasurers in the order determined by the Board of Directors (or if there shall have been no such determination, then in the order of their election), shall, in the absence of the Treasurer or in the event of the Treasurer's inability or refusal to act, perform the duties and exercise the powers of the Treasurer.

4.7 Term of Office

The officers of the Corporation shall hold office until their successors are chosen and qualify or until their earlier resignation or removal. Any officer may resign at any time upon written notice to the Corporation. Any officer elected or appointed by the Board of Directors may be removed at any time, with or without cause, by the affirmative vote of a majority of the Board of Directors.

4.8 Compensation

The compensation of officers of the Corporation shall be fixed by the Board of Directors or by any officer(s) authorized by the Board of Directors to prescribe the compensation of such other officers.

4.9 Fidelity Bonds

The Corporation may secure the fidelity of any or all of its officers or agents by bond or otherwise.

3. CAPITAL STOCK

5.1 Stock Certificates; Uncertificated Shares

The shares of the Corporation shall be represented by certificates, provided that the Board of Directors may provide by resolution that some or all of any or all classes or series of the Corporation's stock shall be uncertificated shares that may be evidenced by a book-entry system maintained by the registrar of such stock. Any such resolution shall not apply to shares represented by a certificate until such certificate is surrendered to the Corporation. Every holder of stock represented by certificates shall be entitled to have a certificate (representing the number of shares registered in certificate form) signed in the name of the Corporation by any two authorized officers of the Corporation. Any or all the signatures on the certificate may be a facsimile. In case any officer, transfer agent or registrar whose signature or facsimile signature appears on a certificate shall have ceased to be such officer, transfer agent or registrar at the date of issue.

5.2 Stock Transfers

Stock of the Corporation shall be transferable in the manner prescribed by law and in these By-Laws. Transfers of stock shall be made on the books of the Corporation only by the holder of record thereof, by such person's attorney lawfully constituted in writing and, in the case of certificated shares, upon the surrender of the certificate thereof, which shall be cancelled before a new certificate or uncertificated shares shall be issued. No transfer of stock shall be valid as against the Corporation for any purpose until it shall have been entered in the stock records of the Corporation by an entry showing from and to whom transferred. To the extent designated by the president or the treasurer of the Corporation, the Corporation may recognize the transfer of fractional uncertificated shares, but shall not otherwise be required to recognize the transfer of fractional shares.

5.3 Transfer Agents and Registrars

The Board of Directors may appoint, or authorize any officer or officers to appoint, one or more transfer agents and one or more registrars.

5.4 Lost Certificates

The Board of Directors may direct a new certificate of stock to be issued in place of any certificate theretofore issued by the Corporation and alleged to have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the person claiming that the certificate of stock has been lost, stolen or destroyed. When authorizing such issuance of a new certificate, the Board of Directors or any such officer may, as a condition precedent to the issuance thereof, require the owner of such lost, stolen or destroyed certificates, or such owner's legal representative, to advertise the same in such manner as the Board of Directors or such officer shall require and/or to give the Corporation a bond or indemnity, in such sum or on such terms and conditions as the Board of Directors or such officer may direct, as indemnity against any claim that may be made against the Corporation on account of the certificate alleged to have been lost, stolen or destroyed or on account of the issuance of such new certificate or uncertificate shares.

5.5 Record Date

5.5.1 Actions by Stockholders

In order that the Corporation may determine the stockholders entitled to notice of or to vote at any meeting of stockholders, the Board of Directors may fix a record date, which record date shall not precede the date upon which the resolution fixing the record date is adopted by the Board of Directors, and which record date shall not be more than 60 days nor less than 10 days before the date of such meeting. If no record date is fixed by the Board of Directors, the record date for determining stockholders entitled to notice of or to vote at a meeting of stockholders shall be the close of business on the day next preceding the day on which notice is given, or, if notice is waived, at the close of business on the day next preceding the day on stockholders of record entitled to notice of or to vote at a meeting of stockholders shall apply to any adjournment of the meeting, unless the Board of Directors fixes a new record date for the adjourned meeting.

In order that the Corporation may determine the stockholders entitled to consent to corporate action in writing without a meeting, the Board of Directors may fix a record date, which record date shall not precede the date upon which the resolution fixing the record date is adopted by the Board of Directors, and which record date shall not be more than 10 days after the date upon which the resolution fixing the record date is adopted by the Board of Directors. If no record date has been fixed by the Board of Directors is required by the Delaware entitled to consent to corporate action in writing without a meeting, when no prior action by the Board of Directors is required by the Delaware General Corporation Law, shall be the first date on which a signed written consent setting forth the action taken or proposed to be taken is delivered to the Corporation in the manner prescribed by Section 213(b) (or any successor section) of the Delaware General Corporation Law. If no record date has been fixed of

Directors is required by the Delaware General Corporation Law, the record date for determining stockholders entitled to consent to corporate action in writing without a meeting shall be at the close of business on the day on which the Board of Directors adopts the resolution taking such prior action.

5.5.2 Payments

In order that the Corporation may determine the stockholders entitled to receive payment of any dividend or other distribution or allotment of any rights or the stockholders entitled to exercise any rights in respect of any change, conversion or exchange of stock, or for the purpose of any other lawful action, the Board of Directors may fix a record date, which record date shall not precede the date upon which the resolution fixing the record date is adopted, and which record date shall be not more than 60 days prior to such action. If no record date is fixed, the record date for determining stockholders for any such purpose shall be at the close of business on the day on which the Board of Directors adopts the resolution relating thereto.

5.6 Stockholders of Record

The Corporation shall be entitled to recognize the exclusive right of a person registered on its books as the owner of shares to receive dividends, to receive notifications, to vote as such owner, and to exercise all the rights and powers of an owner. The Corporation shall not be bound to recognize any equitable or other claim to or interest in such share or shares on the part of any other person, whether or not it shall have express or other notice thereof, except as otherwise may be provided by the Delaware General Corporation Law.

4. INDEMNIFICATION

6.1 Directors

The Corporation shall indemnify its directors and executive officers to the fullest extent not prohibited by the Delaware General Corporation Law or any other applicable law; <u>provided</u>, <u>however</u>, that the Corporation may limit the extent of such indemnification by individual contracts with its directors and executive officers; and, <u>provided</u>, <u>further</u>, that the Corporation shall not be required to indemnify any director or executive officer in connection with any proceeding (or part thereof) initiated by such person or any proceeding by such person against the Corporation or its directors, officers, employees or other agents unless (1) such indemnification is expressly required to be made by law, (2) the proceeding was authorized by the Board of Directors of the Corporation, (3) such indemnification is provided by the Corporation, in its sole discretion, pursuant to the powers vested in the Corporation under the Delaware General Corporation Law or any other applicable law or (4) such indemnification is required under Section 6.5.

6.2 Officers, Employees and Other Agents

The Corporation shall have power to indemnify its officers, employees and other agents as set forth in the Delaware General Corporation Law or any other applicable law. The Board of Directors shall have the power to delegate the determination of whether indemnification shall be given to any such person to such officers or other persons as the Board of Directors shall determine.

6.3 Good Faith

For purposes of any determination under this Section 6.3, a director or executive officer shall be deemed to have acted in good faith and in a manner he or she reasonably believed to be in or not opposed to the best interests of the Corporation, and, with respect to any criminal action or proceeding, to have had no reasonable cause to believe that his or her conduct was unlawful, if his or her action is based on information, opinions, reports and statements, including financial statements and other financial data, in each case prepared

or presented by: (1) one or more officers or employees of the Corporation whom the director or executive officer believed to be reliable and competent in the matters presented; (2) counsel, independent accountants or other persons as to matters which the director or executive officer believed to be within such person's professional competence; and (3) with respect to a director, a committee of the Board of Directors upon which such director does not serve, as to matters within such committee's designated authority, which committee the director believes to merit confidence; so long as, in each case, the director or executive officer acts without knowledge that would cause such reliance to be unwarranted.

The termination of any proceeding by judgment, order, settlement, conviction or upon a plea of nolo contendere or its equivalent shall not, of itself, create a presumption that the person did not act in good faith and in a manner which such person reasonably believed to be in or not opposed to the best interests of the Corporation, and, with respect to any criminal proceeding, that such person had reasonable cause to believe that his or her consent was unlawful.

The provisions of this Section 6.3 shall not be deemed to be exclusive or to limit in any way the circumstances in which a person may be deemed to have met the applicable standard of conduct set forth by the Delaware General Corporation Law.

6.4 Expenses

The Corporation shall advance to any person who was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative, by reason of the fact that he is or was a director of the Corporation, or is or was serving at the request of the Corporation as a director or officer of another corporation, partnership, joint venture, trust or other enterprise, prior to the final disposition of the proceeding, promptly following request therefor, all expenses incurred by any director or executive officer in connection with such proceeding; provided, however, that, if the Delaware General Corporation Law requires, an advancement of expenses incurred by a director in his or her capacity as a director (and not in any other capacity in which service was or is rendered by such indemnitee, including, without limitation, service to an employee benefit plan) shall be made only upon delivery to the Corporation of an undertaking, by or on behalf of such indemnitee, to repay all amounts so advanced if it shall ultimately be determined by final judicial decision from which there is no further right to appeal that such indemnitee is not entitled to be indemnified for such expenses under this section or otherwise.

Notwithstanding the foregoing, unless otherwise determined pursuant to this Section 6.4, no advance shall be made by the Corporation if a determination is reasonably and promptly made (1) by the Board of Directors by a majority vote of a quorum consisting of Directors who were not parties to the proceeding or (2) if such quorum is not obtainable, or, even if obtainable, a quorum of disinterested Directors so directs, by independent legal counsel in a written opinion, that the facts known to the decision-making party at the time such determination is made demonstrate clearly and convincingly that such person acted in bad faith or in a manner that such person did not believe to be in or not opposed to the best interests of the Corporation.

6.5 Enforcement

Without the necessity of entering into an express contract, all rights to indemnification and advances to directors under this By-Law shall be deemed to be contractual rights and be effective to the same extent and as if provided for in a contract between the Corporation and the director or executive officer. Any right to indemnification or advances granted by this By-Law to a director shall be enforceable by or on behalf of the person holding such right in any court of competent jurisdiction if (1) the claim for indemnification or advances is denied, in whole or in part, or (2) no disposition of such claim is made within ninety 90 days of request therefor. To the extent permitted by law, the claimant in such enforcement action, if successful in whole or in part, shall be entitled to be paid also the expense of prosecuting the claim. In connection with any claim for indemnification, the Corporation shall be entitled to raise as a defense to any such action that the

claimant has not met the standards of conduct that make it permissible under the Delaware General Corporation Law or any other applicable law for the Corporation to indemnify the claimant for the amount claim ed. Neither the failure of the Corporation (including its Board of Directors, independent legal counsel or its stockholders) to have made a determination prior to the commencement of such action that indemnification of the claimant is proper in the circumstances because he or she has met the applicable standard of conduct set forth in the Delaware General Corporation Law or any other applicable law, nor an actual determination by the Corporation (including its Board of Directors, independent legal counsel or its stockholders) that the claimant has not met such applicable standard of conduct, shall be a defense to the action or create a presumption that the claimant has not met the applicable standard of conduct. In any suit brought by a director or executive officer to enforce a right to indemnification or to an advancement of expenses hereunder, the burden of proving that the director is not entitled to be indemnified, or to such advancement of expenses, under this section or otherwise shall be on the Corporation.

6.6 Non-Exclusivity of Rights

The rights conferred on any person by this By-Law shall not be exclusive of any other right which such person may have or hereafter acquire under any applicable statute, provision of the Certificate of Incorporation, By-Laws, agreement, vote of stockholders or disinterested directors or otherwise, both as to action in his or her official capacity and as to action in another capacity while holding office. The Corporation is specifically authorized to enter into individual contracts with any or all of its directors, officers, employees or agents respecting indemnification and advances, to the fullest extent not prohibited by the Delaware General Corporation Law, or by any other applicable law.

6.7 Survival of Rights

The rights conferred on any person by this By-Law shall continue as to a person who has ceased to be a director, officer, employee or other agent and shall inure to the benefit of the heirs, executors and administrators of such a person.

6.8 Insurance

To the fullest extent permitted by the Delaware General Corporation Law or any other applicable law, the Corporation, upon approval by the Board of Directors, may purchase insurance on behalf of any person required or permitted to be indemnified pursuant to this section.

6.9 Amendments

Any repeal or modification of this section shall only be prospective and shall not affect the rights under this By-Law in effect at the time of the alleged occurrence of any action or omission to act that is the cause of any proceeding against any agent of the Corporation.

6.10 Savings Clause

If this By-Law or any portion hereof shall be invalidated on any ground by any court of competent jurisdiction, then the Corporation shall nevertheless indemnify each director and executive officer to the fullest extent not prohibited by any applicable portion of this section that shall not have been invalidated, or by any other applicable law. If this section shall be invalidated due to the application of the indemnification provisions of another jurisdiction, then the Corporation shall indemnify each director to the full extent under any other applicable law.

6.11 Certain Definitions.

For the purposes of these By-Laws, the following definitions shall apply:

6.11.1

The term "proceeding" shall be broadly construed and shall include, without limitation, the investigation, preparation, prosecution, defense, settlement, arbitration and appeal of, and the giving of testimony in, any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative.

6.11.2

The term "expenses" shall be broadly construed and shall include, without limitation, court costs, attorneys' fees, witness fees, fines, amounts paid in settlement or judgment and any other costs and expenses of any nature or kind incurred in connection with any proceeding.

6.11.3

The term the "Corporation" shall include, in addition to the resulting Corporation, any constituent corporation (including any constituent of a constituent) absorbed in a consolidation or merger which, if its separate existence had continued, would have had power and authority to indemnify its directors, officers, and employees or agents, so that any person who is or was a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise, shall stand in the same position under the provisions of this section with respect to the resulting or surviving corporation as he would have with respect to such constituent corporation if its separate existence had continued.

6.11.4

References to a "director," "executive officer," "officer," "employee," or "agent" of the Corporation shall include, without limitation, situations where such person is serving at the request of the Corporation as, respectively, a director, executive officer, officer, employee, trustee or agent of another corporation, partnership, joint venture, trust or other enterprise.

6.11.5

References to "other enterprises" shall include employee benefit plans; references to "fines" shall include any excise taxes assessed on a person with respect to an employee benefit plan; and references to "serving at the request of the Corporation" shall include any service as a director, officer, employee or agent of the Corporation which imposes duties on, or involves services by, such director, officer, employee, or agent with respect to an employee benefit plan, its participants, or beneficiaries; and a person who acted in good faith and in a manner such person reasonably believed to be in the interest of the participants and beneficiaries of an employee benefit plan shall be deemed to have acted in a manner "not opposed to the best interests of the Corporation" as referred to in this section.

5. GENERAL PROVISIONS

7.1 Inspection of Books and Records

Any stockholder, in person or by attorney or other agent, shall, upon written demand under oath stating the purpose thereof, have the right during the usual hours for business to inspect for any proper purpose, and to make copies or extracts from: (1) the Corporation's stock ledger, a list of its stockholders, and its other books and records; and (2) other documents as required by law. A proper purpose shall mean a

purpose reasonably related to such person's interest as a stockholder. In every instance where an attorney or other agent shall be the person who seeks the right to inspection, the demand under oath shall be accompanied by a power of attorney or such other writing which authorizes the attorney or other agent to so act on behalf of the stockholder. The demand under oath shall be directed to the Corporation at its registered office or at its principal place of business.

7.2 Dividends

The Board of Directors may declare dividends upon the capital stock of the Corporation, subject to the provisions of the Certificate of Incorporation and the laws of the State of Delaware.

7.3 Reserves

The directors of the Corporation may set apart, out of the funds of the Corporation available for dividends, a reserve or reserves for any proper purpose and may abolish any such reserve.

7.4 Execution of Instruments

All checks, drafts or other orders for the payment of money, and promissory notes of the Corporation shall be signed by such officer or officers or such other persons as the Board of Directors may from time to time designate.

7.5 Fiscal Year

The fiscal year of the Corporation shall be fixed by resolution of the Board of Directors.

7.6 Seal

The corporate seal shall be in such form as the Board of Directors shall approve. The seal may be used by causing it or a facsimile

Exhibit C – Other Agreements

Executive Employment Agreement, dated June 4, 2019, by and between Laboratory Corporation of America Holdings and Adam H. Schechter.

C - 1

AMENDED AND RESTATED CERTIFICATE

OF INCORPORATION OF

LABCORP HOLDINGS INC.

Pursuant to Sections 242, 245 and 228 of the General

General Corporation Law of the State of Delaware

* * * * *

Labcorp Holdings Inc., a corporation organized and existing under the laws of the State of Delaware (the "Corporation"), hereby certifies as follows:

1. The name of the Corporation is Labcorp Holdings Inc. The Corporation's original Certificate of Incorporation was filed with the Secretary of State of the State of Delaware on April 16, 2024.

2. This Amended and Restated Certificate of Incorporation was duly adopted in accordance with the provisions of Section 242, Section 245 and Section 228 (by written consent of the sole stockholder of the Corporation) of the General Corporation Law of the State of Delaware, and restates, integrates and further amends provisions of the Corporation's Certificate of Incorporation.

3. The text of the Certificate of Incorporation of the Corporation is hereby amended and restated in its entirety, effective May 17, 2024, at 4:05 p.m., EDT, as set forth in <u>Exhibit A</u> attached hereto.

IN WITNESS WHEREOF, the Corporation has caused this certificate to be signed by Sandra D. van der Vaart, President and Secretary, on this 17th day of May, 2024.

LABCORP HOLDINGS INC.

By: <u>/s/ Sandra D. van der Vaart</u> Sandra D. van der Vaart President and Secretary

Exhibit A

FIRST: The name of the Corporation is Labcorp Holdings Inc. (hereinafter the "Corporation").

SECOND: The address of the registered office of the Corporation in the State of Delaware is 251 Little Falls Drive in the City of Wilmington, County of New Castle, Delaware 19808. The name of its registered agent at that address is the Corporation Service Company.

THIRD: The purpose of the Corporation is to engage in any lawful act or activity for which a corporation may be organized under the General Corporation Law of the State of Delaware as set forth in Title 8 of the Delaware Code (the "GCL").

FOURTH: The total number of shares of stock which the Corporation has authority to issue is two hundred and ninety-five million (295,000,000) shares of which two hundred and sixty-five million (265,000,000) will be shares of Common Stock, each having a par value of \$0.10, and thirty million (30,000,000) will be shares of Preferred Stock, each having a par value of \$0.10.

The Board of Directors is expressly authorized to provide for the issuance of all or any shares of the Preferred Stock in one or more classes or series, and to fix for each such class or series such voting powers, full or limited, or no voting powers, and such distinctive designations, preferences and relative, participating, optional or other special rights and such qualifications, limitations or restrictions thereof, as shall be stated and expressed in the resolution or resolutions adopted by the Board of Directors providing for the issuance of such class or series and as may be permitted by the GCL, including without limitation, the authority to provide that any such class or series may be (i) subject to redemption at such time or times and at such price or prices; (ii) entitled to receive dividends (which may be cumulative or non-cumulative at such rates, on such conditions, and at such times, and payable in preference to or in such relation to, the dividends payable on any other class or classes or to any other series); (iii) entitled to such rights upon the dissolution of, or upon any distribution of the assets of, the Corporation; or (iv) other class or classes of stock, or of any other series of the same or any other class or classes of stock, of the Corporation at such price or prices or at such rates of exchange and with such adjustments; all as may be stated in such resolution or resolutions.

FIFTH: The following provisions are inserted for the management of the business and the conduct of the affairs of the Corporation, and for further definition, limitation and regulation of the powers of the Corporation and of its directors and stockholders:

(1) The business and affairs of the Corporation shall be managed by or under the direction of the Board of Directors.

(2) The directors shall have concurrent power with the stockholders to make, alter, amend, change, add to or repeal the By-Laws of the Corporation.

(3) The number of directors of the corporation shall be as from time to time fixed by, or in the manner provided in, the By-Laws of the Corporation. Election of directors need not be by written ballot unless the By-Laws so provide.

(4) No director shall be personally liable to the Corporation or any of its stockholders for monetary damages for breach of fiduciary duty as a director, except for liability (i) for any breach of the director's duty of loyalty to the Corporation or its stockholders, (ii) for acts or omissions not in good faith or which involve intentional misconduct or a knowing violation of law, (iii) pursuant to Section 174 of the Delaware General Corporation Law or (iv) for any transaction from which the director derived an improper personal benefit. Any repeal or modification of this Article Sixth by the stockholders of the Corporation shall not adversely affect any right or protection of a director of the corporation existing at the time of such repeal or modification with respect to acts or omissions occurring prior to such repeal or modification.

(5) In addition to the powers and authority hereinbefore or by statute expressly conferred upon them, the directors are hereby empowered to exercise all such powers and do all such acts and things as may be exercised or done by the Corporation, subject, nevertheless, to the provisions of the GCL, this Certificate of Incorporation, and any By-Laws adopted by the stockholders; provided, however, that no By-Laws hereafter adopted by the stockholders shall invalidate any prior act of the directors which would have been valid if such By-Laws had not been adopted.

SIXTH: Meetings of stockholders may be held within or without the State of Delaware, as the By-Laws may provide. The books of the Corporation may be kept (subject to any provision contained in the GCL) outside the State of Delaware at such place or places as may be designated from time to time by the Board of Directors or in the By-Laws of the Corporation.

SEVENTH: The Corporation reserves the right to amend, alter, change or repeal any provision contained in this Certificate of Incorporation, in the manner now or hereafter prescribed by statute, and all rights conferred upon stockholders herein are granted subject to this reservation.

AMENDED AND RESTATED BY-LAWS OF

LABCORP HOLDINGS INC.

(hereinafter called the "Corporation") (as amended as of May 17, 2024)

ARTICLE I MEETINGS OF STOCKHOLDERS

Section 1. Place of Meetings. Meetings of the stockholders for the election of directors or for any other purpose shall be held at such time and place, either within or without the State of Delaware, as shall be designated from time to time by the Board of Directors (or the Chairman or Vice Chairman, if any of the Board of Directors in the absence of a designation by the Board of Directors) and stated in the notice of the meeting or in a duly executed waiver of notice thereof.

Section 2. Annual Meetings. The Annual Meetings of Stockholders shall be held on such date and at such time as shall be designated from time to time by the Board of Directors and stated in the notice of the meeting, at which meetings the stockholders shall elect by a majority vote a Board of Directors, as provided in Section 1 of Article II, and transact such other business as may properly be brought before the meeting. Except as otherwise permitted or required by applicable laws or regulations, notice of the Annual Meeting stating the place, date and hour of the meeting shall be given to each stockholder entitled to vote at such meeting not less than ten nor more than sixty days before the date of the meeting. Notice may be given in any manner permitted by applicable laws and regulations, as provided in Article V.

Section 3. Special Meetings. (a) Unless otherwise prescribed by law or by the Certificate of Incorporation, Special Meetings of Stockholders, for any purpose or purposes, may be called at any time by the Board of Directors.

(b) Except as otherwise permitted or required by applicable laws or regulations, notice of a Special Meeting stating the place, if any, date and hour of the meeting and the purpose or purposes for which the meeting is called shall be given not less than ten nor more than sixty days before the date of the meeting to each stockholder entitled to vote at such meeting. Notice may be given in any manner permitted by applicable laws and regulations, as provided in Article V.

(c) Subject to the provisions of this Section 3(c), a Special Meeting of Stockholders shall be called by a majority of the entire Board of Directors following receipt by the Secretary of the Corporation of a written request for a special meeting (a "Special Meeting Request") from one, or a group of persons who have owned at least ten percent of the combined voting power of the then outstanding shares of all classes and series of capital stock of the Corporation entitled generally to vote in the election of directors of the Corporation, voting as a single class, continuously for at least one year as of both (i) a date within seven days prior to the date of the Special Meeting

Request and (ii) the record date for determining stockholders entitled to vote at the Special Meeting (the "Requisite Holders"), if such Special Meeting Request complies with the requirements of this Section 3(c) and all other applicable sections of these By-Laws. For purposes of satisfying the foregoing ownership requirement under this Section 3(c), (i) the term "owned" shall have the same meaning as Section 13(e) of this Article I, and (ii) the shares of the capital stock of the Corporation owned by one or more stockholders, or by the person or persons who own shares of the capital stock of the Corporation and on whose behalf any stockholder is acting, may be aggregated. For the avoidance of doubt, if a group of stockholders aggregates ownership of shares in order to meet the requirements under this Section 3(c), all shares held by each stockholder constituting their contribution to the foregoing ten percent threshold must be held by that stockholder continuously for at least one year, and evidence of such continuous ownership shall be provided as specified in this Section 3(c). The Board of Directors shall determine whether all requirements set forth in this Section 3 and these By-Laws have been satisfied and such determination shall be binding on the Corporation and its stockholders. If a Special Meeting Request is made that complies with this Section 3(c) and all other applicable sections of these By-Laws, the Board of Directors may (in lieu of calling the Special Meeting of Stockholders requested in such Special Meeting Request) present an identical or substantially similar item (a "Similar Item") for stockholder approval at any other meeting of stockholders that is held within ninety days after the Corporation receives such Special Meeting Request.

A Special Meeting Request must be delivered to the Secretary of the Corporation at the principal executive offices of the Corporation. A Special Meeting Request shall only be valid if it is signed and dated by each of the stockholders that is one of the Requisite Holders and include: (i) a statement of the specific purpose(s) of the Special Meeting of Stockholders, the matter(s) proposed to be acted on at the Special Meeting of Stockholders, and the reasons for conducting such business at the Special Meeting of Stockholders; (ii) the text of any proposed amendment to the By-Laws to be considered at the Special Meeting of Stockholders; (iii) the name and address of each stockholder of record signing such request, the date of each such stockholder's signature, and the name and address of any beneficial owner on whose behalf such request is made; (iv) the class or series and number of shares of the Corporation that are owned of record or beneficially by each such stockholder and any such beneficial owner and documentary evidence of such record or beneficial ownership; (v) any material interest of each stockholder or any such beneficial owner in any of the business proposed to be conducted at the Special Meeting of Stockholders and a description of all arrangements or understandings between any such stockholder and/or beneficial owner and any other person or persons (naming such person or persons) with respect to the business proposed to be conducted; (vi) a representation that one or more of the stockholders submitting the Special Meeting Request intend to appear in person or by proxy at the Special Meeting of Stockholders to present the proposal(s) or business to be brought before the Special Meeting of Stockholders; (vii) if any stockholder submitting such request intends to solicit proxies with respect to the stockholders' proposal(s) or business to be presented at the Special Meeting of Stockholders, a representation to that effect; (viii) all information relating to each stockholder signing the Special Meeting Request that must be disclosed in solicitations for proxies for election of directors in an election contest (even if an election contest is not involved), or is otherwise required, in each case pursuant to Regulation 14A under the 1934 Act (as defined below in Article II); and (ix) if the purpose of the Special Meeting of Stockholders includes the election of one or more directors, all the information such stockholder or stockholders would be required to include in a notice delivered to the Corporation pursuant to Article I, Section 9 of these By-Laws.

In addition, a Special Meeting Request shall not be valid if (i) the Special Meeting Request relates to an item of business that is not a proper subject for stockholder action under applicable law; (ii) the Special Meeting Request is received by the Corporation during the period commencing one hundred and twenty days prior to the first anniversary of the date of the immediately preceding annual meeting and ending on the date of the next annual meeting; (iii) a Similar Item was presented at any meeting of stockholders held within ninety days prior to receipt by the Corporation of such Special Meeting Request (and, for purposes of this clause (iii), the election of directors shall be deemed a "Similar Item" with respect to all items of business involving the election or removal of directors); (iv) a Similar Item is included in the Corporation's notice as an item of business to be brought before a stockholder meeting that has been called but not yet held; or (v) such Special Meeting Request was made in a manner that involved a violation of Regulation 14A under the 1934 Act, or other applicable law.

Stockholders may revoke a Special Meeting Request by written revocation delivered to the Corporation at any time prior to the Special Meeting of Stockholders; provided, however, the Board of Directors shall have the discretion to determine whether or not to proceed with the Special Meeting of Stockholders.

If none of the stockholders who submitted the Special Meeting Request for a Special Meeting of Stockholders appears or sends a qualified representative to present the proposal(s) or business submitted by the stockholders for consideration at the Special Meeting of Stockholders, the Corporation need not present such proposal(s) or business for a vote at such meeting.

Section 4. Quorum. Except as otherwise provided by law or by the Certificate of Incorporation, the holders of a majority of the capital stock issued and outstanding and entitled to vote thereat, present in person or represented by proxy, shall constitute a quorum at all meetings of the stockholders for the transaction of business. If, however, such quorum shall not be present or represented at any meeting of the stockholders, the stockholders entitled to vote thereat, present in person or represented by proxy, may adjourn the meeting from time to time, but no other business shall be transacted at the meeting. Any business may be transacted at the adjourned meeting which might have been transacted at the original meeting. The stockholders present at a duly called or convened meeting, at which a quorum is present, may continue to transact business until adjournment, notwithstanding the withdrawal of enough stockholders to leave less than a quorum. When a meeting is adjourned to another time or place, if any, notice need not be given of the adjourned meeting if the time and place, if any, thereof, by which stockholders and proxy holders may be deemed to be present in person and vote at such meeting, are announced at the meeting at which the adjournment is taken. If the adjournment is for more than thirty days, or, if after adjournment a new record date is fixed, a notice of the adjourned meeting shall be given to each stockholder of record entitled to vote at the meeting. Once a share is represented for any purpose at a meeting (other than solely to object (1) to holding the meeting or transacting business at the meeting, or (2) (if it is a special meeting) to consideration of a particular matter at the meeting that is not within the purpose or purposes described in the meeting notice), it is deemed present for quorum purposes for the remainder of the meeting and or any adjournment of that meeting unless a new record date is set for the adjourned meeting.

Section 5. Voting. Unless otherwise required by law, the Certificate of Incorporation or these By-Laws, any question brought before any meeting of stockholders shall be decided by the vote of the holders of a majority of the stock represented and entitled to vote thereat. Each

stockholder represented at a meeting of stockholders shall be entitled to cast one vote for each share of the capital stock entitled to vote thereat held by such stockholder. Such votes may be cast in person or by proxy but no proxy shall be voted on or after three years from its date, unless such proxy provides for a longer period. Unless required by statute, or determined to be advisable by the Board of Directors, in its discretion, or the officer of the Corporation presiding at a meeting of stockholders, in his discretion, the vote on any matter need not be by ballot.

Section 6. Consent of Stockholders in Lieu of Meeting. Unless otherwise provided in the Certificate of Incorporation, any action required or permitted to be taken at any Annual or Special Meeting of Stockholders of the Corporation, may be taken without a meeting, without prior notice and without a vote, if a consent in writing, setting forth the action so taken, shall be signed by the holders of outstanding stock having not less than the minimum number of votes that would be necessary to authorize or take such action at a meeting at which all shares entitled to vote thereon were present and voted. Prompt notice of the taking of the corporate action without a meeting by less than unanimous written consent shall be given to those stockholders who have not consented in writing. In the event that the action which is consented to is such as would have required the filing of a certificate under the General Corporation Law, if such action had been voted on by stockholders at a meeting thereof, the Certificate filed shall state, in lieu of any statement concerning any vote of stockholders, that written consent and written notice has been given as provided in this Section 6.

In order that the Corporation may determine the stockholders entitled to consent to corporate action in writing without a meeting, the Board of Directors may fix a record date, which record date shall not precede the date upon which the resolution fixing the record date is adopted by the Board of Directors, and which record date shall not be more than 10 days after the date upon which the resolution fixing the record date is adopted by the Board of Directors. Any stockholder of record seeking to have the stockholders authorize or take corporate action by written consent shall, by written notice to the Secretary, request the Board of Directors to fix a record date. The Board of Directors shall promptly, but in all events within 10 days after the date on which such a request is received, adopt a resolution fixing the record date. If no record date has been fixed by the Board of Directors within 10 days after the date on which such a request is received, the record date for determining stockholders entitled to consent to corporate action in writing without a meeting, when no prior action by the Board of Directors is required by applicable law, shall be the first date on which a signed written consent setting forth the action taken or proposed to be taken is delivered to the Corporation by delivery to its registered office in Delaware, its principal place of business, or an officer or agent of the Corporation having custody of the book in which proceedings of meetings of stockholders are recorded. Delivery to the Corporation's registered office shall be by hand or by certified or registered mail, return receipt requested. If no record date has been fixed by the Board of Directors and prior action by the Board of Directors is required by applicable law, the record date for determining stockholders entitled to consent to corporate action in writing without a meeting shall be at the close of business on the date on which the Board of Directors adopts the resolution taking such prior action.

Section 7. List of Stockholders Entitled to Vote. The officer of the Corporation who has charge of the stock ledger of the Corporation shall prepare and make, at least ten days before every meeting of stockholders, a complete list of the stockholders entitled to vote at the meeting,

arranged in alphabetical order, and showing the address of each stockholder and the number of shares registered in the name of each stockholder. Such list shall be open to the examination of any stockholder, for any purpose germane to the meeting, during ordinary business hours, for a period of at least ten days prior to the meeting, in any manner permitted by statute. The list shall also be produced and kept at the time and place of the meeting during the whole time thereof, and may be inspected by any stockholder of the Corporation who is present.

Section 8. Stock Ledger. The stock ledger of the Corporation shall be the only evidence as to who are the stockholders entitled to examine the stock ledger, the list required by Section 7 of this Article I or the books of the Corporation, or to vote in person or by proxy at any meeting of stockholders, or to consent in writing to any action pursuant to Section 6 of this Article I.

Section 9. Notice of Stockholder Business other than Director Nominations. In order for business to be properly brought before an Annual Meeting by a stockholder, the stockholder must have given timely notice thereof in writing to the Secretary of the Corporation. To be timely, a stockholder's notice with respect to an Annual Meeting, other than with respect to nominations of persons for election to the Board of Directors of the Corporation pursuant to Article II, Section 1 or Section 13, of these By-Laws, must be delivered to or mailed and received at the principal executive offices of the Corporation not less than sixty nor more than one hundred twenty days prior to the anniversary date of the preceding year's Annual Meeting; provided, however, that in the event no Annual Meeting was held in the previous year or the date of the Annual Meeting has been changed by more than thirty days, notice by the stockholder to be timely must be so received not later than the close of business on the later of one hundred and twenty days in advance of such Annual Meeting or ten days following the date on which public announcement of the date of the meeting is first made. In no event shall the public announcement of an adjournment or postponement of an Annual Meeting commence a new time period (or extend any time period) for the giving of a notice as described above. For all business other than director nominations, a stockholder's notice to the secretary of the Corporation shall set forth as to each matter the stockholder proposes to bring before the Annual Meeting: (i) a brief description of the business desired to be brought before the Annual Meeting and the reasons for conducting such business at the Annual Meeting, (ii) any other information relating to such stockholder and beneficial owner, if any, on whose behalf the proposal is being made, required to be disclosed in a proxy statement or other filings required to be made in connection with solicitations of proxies for the proposal and pursuant to and in accordance with Section 14(a) of the Exchange Act and the rules and regulations promulgated thereunder and (iii) as to the stockholder: (A) the name and address of the stockholder as they appear on the Corporation's books and of the beneficial owner, if any, on whose behalf the proposal is being made, (B) the class and number of shares of the Corporation which are owned by the stockholder (beneficially and of record) and owned by the beneficial owner, if any, on whose behalf the proposal is being made, as of the date of the stockholder's notice, (C) a description of any agreement, arrangement or understanding with respect to such proposal between or among the stockholder and any of its affiliates or associates, and any others (including their names) acting in concert with any of the foregoing, (D) a description of any agreement, arrangement or understanding (including any derivative or short positions, profit interests, options, hedging transactions, and borrowed or loaned shares) that has been entered into as of the date of the stockholder's notice by, or on behalf of, the stockholder or any of its affiliates or associates, the effect or intent of which is to mitigate loss to, manage risk or benefit of share price changes for, or

increase or decrease the voting power of the stockholder or any of its affiliates or associates with respect to shares of stock of the Corporation, (E) a representation that the stockholder is a holder of record of shares of the Corporation entitled to vote at the meeting and intends to appear in person or by proxy at the meeting to present the proposal specified in the notice, and (F) a representation whether the stockholder intends to solicit proxies from stockholders in support of the proposal. The foregoing notice requirements of Section 9 shall be deemed satisfied by a stockholder with respect to business other than a nomination if the stockholder has notified the Corporation of his, her or its intention to present a proposal at an annual meeting in compliance with the applicable rules and regulations promulgated under Section 14(a) of the Exchange Act and such stockholder's proposal has been included in a proxy statement that has been prepared by the Corporation to solicit proxies for such annual meeting.

No business shall be conducted at any annual meeting except in accordance with the procedures set forth in this Section 9 and Article II, Section 1 and Section 13 of these By-Laws, and unless otherwise required by law, if a stockholder intending to propose business at an annual meeting pursuant to this Section 9 does not provide the information required under this Section 9 to the Corporation promptly following the later of the record date or the date notice of the record date is first publicly disclosed, or the stockholder (or a qualified representative of the stockholder) does not appear at the meeting to present the proposed business, such business shall not be considered, notwithstanding that proxies in respect of such business may have been received by the Corporation. The requirements of this Section 9 shall apply to any business to be brought before an annual meeting by a stockholder whether such business is to be included in the Corporation's proxy statement pursuant to Rule 14a-8 of the Exchange Act or presented to stockholders by means of an independently financed proxy solicitation. The requirements of this Section 9 are included to provide the Corporation notice of a stockholder's intention to bring business before an annual meeting and shall in no event be construed as imposing upon any stockholder the requirement to seek approval from the Corporation as a condition precedent to bringing any such business before an annual meeting.

ARTICLE II DIRECTORS

Section 1. Number and Election of Directors. The Board of Directors shall consist of not less than one nor more than fifteen members, the exact number of which shall be fixed from time to time by the Board of Directors. Except as otherwise required by these By-Laws, by law or by the Certificate of Incorporation, directors shall be elected by a majority of the votes cast at an Annual Meeting of Stockholders at which a quorum is present, provided that directors shall be elected by the vote of a plurality at a meeting at which a quorum is present if (i) the Secretary receives notice that a stockholder has nominated a person for election to the Board of Directors in accordance with the advance notice requirements for stockholder nominations set forth in this Section 1 and (ii) such nomination has not been withdrawn by such stockholders. Each director so elected shall hold office until the next Annual Meeting and until his successor is duly elected and qualified, or until his earlier death, resignation or removal, in the manner hereinafter provided. Any director may resign at any time upon notice to the Corporation. Directors need not be stockholders. For purposes of this Section, a majority of the votes cast means

that the number of shares voted "for" a director must exceed 50% of the votes cast with respect to that director.

Nominations for election to the Board of Directors at an annual or special meeting of the stockholders may be made by the Board of Directors or on behalf of the Board of Directors by a nominating committee duly appointed by the Board of Directors, or by a stockholder of the Corporation entitled to vote for the election of directors. Except as set forth in Article II, Section 13, all nominations, other than those made by or on behalf of the Board of Directors, shall be made by notice in writing delivered or mailed by first class United States mail, postage prepaid, to the Secretary and received by the Secretary not less than sixty nor more than one hundred twenty days prior to the anniversary date of the preceding year's Annual Meeting, in the case of nominations for election at an Annual Meeting, and not more than ten days after the date of the Corporation's notice of a special meeting, in the case of nominations for election at a special meeting; provided, however, that in the event that the Annual Meeting is called for a date that is more than thirty days before or after the anniversary of the preceding year's Annual Meeting, in order to be timely the notice must be so received not later than the close of business on the later of one hundred and twenty days in advance of such Annual Meeting or ten days following the day on which public disclosure of the date of the Annual Meeting was made. In no event shall the public announcement of an adjournment or postponement of an Annual Meeting commence a new time period (or extend any time period) for the giving of a notice as described above. Such stockholder's notice shall set forth as to each proposed nominee who is not an incumbent director, (a) the name, age, business address and, if known, residence address of such nominee, (b) the principal occupation or employment of such nominee during the preceding five years, (c) the number of shares of stock of the Corporation which are beneficially owned by such nominee, (d) any other information relating to such nominee that would be required to be set forth in a definitive proxy statement filed in connection with a proxy solicitation pursuant to Section 14 of the Securities Exchange Act of 1934 ("the 1934 Act"), (e) the written consent of such nominee to being named in the Corporation's proxy statement as a nominee and to serving as a director of the Corporation, if elected; and such stockholder's notice shall set forth as to such stockholder the name and address, as they appear on the Corporation's books, of such stockholder, the number of shares of stock of the Corporation which are beneficially owned by such stockholder, and all other information relating to such stockholder that would be required to be filed with the Securities and Exchange Commission if such stockholder were a participant in a proxy solicitation pursuant to said Section 14. A nomination made otherwise than as provided in this Section 1 or in Article II, Section 13 shall be null and void and shall not be submitted to a vote of stockholders.

Section 2. Vacancies. Vacancies and newly created directorships resulting from any increase in the authorized number of directors may be filled by a majority of the directors then in office, though less than a quorum, or by a sole remaining director, and the directors so chosen shall hold office until the next annual election and until their successors are duly elected and qualified, or until their earlier resignation or removal.

Section 3. Duties and Powers. The business of the Corporation shall be managed by or under the direction of the Board of Directors which may exercise all such powers of the Corporation and do all such lawful acts and things as are not by statute or by the Certificate of Incorporation or by these By-Laws directed or required to be exercised or done by the stockholders.

Section 4. Meetings. The Board of Directors of the Corporation may hold meetings, both regular and special, either within or without the State of Delaware. Regular meetings of the Board of Directors may be held without notice at such time and at such place as may from time to time be determined by the Board of Directors. Special meetings of the Board of Directors may be called by the Chairman, if there be one, the Vice Chairman, if there be one, the President, or any three or more directors. Notice thereof stating the place, date and hour of the meeting shall be given to each director in any manner permitted by statute. Unless otherwise required by these By-Laws, the notice need not state the purpose or purposes of the meeting. Notice need not be given to any director who, either before or after the meeting, signs and submits a written waiver of notice, submits a waiver of notice by electronic transmission, or attends the meeting (except when he attends for the express purpose of objecting, at the beginning of the meeting, to the transaction of any business because the meeting is not lawfully called or convened).

Section 5. Quorum. Except as may be otherwise specifically provided by law, the Certificate of Incorporation or these By-Laws, at all meetings of the Board of Directors, a majority of the entire Board of Directors shall constitute a quorum for the transaction of business and the act of a majority of the entire Board of Directors at any meeting at which there is a quorum shall be the act of the Board of Directors. If a quorum shall not be present at any meeting of the Board of Directors, the directors present thereat may adjourn the meeting from time to time, without notice other than announcement at the meeting, until a quorum shall be present.

Section 6. Actions of Board. Unless otherwise provided by the Certificate of Incorporation or these By-Laws, any action required or permitted to be taken at any meeting of the Board of Directors or of any committee thereof may be taken without a meeting, if all the members of the Board of Directors or committee, as the case may be, consent thereto in writing or by electronic transmission, and the writings or writing or electronic transmission or transmissions are filed with the minutes of proceedings of the Board of Directors or committee. Such filing shall be in paper form if the minutes are maintained in paper form and shall be in electronic form if the minutes are maintained in electronic form.

Section 7. Meetings by Means of Conference Telephone. Unless otherwise provided by the Certificate of Incorporation or these By-Laws, members of the Board of Directors of the Corporation, or any committee designated by the Board of Directors, may participate in a meeting of the Board of Directors or such committee by means of a conference telephone or other communications equipment by means of which all persons participating in the meeting can hear each other, and participation in a meeting pursuant to this Section 7 shall constitute presence in person at such meeting.

Section 8. Committees. The Board of Directors may, by resolution passed by a majority of the entire Board of Directors, designate one or more committees, each committee to consist of one or more of the directors of the Corporation. The Board of Directors may designate one or more directors as alternate members of any committee, who may replace any absent or disqualified member at any meeting of any such committee. In the absence or disqualification of a member of a committee, and in the absence of a designation by the Board of Directors of an alternate member to replace the absent or disqualified member, the member or members thereof present at any meeting and not disqualified from voting, whether or not he or they constitute a quorum, may

unanimously appoint another member of the Board of Directors to act at the meeting in the place of any absent or disqualified member. Any committee, to the extent allowed by law and provided in the resolution establishing such committee, shall have and may exercise all the powers and authority of the Board of Directors in the management of the business and affairs of the Corporation. Each committee shall keep regular minutes and report to the Board of Directors when required. Unless otherwise provided in the resolution of the Board of Directors designating a committee, a committee may create one or more subcommittees, each subcommittee to consist of one or more members of the committee, and may delegate to a subcommittee such powers and authority as the committee deems appropriate.

Section 9. Compensation. The directors may be paid their expenses, if any, of attendance at each meeting of the Board of Directors and may be paid a sum, in cash, securities or a combination thereof for attendance at each meeting of the Board of Directors or a stated salary as director. No such payment shall preclude any director from serving the Corporation in any other capacity and receiving compensation therefor. Members of special or standing committees may be allowed like compensation for attending committee meetings. Compensation of directors shall be as determined by the Board of Directors.

Section 10. Interested Directors. No contract or transaction between the Corporation and one or more of its directors or officers, or between the Corporation and any other corporation, partnership, association or other organization in which one or more of its directors or officers are directors or officers, or have a financial interest, shall be void or voidable solely for this reason, or solely because the director or officer is present at or participates in the meeting of the Board of Directors or committee thereof which authorizes the contract or transaction, or solely because his or their votes are counted for such purpose if (i) the material facts as to his or their relationship or interest and as to the contract or transaction are disclosed or are known to the Board of Directors or the committee, and the Board of Directors, even though the disinterested directors be less than a quorum; or (ii) the material facts as to his or their relationship or interest and as to the contract or transactions are disclosed or are known to the stockholders entitled to vote thereon, and the contract or transaction is specifically approved in good faith by vote of the stockholders; or (iii) the contract or transaction is fair as to the Corporation as of the time it is authorized, approved or ratified, by the Board of Directors, a committee thereof or the stockholders. Common or interested directors may be counted in determining the presence of a quorum at a meeting of the Board of Directors or of a committee which authorizes the contract or transaction.

Section 11. Chairman of the Board of Directors. The Chairman of the Board of Directors, if there be one, shall preside at all meetings of the stockholders and of the Board of Directors. The Chairman of the Board of Directors shall also perform such other duties and may exercise such other powers as from time to time may be assigned to him by these By-Laws or by the Board of Directors.

Section 12. Vice Chairman. The Vice Chairman of the Board of Directors, if there be one, or the Vice Chairmen, if there be more than one, shall perform such duties and may exercise such other powers as from time to time may be assigned by these By-Laws or the Board of Directors.

In the absence or disability of the Chairman of the Board of Directors, or if there be none, the Vice Chairman shall preside at meetings of the stockholders and the Board of Directors.

Section 13. Proxy Access. (a) The Corporation shall include in its proxy statement for an Annual Meeting of stockholders the name, together with the Required Information (as defined below), of any person nominated for election (a "Stockholder Nominee") to the Board of Directors by a stockholder that satisfies, or by a group of no more than twenty stockholders that satisfy, the requirements of this Section 13 (an "Eligible Stockholder"), and that expressly elects at the time of providing the notice required by this Section 13 (the "Nomination Notice") to have its nominee included in the Corporation's proxy materials pursuant to this Section 13.

(b) To be timely, a stockholder's Nomination Notice must be delivered given, either by personal delivery or mailed by United States certified mail, postage prepaid, and received by the Secretary at the principal executive offices of the Corporation not earlier than the close of business on the one hundred fiftieth day prior to, nor later than the close of business on the one hundred twentieth day prior to, the first anniversary of the date of the Corporation's proxy statement released to stockholders in connection with the preceding year's Annual Meeting; provided, however, that in the event that the Annual Meeting is called for a date that is more than thirty days before or seventy days after the anniversary of the preceding year's Annual Meeting, in order to be timely the Nomination Notice must be so received not later than the close of business on the later of one hundred and twenty days in advance of such Annual Meeting or ten days following the day on which public disclosure of the date of the Annual Meeting was made. In no event shall the public announcement of an adjournment or postponement of an Annual Meeting commence a new time period (or extend any time period) for the giving of a Nomination Notice as described above.

(c) For purposes of this Section 13, the "Required Information" that the Corporation will include in its proxy statement is (i) the information concerning the Stockholder Nominee and the Eligible Stockholder that is required to be disclosed in the Corporation's proxy statement by the regulations promulgated under the 1934 Act; and (ii) if the Eligible Stockholder so elects, a Statement (as defined in Section 13(g)). To be timely, the Required Information must be delivered to or mailed to and received by the Secretary within the time period specified in this Section 13 for providing the Nomination Notice.

(d) The number of Stockholder Nominees (including Stockholder Nominees that were submitted by an Eligible Stockholder for inclusion in the Corporation's proxy materials pursuant to this Section 13 but either are subsequently withdrawn or that the Board of Directors decides to nominate as Board of Director nominees), together with any nominees who were previously elected to the Board of Directors as Stockholder Nominees at any of the preceding two Annual Meetings and who are re-nominated for election at such Annual Meeting by the Board of Directors, appearing in the Corporation's proxy materials with respect to an Annual Meeting of stockholders shall not exceed the greater of (i) two or (ii) twenty percent (20%) of the number of directors in office as of the last day on which a Nomination Notice may be delivered pursuant to this Section 13, or if such amount is not a whole number, the closest whole number below twenty percent (20%). In the event that the number of Stockholder Nominees submitted by Eligible Stockholders pursuant to this Section 13 exceeds this maximum number, each Eligible Stockholder will select

one Stockholder Nominee for inclusion in the Corporation's proxy materials until the maximum number is reached, going in order of the amount (largest to smallest) of shares of the capital stock of the Corporation each Eligible Stockholder disclosed as owned in its respective Nomination Notice submitted to the Corporation and confirmed by the Corporation. If the maximum number is not reached after each Eligible Stockholder has selected one Stockholder Nominee, this selection process will continue as many times as necessary, following the same order each time, until the maximum number is reached.

(e) For purposes of this Section 13, an Eligible Stockholder shall be deemed to "own" only those outstanding shares of the capital stock of the Corporation as to which the stockholder possesses both (i) the full voting and investment rights pertaining to the shares and (ii) the full economic interest in (including the opportunity for profit and risk of loss on) such shares; provided that the number of shares calculated in accordance with clauses (i) and (ii) shall not include any shares (x) sold by such stockholder or any of its affiliates in any transaction that has not been settled or closed, (y) borrowed by such stockholder or any of its affiliates for any purposes or purchased by such stockholder or any of its affiliates pursuant to an agreement to resell or (z) subject to any option, warrant, forward contract, swap, contract of sale, or other derivative or similar agreement entered into by such stockholder or any of its affiliates, whether any such instrument or agreement is to be settled with shares or with cash based on the notional amount or value of shares of outstanding capital stock of the Corporation, in any such case which instrument or agreement has, or is intended to have, the purpose or effect of (1) reducing in any manner, to any extent or at any time in the future, such stockholder's or its affiliates' full right to vote or direct the voting of any such shares, and/or (2) hedging, offsetting or altering to any degree gain or loss arising from the full economic ownership of such shares by such stockholder or affiliate. A stockholder shall "own" shares held in the name of a nominee or other intermediary so long as the stockholder retains the right to instruct how the shares are voted with respect to the election of directors and possesses the full economic interest in the shares. A person's ownership of shares shall be deemed to continue during any period in which (i) the person has loaned such shares. provided that the person has the power to recall such loaned shares on no more than five business days' notice; or (ii) the person has delegated any voting power by means of a proxy, power of attorney or other instrument or arrangement that is revocable at any time by the person. The terms "owned," "owning" and other variations of the word "own" shall have correlative meanings. Whether outstanding shares of the capital stock of the Corporation are "owned" for these purposes shall be determined by the Board of Directors, which determination shall be conclusive and binding on the Corporation and its stockholders.

(f) An Eligible Stockholder must have owned (as defined above) continuously for at least three years that number of shares of capital stock as shall constitute three percent (3%) or more of the outstanding capital stock of the Corporation (the "Required Shares") as of both (i) a date within seven days prior to the date of the Nomination Notice and (ii) the record date for determining stockholders entitled to vote at the Annual Meeting. For purposes of satisfying the foregoing ownership requirement under this Section 13, (i) the shares of the capital stock of the Corporation and on whose behalf any stockholder is acting, may be aggregated, provided that the number of stockholders and other persons whose ownership of shares of capital stock of the Corporation is aggregated for such purpose shall not exceed twenty, and (ii) a group

of funds under common management and investment control shall be treated as one stockholder or person for this purpose. No person may be a member of more than one group of persons constituting an Eligible Stockholder under this Section 13. For the avoidance of doubt, if a group of stockholders aggregates ownership of shares in order to meet the requirements under this Section 13, all shares held by each stockholder constituting their contribution to the foregoing 3% threshold must be held by that stockholder continuously for at least three years, and evidence of such continuous ownership shall be provided as specified in this Section 13(f).

Within the time period specified in this Section 13 for providing the Nomination Notice, an Eligible Stockholder must provide the following information in writing to the Secretary of the Corporation:

- (i) one or more written statements from the record holder of the shares (and from each intermediary through which the shares are or have been held during the requisite three-year holding period) verifying that, as of a date within seven days prior to the date of the Nomination Notice, the Eligible Stockholder owns, and has owned continuously for the preceding three years, the Required Shares, and the Eligible Stockholder's agreement to provide, within five business days after the record date for the Annual Meeting, written statements from the record holder and intermediaries verifying the Eligible Stockholder's continuous ownership of the Required Shares through the record date;
- (ii) the written consent of each Stockholder Nominee to being named in the proxy statement as a nominee and to serving as a director if elected;
- (iii) a copy of the Schedule 14N that has been filed with the Securities and Exchange Commission as required by Rule 14a-18 under the 1934 Act, as such rule may be amended;
- (iv) a description of all direct and indirect compensation and other material monetary agreements, arrangements, and understandings during the past three years, and any other material relationships, between or among the Eligible Stockholder and its affiliates and associates, or others acting in concert therewith, on the one hand, and each Stockholder Nominee, and each Stockholder Nominee's respective affiliates and associates, or others acting in concert therewith, on the other hand, including, without limitation all information that would be required to be disclosed pursuant to Item 404 of Regulation S-K if the Eligible Stockholder making the nomination or on whose behalf the nomination is made, or any affiliate or associate thereof or person acting in concert therewith, were the "registrant" for purposes of Item 404 and the nominee were a director or executive officer of such registrant;
- (v) a description of any agreement, arrangement or understanding (including any derivative or short positions, profit interests, options, warrants, stock appreciation or similar rights, hedging transactions, and borrowed or loaned shares) that has been entered into as of the date of the stockholder's notice by, or on behalf of, the Eligible Stockholder, the effect or intent of which is to mitigate loss, manage risk or benefit

from share price change for, or maintain, increase or decrease the voting power of, such Eligible Stockholder with respect to shares of stock of the corporation, and a representation that the Eligible Stockholder will notify the corporation in writing of any such agreement, arrangement or understanding in effect as of the record date for the meeting promptly following the later of the record date or the date notice of the record date is first publicly disclosed;

- (vi) a representation that the Eligible Stockholder (including each member of any group of stockholders that together is an Eligible Stockholder under this Section 13) (A) acquired the Required Shares in the ordinary course of business and not with the intent to change or influence control at the Corporation, and does not presently have such intent, (B) intends to appear in person or by proxy at the Annual Meeting to present the nomination, (C) has not nominated and will not nominate for election to the Board of Directors at the Annual Meeting any person other than the Stockholder Nominee(s) being nominated pursuant to this Section 13, (D) has not engaged and will not engage in, and has not and will not be a "participant" in another person's, "solicitation" within the meaning of Rule 14a-1(l) under the 1934 Act in support of the election of any individual as a director at the Annual Meeting other than its Stockholder Nominee or a nominee of the Board of Directors, (E) will not distribute to any stockholder any form of proxy for the Annual Meeting other than the form distributed by the Corporation and (F) in the case of a nomination by a group of stockholders that together is an Eligible Stockholder, the designation by all group members of one group member that is authorized to act on behalf of all such members with respect to the nomination and matters related thereto, including any withdrawal of the nomination; and
- (vii) an undertaking that the Eligible Stockholder agrees to (A) own the Required Shares through the date of the Annual Meeting, (B) assume all liability stemming from any legal or regulatory violation arising out of the Eligible Stockholder's communications with the stockholders of the Corporation or out of the information that the Eligible Stockholder provided to the Corporation, (C) indemnify and hold harmless the Corporation and each of its directors, officers and employees individually against any liability, loss or damages in connection with any threatened or pending action, suit or proceeding, whether legal, administrative or investigative, against the Corporation or any of its directors, officers or employees arising out of any nomination, solicitation or other activity by the Eligible Stockholder in connection with its efforts to elect the Stockholder Nominee pursuant to this Section 13, (D) comply with all other laws and regulations applicable to any solicitation in connection with the Annual Meeting and (E) provide to the Corporation prior to the Annual Meeting such additional information as necessary with respect thereto.

(g) The Eligible Stockholder may provide to the Secretary of the Corporation, at the time the information required by this Section 13 is provided, a written statement for inclusion in the Corporation's proxy statement for the Annual Meeting, not to exceed five hundred words, in support of the Stockholder Nominee's candidacy (the "Statement"). Notwithstanding anything to the contrary contained in this Section 13, the Corporation may omit from its proxy materials any

information or Statement (or portion thereof) that it, in good faith, believes would violate any applicable law or regulation.

(h) Within the time period specified in this Section 13 for delivering the Nomination Notice, a Stockholder Nominee must deliver to the Secretary of the Corporation a written representation and agreement that the Stockholder Nominee (i) is not and will not become a party to any agreement, arrangement or understanding with, and has not given any commitment or assurance to, any person or entity as to how such person, if elected as a director of the Corporation, will act or vote on any issue or question (ii) is not and will not become a party to any agreement, arrangement or understanding with any person or entity other than the Corporation with respect to any direct or indirect compensation, reimbursement or indemnification in connection with service or action as a director, and (iii) will comply with all of the Corporation's corporate governance, conflict of interest, confidentiality and stock ownership and trading policies and guidelines, and any other Corporation policies and guidelines applicable to directors, as well as any applicable law, rule or regulation or listing requirement. At the request of the Corporation, the Stockholder Nominee must submit all completed and signed questionnaires required of the Corporation's directors and officers. The Corporation may request such additional information as necessary to permit the Board of Directors to determine if each Stockholder Nominee is independent under the listing standards of the principal U.S. exchange upon which the Corporation's capital stock is listed, any applicable rules of the Securities and Exchange Commission and any publicly disclosed standards used by the Board of Directors in determining and disclosing the independence of the Corporation's directors (the "Applicable Independence Standards").

(i) Any Stockholder Nominee who is included in the Corporation's proxy materials for a particular Annual Meeting of stockholders but either (i) withdraws from or becomes ineligible or unavailable for election at the Annual Meeting, or (ii) does not receive at least twenty-five percent (25%) of the votes cast "for" the Stockholder Nominee's election, will be ineligible to be a Stockholder Nominee pursuant to this Section 13 for the next two (2) Annual Meetings.

(j) The Corporation shall not be required to include, pursuant to this Section 13, any Stockholder Nominees in its proxy materials for any meeting of stockholders (i) for which the Secretary of Corporation receives a notice that a stockholder has nominated a person for election to the Board of Directors pursuant to the advance notice requirements for stockholder nominees for director set forth in Section 1 of this Article II and such stockholder does not expressly elect at the time of providing the notice to have its nominee included in the Corporation's proxy materials pursuant to this Section 13, (ii) if the Eligible Stockholder who has nominated such Stockholder Nominee has engaged in or is currently engaged in, or has been or is a "participant" in another person's, "solicitation" within the meaning of Rule 14a-1(1) under the 1934 Act in support of the election of any individual as a director at the meeting other than its Stockholder Nominee(s) or a nominee of the Board of Directors, (iii) who is not independent under the Applicable Independence Standards, as determined by the Board of Directors, (iv) whose election as a member of the Board of Directors would cause the Corporation to be in violation of these By-Laws, the Certificate of Incorporation, the listing standards of the principal exchange upon which the Corporation's capital stock is traded, or any applicable law, rule or regulation, (v) who is or has been, within the past three years, an officer or director of a competitor, as defined in Section 8 of the Clayton Antitrust Act of 1914, as determined by the Board of Directors, (vi) who is a named subject of a pending

criminal proceeding (excluding traffic violations and other minor offenses) or has been convicted in such a criminal proceeding within the past ten years, (vii) who is subject to any order of the type specified in Rule 506(d) of Regulation D promulgated under the Securities Act of 1933, as amended, (viii) if such Stockholder Nominee or the applicable Eligible Stockholder shall have provided information to the Corporation in respect to such nomination that was untrue in any material respect or omitted to state a material fact necessary in order to make the statement made, in light of the circumstances under which it was made, not misleading, as determined by the Board of Directors, or (ix) if the Eligible Stockholder or applicable Stockholder Nominee or fails to comply with its obligations pursuant to this Section 13.

(k) Notwithstanding anything to the contrary set forth herein, the Board of Directors or the person presiding at the meeting shall declare a nomination by an Eligible Stockholder to be invalid, and such nomination shall be disregarded notwithstanding that proxies in respect of such vote may have been received by the Corporation, if (i) the Stockholder Nominee(s) and/or the applicable Eligible Stockholder shall have breached its or their obligations, agreements or representations under this Section 13, as determined by the Board of Directors or the person presiding at the Annual Meeting of stockholders, or (ii) the Eligible Stockholder (or a qualified representative thereof) does not appear at the Annual Meeting of stockholders to present any nomination pursuant to this Section 13.

(1) The Eligible Stockholder (including any person who owns shares of capital stock of the Corporation that constitute part of the Eligible Stockholder's ownership for purposes of satisfying Section 13(f) hereof) shall file with the Securities and Exchange Commission any solicitation or other communication with the Corporation's stockholders relating to the meeting at which the Stockholder Nominee will be nominated, regardless of whether any such filing is required under Regulation 14A of the 1934 Act or whether any exemption from filing is available for such solicitation or other communication under Regulation 14A of the 1934 Act.

ARTICLE III OFFICERS

Section 1. General. The officers of the Corporation shall be chosen by the Board of Directors and shall be a President and a Secretary. The Board of directors, in its discretion, may also choose a Treasurer and one or more Executive Vice Presidents, Senior Vice Presidents, Vice Presidents, Assistant Secretaries, Assistant Treasurers and other officers. Any number of offices may be held by the same person, unless otherwise prohibited by law, the Certificate of Incorporation or these By-Laws. The officers of the Corporation need not be stockholders of the Corporation nor need such officers be directors of the Corporation.

Section 2. Election. The Board of Directors at its first meeting held after each Annual Meeting of Stockholders shall elect the officers of the Corporation who shall hold their offices for such terms and shall exercise such powers and perform such duties as shall be determined from time to time by the Board of Directors; and all officers of the Corporation shall hold office until their successors are chosen and qualified, or until their earlier resignation or removal. Any officer elected by the Board of Directors may be removed at any time by the affirmative vote of a majority

of the entire Board of Directors. Any vacancy occurring in any office of the Corporation shall be filled by the Board of Directors. The salaries of all executive officers of the Corporation shall be fixed by the Board of Directors.

Section 3. Voting Securities Owned by the Corporation. Powers of attorney, proxies, waivers of notice of meeting, consents and other instruments relating to securities owned by the Corporation may be executed in the name and on behalf of the Corporation by any officer of the Corporation and any such officer may, in the name of and on behalf of the Corporation, take all such action as any such officer may deem advisable to vote in person or by proxy at any meeting of security holders of any corporation in which the Corporation may own securities and at any such meeting shall possess and may exercise any and all rights and power incident to the ownership of such securities and which, as the owner thereof, the Corporation might have exercised and possessed if present. The Board of Directors may, by resolution, from time to time confer like powers upon any other person or persons.

Section 4. President. The President shall, subject to the control of the Board of Directors, be the Chief Executive Officer of the Corporation and shall have general supervision of the business of the Corporation and shall see that all orders and resolutions of the Board of Directors are carried into effect. He shall execute all bonds, mortgages, contracts and other instruments of the Corporation requiring a seal, under the seal of the Corporation, except where required or permitted by law to be otherwise signed and executed and except that the other officers of the Corporation may sign and execute documents when so authorized by these By-Laws, the Board of Directors or the President. In the absence or disability of the Chairman and the Vice Chairman of the Board of Directors, or if there be none, the President shall preside at all meetings of the stockholders and the Board of Directors. The President shall also perform such other duties and may exercise such other powers as from time to time may be assigned to him by these By-Laws or by the Board of Directors.

Section 5. Executive Vice Presidents/Senior Vice Presidents. At the request of the President or in his absence or in the event of his inability or refusal to act (and if there be no Chairman of the Board of Directors or Vice Chairman of the Board of Directors), the Executive Vice President or Senior Vice Presidents if there is more than one (in the order designated by the Board of Directors) shall perform the duties of the President, and when so acting, shall have all the powers of and be subject to all the restrictions upon the President. Each Executive Vice President or Senior Vice President shall perform such other duties and have such other powers as the Board of Directors from time to time may prescribe. If there be no Chairman of the Board of Directors, no Vice Chairman of the Board and no Executive Vice President or Senior Vice President, the Board of Directors shall designate the officer of the Corporation who, in the absence of the President or in the event of the inability or refusal of the President to act, shall perform the duties of the President, and when so acting, shall have all the powers of and be subject to all the restrictions upon the President.

Section 6. Secretary. The Secretary shall attend all meetings of the Board of Directors and all meetings of stockholders and record all the proceedings thereat in a book or books to be kept for that purpose; the Secretary shall also perform like duties for the standing committees when required. The Secretary shall give, or cause to be given, notice of all meetings of the stockholders

and special meetings of the Board of Directors, and shall perform such other duties as may be prescribed by the Board of Directors, the Chairman or Vice Chairman of the Board of Directors or President, under whose supervision he shall be. If the Secretary shall be unable or shall refuse to cause to be given notice of all meetings of the stockholders and special meetings of the Board of Directors, and if there be no Assistant Secretary, then either the Board of Directors, the Chairman or Vice Chairman of the Board of Directors or the President may choose another officer to cause such notice to be given. The Secretary shall have custody of the seal of the Corporation and the Secretary or any Assistant Secretary, if there be one, shall have authority to affix the same to any instrument requiring it and when so affixed, it may be attested by the signature of the Secretary or by the signature of any such Assistant Secretary. The Board of Directors may give general authority to any other officer to affix the seal of the Corporation and to attest the affixing by his signature. The Secretary shall see that all books, reports, statements, certificates and other documents and records required by law to be kept or filed are properly kept or filed, as the case may be, in any manner permitted by statute and as directed by the Board of Directors.

Section 7. Treasurer. The Treasurer, if there be one, shall have the custody of the corporate funds and securities and shall keep full and accurate accounts of receipts and disbursements in books belonging to the Corporation and shall deposit all moneys and other valuable effects in the name and to the credit of the Corporation in such depositories as may be designated by the Board of Directors. The Treasurer shall disburse the funds of the Corporation as may be ordered by the Board of Directors, taking proper vouchers for such disbursements, and shall render to the Chairman or Vice Chairman of the Board of Directors or to the President and the Board of Directors, at its regular meetings, or when the Board of Directors so requires, an account of all his transactions as Treasurer and of the financial condition of the Corporation. If required by the Board of Directors, the Treasurer shall give the Corporation a bond in such sum and with such surety or sureties as shall be satisfactory to the Board of Directors for the faithful performance of the duties of his office and for the restoration to the Corporation, in case of his death, resignation, retirement or removal from office, of all books, papers, vouchers, money and other property of whatever kind in his possession or under his control belonging to the Corporation.

Section 8. Assistant Secretaries. Except as may be otherwise provided in these By-Laws, Assistant Secretaries, if there be any, shall perform such duties and have such powers as from time to time may be assigned to them by the Board of Directors, the Chairman or Vice Chairman of the Board of Directors, the President, any Vice President, if there be one, or the Secretary, and in the absence of the Secretary or in the event of his disability or refusal to act, shall perform the duties of the Secretary, and when so acting, shall have all the powers of and be subject to all the restrictions upon the Secretary.

Section 9. Assistant Treasurers. Assistant Treasurers, if there be any, shall perform such duties and have such powers as from time to time may be assigned to them by the Board of Directors, the Chairman or Vice Chairman of the Board of Directors, the President, any Vice President, if there be one, or the Treasurer, and in the absence of the Treasurer or in the event of his disability or refusal to act, shall perform the duties of the Treasurer, and when so acting, shall have all the powers of and be subject to all the restrictions upon the Treasurer. If required by the Board of Directors, an Assistant Treasurer shall give the Corporation a bond in such sum and with such surety or sureties as shall be satisfactory to the Board of Directors for the faithful performance

of the duties of his office and for the restoration to the Corporation, in case of his death, resignation, retirement or removal from office, of all books, papers, vouchers, money and other property of whatever kind in his possession or under his control belonging to the Corporation.

Section 10. Other Officers. Such other officers as the Board of Directors may choose shall perform such duties and have such powers as from time to time may be assigned to them by the Board of Directors. The Board of Directors may delegate to any other officer of the Corporation the power to choose such other officers and to prescribe their respective duties and powers.

ARTICLE IV STOCK

Section 1. Form of Certificates. The shares of the Corporation shall be represented by certificates, unless the Board of Directors provides by resolution or resolutions that some or all of any or all classes or series of the Corporation's stock shall be uncertificated shares. Any such resolution shall not apply to shares represented by a certificate until the certificate is surrendered to the Corporation. Every holder of stock in the Corporation represented by certificates shall be entitled to have a certificate signed, in the name of the Corporation (i) by the Chairman or the Vice Chairman of the Board of Directors, the President or a Vice President and (ii) by the Treasurer or an Assistant Treasurer, or the Secretary or an Assistant Secretary of the Corporation, representing the number of shares registered in certificate form. The Board of Directors may make such additional rules and regulations, not inconsistent with these By-Laws, as it may deem appropriate concerning the issue, transfer and registration of certificates for shares of stock or uncertificated shares of the Corporation.

Section 2. Signatures. Any or all of the signatures on certificate may be a facsimile. In case any officer, transfer agent or registrar who has signed or whose facsimile signature has been placed upon a certificate shall have ceased to be such officer, transfer agent or registrar before such certificate is issued, it may be issued by the Corporation with the same effect as if he were such officer, transfer agent or registrar at the date of issue.

Section 3. Lost Certificates. The Board of Directors may direct a new certificate to be issued in place of any certificate theretofore issued by the Corporation alleged to have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the person claiming the certificate of stock to be lost, stolen or destroyed. When authorizing such issue of a new certificate, the Board of Directors may, in its discretion and as a condition precedent to the issuance thereof, require the owner of such alleged lost, stolen or destroyed certificate, or his legal representative, to advertise the same in such manner as the Board of Directors shall require and/or to give the Corporation a bond in such sum as it may direct as indemnity against any claim that may be made against the Corporation on account of the alleged lost, stolen or destroyed certificate or the issuance of the new certificate or uncertificated shares. The Corporation may adopt such other provisions and restrictions with reference to lost certificates, not inconsistent with applicable law, as it shall in its discretion deem appropriate.

Section 4. Transfers. Stock of the Corporation shall be transferable in the manner prescribed by law and in these By-Laws. Transfers of stock shall be made on the books of the Corporation only by the person named in the certificate or by his attorney lawfully constituted in writing and upon the surrender of the certificate therefor, which shall be canceled before a new certificate or uncertificated shares shall be issued.

Section 5. Record Date. In order that the Corporation may determine the stockholders entitled to notice of or to vote at any meeting of stockholders or any adjournment thereof, or entitled to receive payment of any dividend or other distribution or allotment of any rights, or entitled to exercise any rights in respect of any change, conversion or exchange of stock, or for the purpose of any other lawful action, the Board of Directors may fix, in advance, a record date, which shall not be more than sixty days nor less than ten days before the date of such meeting, nor more than sixty days prior to any other action. A determination of stockholders of record entitled to notice of or to vote at a meeting of stockholders shall apply to any adjournment of the meeting; provided, however, that the Board of Directors may fix a new record date for the adjourned meeting. The manner of fixing a record date for the determination of stockholders entitled to express consent to corporate action in writing without a meeting shall be as provided for in Article I, Section 6.

Section 6. Beneficial Owners. The Corporation shall be entitled to recognize the exclusive right of a person registered on its books as the owner of shares to receive dividends, and to vote as such owner, and to hold liable for calls and assessments a person registered on its books as the owner of shares, and shall not be bound to recognize any equitable or other claim to or interest in such share or shares on the part of any other person, whether or not it shall have express or other notice thereof, except as otherwise provided by law.

ARTICLE V NOTICES

Section 1. Notices. Whenever notice is required by law, the Certificate of Incorporation or these By-Laws, to be given to any director, member of a committee or stockholder, such notice may be given in any manner permitted by applicable laws and regulations, and shall be deemed given at the time prescribed by applicable laws and regulations for such manner of notice. Notice of any meeting shall not be required to be given to any person who attends the meeting, except when the person attends the meeting in person or by proxy for the express purpose of objecting, at the beginning of the meeting, to the transaction of any business because the meeting is not lawfully called or convened.

Section 2. Waivers of Notice. Whenever any notice is required by law, the Certificate of Incorporation or these By-Laws, to be given to any director, member of a committee or stockholder, a waiver thereof in writing, signed by the person or persons entitled to said notice, or an electronically transmitted waiver of notice, whether before or after the time stated therein, shall be deemed equivalent thereto.

ARTICLE VI GENERAL PROVISIONS

Section 1. Dividends. Dividends upon the capital stock of the Corporation, subject to the provisions of the Certificate of Incorporation, if any, may be declared by the Board of Directors at any regular or special meeting, and may be paid in cash, in property, or in shares of the capital stock. Before payment of any dividend, there may be set aside out of any funds of the Corporation available for dividends such sum or sums as the Board of Directors from time to time, in its absolute discretion, deems proper as a reserve or reserves to meet contingencies, or for equalizing dividends, or for repairing or maintaining any property of the Corporation, or for any proper purpose, and the Board of Directors may modify or abolish any such reserve.

Section 2. Disbursements. All checks or demands for money and notes of the Corporation shall be signed by such officer or officers or such other person or persons as the Board of Directors may from time to time designate.

Section 3. Fiscal Year. The fiscal year of the Corporation shall be fixed by resolution of the Board of Directors.

Section 4. Corporate Seal. The corporate seal shall have inscribed thereon the name of the Corporation, the year of its organization and the words "Corporate Seal, Delaware". The seal may be used by causing it or a facsimile thereof to be impressed or affixed or reproduced or otherwise.

Section 5. Section 203 Election. The Corporation hereby expressly elects not to be governed by Section 203 of the General Corporation Law of the State of Delaware.

Section 6. Electronic Transmission. When used in these By-Laws, "electronic transmission" means any form of communication, not directly involving the physical transmission of paper, that creates a record that may be retained, retrieved and reviewed by a recipient thereof, and that may be directly reproduced in paper form by such a recipient through an automated process, including without limitation any telegram, cablegram, facsimile transmission and communication by electronic mail.

Section 7. Facsimile Signatures. In addition to the provisions for use of facsimile signatures specifically authorized in these By-Laws, facsimile signatures of any officer or officers of the Corporation may be used whenever and as authorized by the Board of Directors or any committee thereof.

Section 8. Form of Records. Any records required by these By-Laws or otherwise maintained by the Corporation in the regular course of its business, including its stock ledger, books of account, and minute books, may be kept on, or be in the form of, magnetic tape, computer diskettes and discs, photographs, microphotographs, or any other information storage device, provided that the records so kept can be converted into clearly legible form within a reasonable time.

ARTICLE VII INDEMNIFICATION

Section 1. Power to Indemnify in Actions, Suits or Proceedings Other Than Those by or in the Right of the Corporation. Subject to Section 3 of this Article VII, the Corporation shall indemnify any person who was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative (other than an action by or in the right of the Corporation) by reason of the fact that he is or was a director, officer, employee or agent of the Corporation, or is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust, employee benefit plan or other enterprise, against expenses (including attorneys' fees), judgments, fines and amounts paid in settlement actually and reasonably incurred by him in connection with such action, suit or proceeding if he acted in good faith and in a manner he reasonably believed to be in or not opposed to the best interests of the Corporation, and, with respect to any criminal action or proceeding, had no reasonable cause to believe his conduct was unlawful. The termination of any action, suit or proceeding by judgment, order, settlement, conviction, or upon a plea of nolo contendere or its equivalent, shall not, of itself, create a presumption that the person did not act in good faith and in a manner which he reasonably believed to be in or not opposed to the best interests of the Corporation, shall not, of itself, create a presumption that the person did not act in good faith and in a manner which he reasonably believed to be in or not opposed to the best interests of the Corporation, shall not, of itself, create a presumption that the person did not act in good faith and in a manner which he reasonably believed to be in or not opposed to the best interests of the Corporation, and, with respect to any criminal action or proceeding, had reasonable cause to believe that his conduct was unlawful.

Section 2. Power to Indemnify in Actions, Suits or Proceedings by or in the Right of the Corporation. Subject to Section 3 of this Article VII, the Corporation shall indemnify any person who was or is a party or is threatened to be made a party to any threatened, pending or completed action or suit by or in the right of the Corporation to procure a judgment in its favor by reason of the fact that he is or was a director, officer, employee or agent of the Corporation, or is or was serving at the request of the corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust, employee benefit plan or other enterprise against expenses (including attorneys' fees) actually and reasonably incurred by him in connection with the defense or settlement of such action or suit if he acted in good faith and in a manner he reasonably believed to be in or not opposed to the best interests of the Corporation; except that no indemnification shall be made in respect of any claim, issue or matter as to which such person shall have been adjudged to be liable to the Corporation unless and only to the extent that the Court of Chancery or the court in which such action or suit was brought shall determine upon application that, despite the adjudication of liability but in view of all the circumstances of the case, such person is fairly and reasonably entitled to indemnity for such expenses which the Court of Chancery or such other court shall deem proper.

Section 3. Authorization of Indemnification. Any indemnification under this Article VII (unless ordered by a court) shall be made by the Corporation only as authorized in the specific case upon a determination that indemnification of the director, officer, employee or agent is proper in the circumstances because he has met the applicable standard of conduct set forth in Section 1 or Section 2 of this Article VII, as the case may be. Such determination shall be made (i) by the Board of Directors by a majority vote of a quorum consisting of directors who were not parties to such action, suit or proceeding, or (ii) if such a quorum is not obtainable, or, even if obtainable a quorum of disinterested directors so directs, by independent legal counsel in a written opinion, or (iii) by the stockholders. To the extent, however, that a director, officer, employee or agent of the Corporation has been successful on the merits or otherwise in defense of any action, suit or

proceeding described above, or in defense of any claim, issue or matter therein, he shall be indemnified against expenses (including attorneys' fees) actually and reasonably incurred by him in connection therewith, without the necessity of authorization in the specific case.

Section 4. Good Faith Defined. For purposes of any determination under Section 3 of this Article VII, a person shall be deemed to have acted in good faith and in a manner he reasonably believed to be in or not opposed to the best interests of the Corporation, or, with respect to any criminal action or proceeding, to have had no reasonable cause to believe his conduct was unlawful, if his action is based on the records or books of account of the Corporation or another enterprise, or on information supplied to him by the officers of the Corporation or another enterprise in the course of their duties, or on the advice of legal counsel for the Corporation or another enterprise or on information or records given or reports made to the Corporation or another enterprise by an independent certified public accountant or by an appraiser or other expert selected with reasonable care by the Corporation or another enterprise. The term "another enterprise" as used in this Section 4 shall mean any other corporation or any partnership, joint venture, trust, employee benefit plan or other enterprise of which such person is or was serving at the request of the Corporation as a director, officer, employee or agent. The provisions of this Section 4 shall not be deemed to be exclusive or to limit in any way the circumstances in which a person may be deemed to have met the applicable standard of conduct set forth in Section 1 or Section 2 of this Article VII, as the case may be.

Section 5. Indemnification by a Court. Notwithstanding any contrary determination in the specific case under Section 3 of this Article VII, and notwithstanding the absence of any determination thereunder, any director, officer, employee or agent may apply to any court of competent jurisdiction in the State of Delaware for indemnification to the extent otherwise permissible under Section 1 or Section 2 of this Article VII. The basis of such indemnification by a court shall be a determination by such court that indemnification of the director, officer, employee or agent is proper in the circumstances because he has met the applicable standards of conduct set forth in Section 1 or Section 2 of this Article VII nor the absence of any determination thereunder shall be a defense to such application or create a presumption that the director, officer, employee or agent seeking indemnification has not met any applicable standard of conduct. Notice of any application for indemnification pursuant to this Section 5 shall be given to the Corporation promptly upon the filing of such application. If successful, in whole or in part, the director, officer, employee or agent seeking indemnification shall also be entitled to be paid the expense of prosecuting such application.

Section 6. Expenses Payable in Advance. Expenses incurred in defending or investigating a threatened or pending action, suit or proceeding shall be paid by the Corporation in advance of the final disposition of such action, suit or proceeding upon receipt of an undertaking by or on behalf of such director, officer, employee or agent to repay such amount if it shall ultimately be determined that he is not entitled to be indemnified by the Corporation as authorized in this Article VII.

Section 7. Non-exclusivity of Indemnification and Advancement of Expenses. The indemnification and advancement of expenses provided by or granted pursuant to this Article VII

shall not be deemed exclusive of any other rights to which those seeking indemnification or advancement of expenses may be entitled under any By-Law, agreement, contract, vote of stockholders or disinterested directors or pursuant to the direction (howsoever embodied) of any court of competent jurisdiction or otherwise, both as to action in his official capacity and as to action in another capacity while holding such office, it being the policy of the Corporation that indemnification of the persons specified in Section 1 or Section 2 of this Article VII shall be made to the fullest extent permitted by law. The provisions of this Article VII shall not be deemed to preclude the indemnification of any person who is not specified in Section 1 or Section 2 of this Article VII shall be more or obligation to indemnify under the provisions of the General Corporation Law of the State of Delaware, or otherwise.

Section 8. Insurance. The Corporation may purchase and maintain insurance on behalf of any person who is or was a director, officer, employee or agent of the Corporation, or is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust, employee benefit plan or other enterprise against any liability asserted against him and incurred by him in any such capacity, or arising out of his status as such, whether or not the Corporation would have the power or the obligation to indemnify him against such liability under the provisions of this Article VII.

Section 9. Certain Definitions. For purposes of this Article VII, references to "the Corporation" shall include, in addition to the resulting corporation, any constituent corporation (including any constituent of a constituent) absorbed in a consolidation or merger which, if its separate existence had continued, would have had power and authority to indemnify its director, officers, employees or agents, so that any person who is or was a director, officer, employee or agent of such constituent corporation, or is or was serving at the request of such constituent corporation as a directors, officer, employee or agent of another corporation, partnership, joint venture, trust, employee benefit plan or other enterprise, shall stand in the same position under the provisions of this Article VII with respect to the resulting or surviving corporation as he would have with respect to such constituent corporation if its separate existence had continued. For purposes of this Article VII, references to "fines" shall include any excise taxes assessed on a person with respect to an employee benefit plan; and references to "serving at the request of the Corporation" shall include any service as a director, officer, employee or agent of the Corporation" shall include any services by, such director, officer, employee or agent with respect to an employee or agent with respect to be in the interest of the participants and beneficiaries of an employee benefit plan shall be deemed to have acted in a manner "not opposed to the best interests of the Corporation" as referred to in this Article VII.

Section 10. Survival of Indemnification and Advancement of Expenses. The indemnification and advancement of expenses provided by, or granted pursuant to, this Article VII shall, unless otherwise provided when authorized or ratified, continue as to a person who has ceased to be a director, officer, employee or agent and shall inure to the benefit of the heirs, executors and administrators of such a person.

Section 11. Limitation on Indemnification. Notwithstanding anything contained in this Article VII to the contrary, except for proceedings to enforce rights to indemnification (which shall

be governed by Section 5 hereof), the Corporation shall not be obligated to indemnify any director, officer, employee or agent in connection with a proceeding (or part thereof) initiated by such person unless such proceeding (or part thereof) was authorized or consented to by the Board of Directors of the Corporation.

ARTICLE VIII AMENDMENTS

Section 1. Amendments. These By-Laws may be altered, amended or repealed, in whole or in part, or new By-Laws may be adopted by the stockholders or by the Board of Directors, provided, however, that notice of such alteration, amendment, repeal or adoption of new By-Laws be contained in the notice of such meeting of stockholders or Board of Directors, as the case may be. All such amendments must be approved by either the holders of a majority of the outstanding capital stock entitled to vote thereon or by a majority of the entire Board of Directors then in office.

Section 2. Entire Board of Directors. As used in this Article VIII and in these By-Laws generally, the term "entire Board of Directors" means the total number of directors which the Corporation would have if there were no vacancies.

CERTIFICATE OF MERGER

OF

RADIANCE MERGER SUB INC. (a Delaware corporation)

WITH AND INTO

LABORATORY CORPORATION OF AMERICA HOLDINGS (a Delaware corporation)

Pursuant to Section 251 of the General Corporation Law of the State of Delaware (the "DGCL"), the undersigned corporation does hereby certify the following information in connection with the merger (the "Merger") of Radiance Merger Sub Inc., a Delaware corporation ("Merger Sub"), with and into Laboratory Corporation of America Holdings, a Delaware corporation ("Corporation"):

FIRST: The name and state of incorporation of each of the constituent corporations in the Merger (the "Constituent Corporations") are as follows:

NameState of IncorporationRadiance Merger Sub Inc.DelawareLaboratory Corporation of America HoldingsDelaware

- **SECOND:** The Agreement and Plan of Merger, dated as of May 17, 2024 (the "Merger Agreement"), by and among Labcorp Holdings Inc., a corporation organized under the laws of the State of Delaware, Merger Sub and the Corporation, which expressly provides for the Merger to be effected under Section 251(g) of the DGCL, has been approved, adopted, executed and acknowledged by each of the constituent corporations in accordance with Section 251 of the DGCL.
- **THIRD:** The Corporation shall be the surviving corporation in the Merger (the "Surviving Corporation"), and the name of the Surviving Corporation shall be Laboratory Corporation of America Holdings.
- **FOURTH:** The certificate of incorporation of the Corporation, as in effect immediately prior to the Merger, shall be amended and restated in its entirety to read as set forth on <u>Exhibit A</u> attached hereto and, as so amended, shall be the amended and restated certificate of incorporation of the Surviving Corporation until further amended in accordance with its terms and the provisions of the DGCL.
- **FIFTH:** The executed Merger Agreement is on file at an office of the Surviving Corporation at 531 South Spring Street, Burlington, North Carolina, 27215.

- **SIXTH:** A copy of the Merger Agreement will be furnished by the Surviving Corporation, on request and without cost, to any stockholder of any Constituent Corporation.
- SEVENTH: This Certificate of Merger shall become effective at 4:05 p.m., Eastern Time, on May 17, 2024.

[Signature Page Follows]

IN WITNESS WHEREOF, the Surviving Corporation has caused this Certificate of Merger to be signed by an authorized officer, this 17th day of May, 2024.

LABORATORY CORPORATION OF AMERICA HOLDINGS

By: <u>/s/ Sandra D. van der Vaart</u> Name: Sandra D. van der Vaart Title: Executive Vice President, Chief Legal Officer, Chief Compliance Officer and Secretary

EXHIBIT A

AMENDED AND RESTATED CERTIFICATE OF INCORPORATION

OF

LABORATORY CORPORATION OF AMERICA HOLDINGS

FIRST: The name of the Corporation is Laboratory Corporation of America Holdings (hereinafter the "Corporation").

SECOND: The address of the registered office of the Corporation in the State of Delaware is Corporation Trust Center, 1209 Orange Street, in the City of Wilmington, County of New Castle. The name of its registered agent at that address is The Corporation Trust Company.

THIRD: The purpose of the Corporation is to engage in any lawful act or activity for which a corporation may be organized under the General Corporation Law of the State of Delaware as set forth in Title 8 of the Delaware Code (the "GCL").

FOURTH: The total number of shares of stock which the Corporation has authority to issue is one thousand two hundred fifty (1,250) shares of which one thousand (1,000) will be shares of Common Stock, each having a par value of \$0.10, and two hundred fifty (250) will be shares of Preferred Stock, each having a par value of \$0.10.

The Board of Directors is expressly authorized to provide for the issuance of all or any shares of the Preferred Stock in one or more classes or series, and to fix for each such class or series such voting powers, full or limited, or no voting powers, and such distinctive designations, preferences and relative, participating, optional or other special rights and such qualifications, limitations or restrictions thereof, as shall be stated and expressed in the resolution or resolutions adopted by the Board of Directors providing for the issuance of such class or series and as may be permitted by the GCL, including without limitation, the authority to provide that any such class or series may be (i) subject to redemption at such time or times and at such price or prices; (ii) entitled to receive dividends (which may be cumulative or non-cumulative at such rates, on such conditions, and at such times, and payable in preference to or in such relation to, the dividends payable on any other class or classes or to any other series); (iii) entitled to such rights upon the dissolution of, or upon any distribution of the assets of, the Corporation; or (iv) other class or classes of stock, or of any other series of the same or any other class or classes of stock, of the Corporation at such price or prices or at such rates of exchange and with such adjustments; all as may be stated in such resolution or resolutions.

FIFTH: The following provisions are inserted for the management of the business and the conduct of the affairs of the Corporation, and for further definition, limitation and regulation of the powers of the Corporation and of its directors and stockholders:

(1) The business and affairs of the Corporation shall be managed by or under the direction of the Board of Directors.

(2) The directors shall have concurrent power with the stockholders to make, alter, amend, change, add to or repeal the By-Laws of the Corporation.

(3) The number of directors of the corporation shall be as from time to time fixed by, or in the manner provided in, the By-Laws of the Corporation. Election of directors need not be by written ballot unless the By-Laws so provide.

(4) No director shall be personally liable to the Corporation or any of its stockholders for monetary damages for breach of fiduciary duty as a director, except for liability (i) for any breach of the director's duty of loyalty to the Corporation or its stockholders, (ii) for acts or omissions not in good faith or which involve intentional misconduct or a knowing violation of law, (iii) pursuant to Section 174 of the Delaware General Corporation Law or (iv) for any transaction from which the director derived an improper personal benefit. Any repeal or modification of this Article Sixth by the stockholders of the Corporation shall not adversely affect any right or protection of a director of the corporation existing at the time of such repeal or modification with respect to acts or omissions occurring prior to such repeal or modification.

(5) In addition to the powers and authority hereinbefore or by statute expressly conferred upon them, the directors are hereby empowered to exercise all such powers and do all such acts and things as may be exercised or done by the Corporation, subject, nevertheless, to the provisions of the GCL, this Certificate of Incorporation, and any By-Laws adopted by the stockholders; provided, however, that no By-Laws hereafter adopted by the stockholders shall invalidate any prior act of the directors which would have been valid if such By-Laws had not been adopted.

SIXTH: Meetings of stockholders may be held within or without the State of Delaware, as the By-Laws may provide. The books of the Corporation may be kept (subject to any provision contained in the GCL) outside the State of Delaware at such place or places as may be designated from time to time by the Board of Directors or in the By-Laws of the Corporation.

SEVENTH: The Corporation reserves the right to amend, alter, change or repeal any provision contained in this Certificate of Incorporation, in the manner now or hereafter prescribed by statute, and all rights conferred upon stockholders herein are granted subject to this reservation.

EIGHTH: Any act or transaction by or involving the Corporation, other than the election or removal of directors of the Corporation, that requires for its adoption under the GCL or this Certificate of Incorporation the approval of the stockholders of the Corporation shall, in accordance with Section 251(g) of the GCL, require, in addition, the approval of the stockholders of Laboratory Holdings Inc. (or any successor thereto by merger), by the same vote as is required by the GCL and/or this Certificate.

DESCRIPTION OF THE REGISTRANT'S SECURITIES REGISTERED PURSUANT TO SECTION 12 OF THE SECURITIES EXCHANGE ACT OF 1934

Labcorp Holdings Inc. ("we" and "our") has only one class of securities of registered under Section 12 of the Securities Exchange Act of 1934, as amended: our common stock, \$0.10 par value per share ("Common Stock").

DESCRIPTION OF COMMON STOCK

The following description of our common stock summarizes certain material terms and provisions of our certificate of incorporation, by-laws, and the Delaware General Corporation Law ("DGCL"). For the complete terms of our common stock, please refer to our certificate of incorporation and by-laws, which are incorporated by reference as exhibits to the Annual Report on Form 10-K of which this exhibit is a part, and to the applicable provisions of the DGCL.

Authorized Common Stock

We have authority to issue 265 million shares of Common Stock, par value \$0.10 per share.

Rights of Common Stock

Voting Rights; Liquidation; Dividends. Holders of our common stock are entitled:

- to attend all special and annual meetings of the stockholders and to one vote per share upon any matter, including, without limitation, the election of directors, on which stockholders are entitled to vote;
- upon our liquidation, dissolution or winding up, whether voluntary or involuntary, to participate in the distribution of any assets remaining after the payment of all debts and liabilities, subject to any preferential rights of holders of any outstanding shares of preferred stock; and
- to receive dividends, which may be cumulative or non-cumulative, as may be lawfully declared from time to time by our board of directors.

Other Rights and Restrictions. The holders of our common stock do not have any preemptive or subscription rights to purchase additional securities issued by us, nor any rights to convert their common stock into other of our securities or to have their shares of common stock redeemed by us. Our common stock is not subject to redemption by us. Our certificate of incorporation and by-laws do not restrict the ability of a holder of common stock to transfer his or her shares of common stock. Our by-laws provide that holders of our common stock may act by written consent on any matters that could otherwise be brought at annual or special meetings.

Preferred Stock. Our board of directors has the authority, without further action by our stockholders, to issue up to 30 million shares of preferred stock, par value \$0.10 per share, in one or more classes or series and to fix the number of shares, designations, relative rights (including voting, conversion, redemption, and dividend rights), terms of redemption, preferences, and limitations of such series to the full extent now or hereafter permitted by the DGCL.

Anti-Takeover Effects of Our Certificate of Incorporation and By-Law Provisions

Undesignated Preferred Stock. Because the board of directors has the power to establish the preferences and rights of the shares of any additional series of preferred stock, it may afford holders of any preferred stock preferences, powers and rights, including voting and dividend rights, senior to the rights of holders of the common stock, which could adversely affect the holders of the common stock and could discourage a takeover of us even if a change of control of our company would be beneficial to the interests of our stockholders.

Special Stockholder Meetings. Our by-laws provide that a special meeting of stockholders may be called only by a resolution adopted by a majority of our board of directors or by stockholders owning at least 10% of our outstanding common stock, subject to the requirements and procedures as set forth in our by-laws.

Stockholder Advance Notice Procedure. Our by-laws establish an advance notice procedure for stockholders to make nominations of candidates for election as directors or to bring other business before an annual meeting of our stockholders.

Section 203 of the Delaware General Corporation Law. We are subject to Section 203 of the DGCL

("Section 203"), which, with specified exceptions, prohibits a Delaware corporation from engaging in any "business combination" with any "interested stockholder" for a period of three years following the time that the stockholder became an interested stockholder unless:

- prior to that time, the board of directors of the corporation approved either the business combination or the transaction which resulted in the stockholder becoming an interested stockholder;
- upon consummation of the transaction which resulted in the stockholder becoming an interested stockholder, the interested stockholder owned at least 85% of the voting stock of the corporation outstanding at the time the transaction commenced, excluding for purposes of determining the number of shares outstanding those shares owned by persons who are directors and also officers and by employee stock plans in which employee participants do not have the right to determine confidentially whether shares held subject to the plan will be tendered in a tender or exchange offer; or
- at or after that time, the business combination is approved by the board of directors and authorized at an annual or special meeting of stockholders, and not by written consent, by the affirmative vote of at least 66 2/3% of the outstanding voting stock that is not owned by the interested stockholder.

Section 203 defines "business combination" to include the following:

- any merger or consolidation of the corporation with the interested stockholder;
- any sale, lease, exchange, mortgage, transfer, pledge or other disposition of 10% or more of the assets of the corporation involving the interested stockholder;
- subject to specified exceptions, any transaction that results in the issuance or transfer by the corporation of any stock of the corporation to the interested stockholder;
- any transaction involving the corporation that has the effect of increasing the proportionate share of the stock of any class or series of the corporation beneficially owned by the interested stockholder; or
- any receipt by the interested stockholder of the benefit of any loans, advances, guarantees, pledges, or other financial benefits provided by or through the corporation.

In general, Section 203 defines an "interested stockholder" as any entity or person beneficially owning 15% or more of the outstanding voting stock of the corporation and any entity or person affiliated with or controlling or controlled by that entity or person.

The application of Section 203 may make it difficult and expensive for a third party to pursue a takeover attempt that we do not approve, even if a change in control would be beneficial to the interests of our stockholders.



LABORATORY CORPORATION OF AMERICA HOLDINGS

as Issuer

AND

LABCORP HOLDINGS INC. as Guarantor

AND

U.S. BANK TRUST COMPANY, NATIONAL ASSOCIATION

as

Trustee

SEVENTEENTH SUPPLEMENTAL INDENTURE

DATED AS OF MAY 17, 2024

SEVENTEENTH SUPPLEMENTAL INDENTURE (this "<u>Seventeenth Supplemental Indenture</u>"), dated as of May 17, 2024, among Labcorp Holdings Inc., a Delaware corporation (or its permitted successor) ("<u>New Holding Company</u>"), its whollyowned subsidiary, Laboratory Corporation of America Holdings, a Delaware corporation (or its permitted successor) (the "<u>Company</u>"), and U.S. Bank Trust Company, National Association, as Trustee under the Indenture (the "<u>Trustee</u>").

WITNESSETH

WHEREAS, the Company has heretofore executed and delivered to the Trustee an indenture (the "<u>Base Indenture</u>"), dated as of November 19, 2010, as amended by the Ninth Supplemental Indenture, dated as of January 30, 2015 (the "<u>Ninth Supplemental Indenture</u>"), the Tenth Supplemental Indenture, dated as of January 30, 2015 (the "<u>Tenth Supplemental Indenture</u>"), the Eleventh Supplemental Indenture, dated as of August 22, 2017 (the "<u>Eleventh Supplemental Indenture</u>"), the Twelfth Supplemental Indenture, dated as of August 22, 2017 (the "<u>Tenth Supplemental Indenture</u>"), the Twelfth Supplemental Indenture, dated as of November 25, 2019 (the "<u>Thirteenth Supplemental Indenture</u>"), the Thirteenth Supplemental Indenture, dated as of May 26, 2021 (the "<u>Fifteenth Supplemental Indenture</u>"), the Fifteenth Supplemental Indenture, dated as of May 26, 2021 (the "<u>Fifteenth Supplemental Indenture</u>"), and the Sixteenth Supplemental Indenture, dated as of May 26, 2021 (the "<u>Fifteenth Supplemental Indenture</u>"), and the Sixteenth Supplemental Indenture, the Tenth Supplemental Indenture, the Tenth Supplemental Indenture, the Twelfth Supplemental Indenture, the Tenth Supplemental Indenture, the Thirteenth Supplemental Indenture, the Thirteenth Supplemental Indenture, the Tenth Supplemental Indenture, the Twelfth Supplemental Indenture, the Thirteenth Supplemental Indenture, the Tenth

WHEREAS, New Holding Company desires to fully and unconditionally guarantee all of the Company's obligations under the Indenture and the Securities, including the due and punctual payment of the principal of, premium, if any, and interest on all the Securities according to their tenor (the "<u>New Holding Company Guarantee</u>"); and

WHEREAS, the Company desires, by this Seventeenth Supplemental Indenture, to add the New Holding Company Guarantee and make certain other conforming changes to Section 4.08 of the Base Indenture; and

WHEREAS, Sections 9.01(4) and 9.05 of the Base Indenture provide, in relevant part, that the Company, when authorized by a Board Resolution, and the Trustee, may enter into a supplemental indenture without the consent of the Securityholders "to add guarantees with respect to any series of Securities;" and

WHEREAS, Sections 9.01(5) and 9.05 of the Base Indenture provide, in relevant part, that the Company, when authorized by a Board Resolution, and the Trustee, may enter into a supplemental indenture without the consent of the Securityholders "to make any change that does not adversely affect the rights of any Securityholder in any material respect;"

WHEREAS, the Company has provided the Trustee with an Opinion of Counsel and an Officers' Certificate pursuant to Section 9.05 of the Base Indenture; and

WHEREAS, all acts and things prescribed by the Indenture, by law and by the Certificate of Incorporation and By-Laws (or comparable constituent documents) of New Holding Company, the Company and the Trustee necessary to make this Seventeenth Supplemental Indenture a valid instrument legally binding on New Holding Company, the Company and the Trustee, in accordance with its terms, have been legally done and performed.

NOW THEREFORE, to comply with the provisions of the Indenture and in consideration of the foregoing and for good and valuable consideration, the receipt of which is hereby acknowledged, New Holding Company, the Company and the Trustee mutually covenant and agree for the equal and proportionate benefit of the respective Securityholders of the Securities as follows:

Article 1

Relation To Indenture; Definitions; Rules Of Construction

Section 1.1. *Relation to Indenture*. This Seventeenth Supplemental Indenture is supplemental to the Indenture and does and shall be deemed to form a part of, and shall be construed in connection with and as part of, the Indenture for any and all purposes.

Section 1.2. *The New Holding Company Guarantee.* New Holding Company hereby fully and unconditionally guarantees (in the form attached hereto as Exhibit A) to each Securityholder and to the Trustee and its successors and assigns the full and punctual payment of principal of, premium, if any, and interest on the Securities when due, whether at maturity, by acceleration, by redemption or otherwise, and all other monetary obligations of the Company under the Indenture and the Securities and the full and punctual performance of all other obligations of the Company under the Indenture and the Securities.

Section 1.3. *Amendment to Section 4.08 of the Indenture*. Section 4.08 (Commission Reports) of the Base Indenture is hereby amended by adding the following language at the end of the first sentence of the existing paragraph:

"; *provided* that the filing and provision to the Trustee and Securityholders of the annual reports and other information, documents and other reports specified in Sections 13 or 15(d) of the Exchange Act, at the times specified for the filing of such information, documents and reports under such Sections, by an entity that is the direct or indirect parent of the Company shall satisfy the requirements of this Section 4.08, in lieu of any such reporting by the Company, so long as (x) such entity is an obligor on, or full and unconditional guarantor of, the Securities and (y) the information, documents and reports of such parent entity include all financial statements and other information regarding the Company as may be required by Rule 3-10 of Regulation S-X or other applicable rule of the Commission to be included in reports filed by a parent company that guarantees debt issued by a subsidiary that is thereby exempt from the reporting obligations under Sections 13 and 15(d) of the Exchange Act, and such other quantitative or qualitative

information regarding the Company that would be material to Securityholders and would have been filed by the Company under Sections 13 and 15(d) of the Exchange Act but for such exemption."

Section 1.4. *Amendment to Section 12.04 of the Indenture*. Section 12.04 (Notices) of the Base Indenture is hereby amended by adding the following language after the existing paragraph:

"All notices, approvals, consents, requests and any communications hereunder must be in writing (provided that any communication sent to the Trustee hereunder must be in the form of a document that is signed manually or by way of a digital signature provided by DocuSign or AdobeSign (or such other digital signature provider as specified in writing to the Trustee by the authorized representative), in English. The Company agrees to assume all risks arising out of the use of using digital signatures and electronic methods to submit communications to the Trustee, including without limitation the risk of the Trustee acting on unauthorized instructions, and the risk of interception and misuse by third parties.

The Company may send notices and other communications hereunder by electronic mail. The Company agrees that the Trustee cannot determine the identity of the actual sender of such communications and that the Trustee shall conclusively presume that communications that purport to have been sent by an authorized officer of the Company have been sent by such authorized officer. The Company shall be responsible for ensuring that only authorized officers transmit such communications to the Trustee, and the Company and the authorized officers are responsible for safeguarding the use and confidentiality of applicable user and authorization codes, passwords and authentication keys provided by the Trustee. The Trustee shall not be liable for any losses, costs, or expenses arising directly or indirectly from the Trustee's reliance upon and compliance with such communications notwithstanding such communications conflict or are inconsistent with a subsequent written communication. The Company agrees (i) to assume all risks arising out of the use of electronic mail to submit communications to the Trustee, including without limitation the risk of the Trustee acting on unauthorized instructions and the risk or interception and misuse by third parties; (ii) that it is fully informed of the protections and risks associated with the various methods of transmitting communications to the Trustee and that there may be more secure methods of transmitting communications than the method(s) selected by the Company; (iii) that the security procedures (if any) to be followed in connection with its transmission of communications provide to it a commercially reasonable degree of protection in light of its particular needs and circumstances; and (iv) that it will notify the Trustee immediately upon learning of any compromise or unauthorized use of the security procedures."

Section 1.5. *Amendment to Section 12.10 of the Indenture*. Section 12.10 (Counterparts) of the Base Indenture is hereby amended by adding the following language at the end of the first sentence of the existing paragraph:

"Delivery of an executed signature page of this Indenture by facsimile or any other rapid transmission device designed to produce a written record of the communication transmitted shall be as effective as delivery of a manually executed counterpart thereof. The words "execution," "executed," "signed," signature" and words of like import in this Indenture or in any other certificate, agreement or document related to this Indenture shall include images of manually executed signatures transmitted by facsimile, email or other electronic format (including, without limitation, "pdf," "tif" or "jpg") and other electronic signatures (including, without limitation, DocuSign and AdobeSign). The use of electronic signatures and electronic records (including, without limitation, any contract or other record created, generated, sent, communicated, received, or stored by electronic means) shall be of the same legal effect, validity and enforceability as a manually executed signature or use of a paper-based record-keeping system to the fullest extent permitted by applicable law, including the Federal Electronic Signatures in Global and National Commerce Act and any other applicable law, including, without limitation, any state law based on the Uniform Electronic Transactions Act or the Uniform Commercial Code."

Section 1.6. Notices. The address for notices to New Holding Company shall be:

Labcorp Holdings Inc. 358 South Main Street Burlington, North Carolina 27215 Attention: Executive Vice President and Chief Financial Officer

Section 1.7. *Capitalized Terms*. Capitalized terms used herein but not defined shall have the meanings assigned to them in the Base Indenture.

Article 2

Miscellaneous

Section 2.1. *Successors and Assigns*. All of the covenants, stipulations, promises and agreements in this Seventeenth Supplemental Indenture contained by or on the behalf of the Company shall bind its successors and assigns, whether so expressed or not.

Section 2.2. *Effectiveness*. This Seventeenth Supplemental Indenture shall become effective upon its execution and delivery.

Section 2.3. *Ratification of Indenture*. The Indenture, as supplemented by this Seventeenth Supplemental Indenture, is in all respects ratified and confirmed, and this Seventeenth Supplemental Indenture shall be deemed part of the Indenture in the manner and to the extent herein and therein provided.

Section 2.4. *Governing Law*. This Seventeenth Supplemental Indenture shall be deemed to be a contract made under the internal laws of the State of New York, and for all purposes shall be construed in accordance with the laws of said State, but without giving effect to applicable principles of conflicts of law to the extent that the application of the laws of another jurisdiction would be required thereby.

Section 2.5. *Multiple Originals*. The parties may sign any number of copies of this Seventeenth Supplemental Indenture. Each signed copy shall be an original, but all of them together represent the same agreement. One signed copy is enough to prove this Seventeenth Supplemental Indenture.

Section 2.6. *Headings*. The headings of the Sections of this Seventeenth Supplemental Indenture have been inserted for convenience of reference only, are not intended to be considered a part hereof and shall not modify or restrict any of the terms or provisions hereof.

IN WITNESS WHEREOF, the parties have caused this Seventeenth Supplemental Indenture to be duly executed as of the date first written above.

LABORATORY CORPORATION OF AMERICA HOLDINGS

By: /s/ Glenn A. Eisenberg

Name: Glenn A. Eisenberg Title: Executive Vice President and Chief Financial Officer

LABCORP HOLDINGS INC.

By: /s/ Glenn A. Eisenberg

Name: Glenn A. Eisenberg Title: Executive Vice President and Chief Financial Officer

U.S. BANK TRUST COMPANY, NATIONAL ASSOCIATION, Trustee

By: /s/ Allison Lancaster-Poole Name: Allison Lancaster-Poole Title: Vice President

[Signature Page to Seventeenth Supplemental Indenture]

2.30% Senior Notes due 2024 (issued under the Thirteenth Supplemental Indenture)
3.25% Senior Notes due 2024 (issued under the Eleventh Supplemental Indenture)
3.60% Senior Notes due 2025 (issued under the Ninth Supplemental Indenture)
1.55% Senior Notes due 2026 (issued under the Fifteenth Supplemental Indenture)
3.60% Senior Notes due 2027 (issued under the Twelfth Supplemental Indenture)
2.95% Senior Notes due 2029 (issued under the Fourteenth Supplemental Indenture)
2.70% Senior Notes due 2031 (issued under the Sixteenth Supplemental Indenture)
4.70% Senior Notes due 2045 (issued under the Tenth Supplemental Indenture)

FORM OF NEW HOLDING COMPANY GUARANTEE

From and after the date hereof, Labcorp Holdings Inc., a Delaware corporation ("New Holding Company"), unconditionally and irrevocably guarantees (the "Guarantee") to each Securityholder of a Security authenticated and delivered by the Trustee, and to the Trustee and its successors and assigns, the due and punctual payment of the principal of, premium, if any, and interest on the Securities, subject to any applicable grace period, whether at stated maturity, by acceleration or otherwise, and the due and punctual performance of all other obligations of Laboratory Corporation of America Holdings, a Delaware corporation (the "Company"), to the Securityholders or the Trustee under the Indenture in accordance with the terms hereof and thereof. New Holding Company hereby agrees that its obligations hereunder shall be unconditional, irrespective of the validity or enforceability of the Securities or the obligations of the Company under the Indenture, the absence of any action to enforce the same, the recovery of any judgment against the Company or any other obligor with respect to the Indenture or the Securities, any action to enforce the same or any other circumstances (other than complete performance) which might otherwise constitute a legal or equitable discharge or defense of a guarantor; provided, however, that nothing contained herein shall be constituted to be a waiver by New Holding Company of presentment or demand of payment or notice to New Holding Company with respect to the Securities and the obligations evidenced thereby or hereby. New Holding Company further waives any right of set-off or counterclaim it may have against registered owners of the Securities arising from any other obligations of New Holding Company that any such registered owners may have. New Holding Company covenants that this Guarantee will not be discharged except by complete performance of the obligations contained in the Indenture, the Securities and in this Guarantee.

New Holding Company shall be subrogated to all rights of the registered owners of the Securities in respect of any amounts paid by New Holding Company pursuant to the provisions of this Guarantee; *provided, however*, that New Holding Company shall be entitled to enforce, or to receive any payments arising out of or based upon, such right of subrogation only after the principal of, premium, if any, and interest on the Securities and all other amounts owed to the registered owners of the Securities hereunder have been paid in full.

This Guarantee shall continue to be effective or reinstated, as the case may be, if at any time any payment of the principal of, premium, if any, or interest on the Securities or any other amounts owed to the registered owners of the Securities hereunder or thereunder is rescinded or must otherwise be returned by such registered owners upon the insolvency, bankruptcy or reorganization of New Holding Company, the Company or otherwise, all as though such payment had not been made. Capitalized terms used herein

but not defined shall have the meanings assigned to them in the Seventeenth Supplemental Indenture, of which this Guarantee forms a part.

IN WITNESS WHEREOF, New Holding Company has caused this instrument to be duly executed as of the 17th day of May, 2024.

LABCORP HOLDINGS INC.

By: Name:

Title:

NEW HOLDING COMPANY GUARANTEE

From and after the date hereof, Labcorp Holdings Inc., a Delaware corporation ("New Holding Company"), unconditionally and irrevocably guarantees (the "Guarantee") to each Securityholder of a Security authenticated and delivered by the Trustee, and to the Trustee and its successors and assigns, the due and punctual payment of the principal of, premium, if any, and interest on the Securities, subject to any applicable grace period, whether at stated maturity, by acceleration or otherwise, and the due and punctual performance of all other obligations of Laboratory Corporation of America Holdings, a Delaware corporation (the "Company"), to the Securityholders or the Trustee under the Indenture in accordance with the terms hereof and thereof. New Holding Company hereby agrees that its obligations hereunder shall be unconditional, irrespective of the validity or enforceability of the Securities or the obligations of the Company under the Indenture, the absence of any action to enforce the same, the recovery of any judgment against the Company or any other obligor with respect to the Indenture or the Securities, any action to enforce the same or any other circumstances (other than complete performance) which might otherwise constitute a legal or equitable discharge or defense of a guarantor; provided, however, that nothing contained herein shall be constituted to be a waiver by New Holding Company of presentment or demand of payment or notice to New Holding Company with respect to the Securities and the obligations evidenced thereby or hereby. New Holding Company further waives any right of set-off or counterclaim it may have against registered owners of the Securities arising from any other obligations of New Holding Company that any such registered owners may have. New Holding Company covenants that this Guarantee will not be discharged except by complete performance of the obligations contained in the Indenture, the Securities and in this Guarantee.

New Holding Company shall be subrogated to all rights of the registered owners of the Securities in respect of any amounts paid by New Holding Company pursuant to the provisions of this Guarantee; *provided, however*; that New Holding Company shall be entitled to enforce, or to receive any payments arising out of or based upon, such right of subrogation only after the principal of, premium, if any, and interest on the Securities and all other amounts owed to the registered owners of the Securities hereunder have been paid in full.

This Guarantee shall continue to be effective or reinstated, as the case may be, if at any time any payment of the principal of, premium, if any, or interest on the Securities or any other amounts owed to the registered owners of the Securities hereunder or thereunder is rescinded or must otherwise be returned by such registered owners upon the insolvency, bankruptcy or reorganization of New Holding Company, the Company or otherwise, all as though such payment had not been made. Capitalized terms used herein but not defined shall have the meanings assigned to them in the Seventeenth Supplemental Indenture, of which this Guarantee forms a part.

IN WITNESS WHEREOF, New Holding Company has caused this instrument to be duly executed as of the 17th day of May, 2024.

LABCORP HOLDINGS INC.

By: /s/ Glenn A. Eisenberg

Name:Glenn A. EisenbergTitle:Executive Vice Presidentand Chief Financial Officer

GUARANTOR JOINDER AGREEMENT

May 17, 2024

Reference is made to the Third Amended and Restated Credit Agreement, dated as of April 30, 2021 (as amended, amended and restated, supplemented, waived or otherwise modified from time to time, the "**Credit Agreement**") among the Borrower, the Lenders party thereto and Bank of America, N.A., as Administrative Agent. Terms defined in the Credit Agreement are used herein (this "**Guarantor Joinder Agreement**") with the same meanings.

NOW, THEREFORE, the Administrative Agent and Labcorp Holdings Inc., a Delaware corporation (the "New Guarantor"), in its capacity as a Guarantor, hereby agree as follows:

1. Joinder as Guarantor and as the Parent and the Company. In accordance with Section 6.09(a) and Section 11.23(c) of the Credit Agreement, the New Guarantor by its signature below becomes a Guarantor under the Credit Agreement with the same force and effect as if originally named therein as a Guarantor, but in any event subject to the same terms, provisions and limitations set forth in Article X of the Credit Agreement. The New Guarantor hereby (x) becomes party to the Credit Agreement as the Parent and the Company and assumes the obligations set forth in the Credit Agreement applicable to it in such capacities and (y) agrees to all the terms and provisions of the Credit Agreement applicable to it as a Guarantor and a Loan Party and as the Parent and the Company. Each reference to a Guarantor and a Loan Party and to the Parent and the Company in the Credit Agreement shall be deemed to include the New Guarantor.

2. **Representations and Warranties**. The New Guarantor represents and warrants that each of the representations and warranties set forth in the Credit Agreement and applicable to the New Guarantor is true in all material respects on and as of the date hereof, except to the extent any such representation and warranty (i) expressly relates to an earlier date in which case such representation and warranty shall be true and correct in all material respects as of such earlier date or (ii) is qualified by materiality, in which case such representation and warranty shall be true and correct in all respects.

3. Severability. Any provision of this Guarantor Joinder Agreement which is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof, and any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.

4. **Counterparts**. This Guarantor Joinder Agreement may be executed in counterparts, each of which shall constitute an original. Delivery of an executed counterpart of a signature page to this Guarantor Joinder Agreement by facsimile or other electronic transmission (including ".pdf" or ".tif" files) shall be effective as delivery of an original executed counterpart of this Guarantor Joinder Agreement. The words "execution," "signed," "signature," and words of similar import in this Guarantor Joinder Agreement or any notice, certificate, document, agreement or instrument in respect thereof shall be deemed to include Electronic Signatures, deliveries or the keeping of records in electronic form, each of which shall be of the same legal effect, validity and enforceability as manually executed signatures, physical delivery or the use of a paper-based recordkeeping system, as the case may be, to the extent and as provided for in any applicable law, including the Federal Electronic Signatures in Global and National Commerce Act, the New York State Electronic Signatures and Records Act, or any other similar state laws based on the Uniform Electronic Transactions Act. For purposes hereof, "<u>Electronic Signature</u>" shall have the meaning assigned to it by 15 USC §7006, as it may be amended from time to time.

5. No Waiver. Except as expressly supplemented hereby, the Credit Agreement shall remain in full force and effect.

6. **Notices**. All notices, requests and demands to or upon the New Guarantor, the Administrative Agent or any Lender shall be governed by the terms of Section 11.02 of the Credit Agreement.

7. **Governing Law**. THIS AGREEMENT SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK.

[Signature Pages Follow]

IN WITNESS WHEREOF, the undersigned have caused this Guarantor Joinder Agreement to be duly executed and delivered by their duly authorized officer as of the date first set forth above.

LABCORP HOLDINGS INC., as New Guarantor and as the Parent and the Company

By: /s/ Glenn A. Eisenberg

Name: Glenn A. Eisenberg Title: Executive Vice President and Chief Financial Officer

[Guarantor Joinder Agreement – LabCorp]

BANK OF AMERICA, N.A., as Administrative Agent By: /s/ Aamir Saleem

Name: Aamir Saleem Title: Vice President

[Guarantor Joinder Agreement – LabCorp]

ASSIGNMENT AND ASSUMPTION AGREEMENT

This Assignment and Assumption Agreement (this "<u>Agreement</u>") is entered into this 17th day of May, 2024 by and between Laboratory Corporation of America Holdings ("<u>Labcorp</u>"), Labcorp Holdings Inc. ("<u>LHI</u>") and Adam H. Schechter ("<u>Executive</u>").

RECITALS

WHEREAS, Labcorp and Executive have previously entered into an Executive Employment Agreement as of June 4, 2019 pursuant to which Executive agreed to be employed as the President and Chief Executive Officer of Labcorp (the "<u>Executive</u> <u>Employment Agreement</u>");

WHEREAS, pursuant to Section 15 of the Executive Employment Agreement, the rights and obligations of Labcorp under the Executive Employment Agreement shall be assignable and may be assumed by a successor entity in connection with the formation of a holding company or similar corporate reorganization approved by the Board of Directors of Labcorp

WHEREAS, Labcorp and LHI have entered into a series of transactions effective as of May 17, 2024 to implement a new holding company structure pursuant to which Labcorp will become a direct, wholly owned subsidiary of Labcorp Holdings Inc.;

WHEREAS, in connection with the implementation of the new holding company structure, effective as of May 17, 2024, LHI desires to assume the obligations of Labcorp under the Executive Employment Agreement and Executive desires to consent to and agree to the assignment and assumption of the Executive Employment Agreement from Labcorp to LHI.

NOW, THEREFORE, in consideration of the mutual covenants and agreements set forth herein and for other good and valuable consideration, the receipt and sufficiency of which hereby are acknowledged, the parties hereto agree as follows:

1. Pursuant to Section 15 of the Executive Employment Agreement, Labcorp, LHI and Executive agree to the assignment of the Executive Employment Agreement from Labcorp to LHI and the assumption by LHI of all of the obligations under the Executive Employment Agreement.

2. Consistent with the foregoing, and in accordance with the terms of Section 15 of the Executive Employment Agreement, effective as of the May 17, 2024, the term "Company" as used in the Executive Employment Agreement refers to LHI instead of Labcorp.

3. Except as otherwise provided in this Agreement, all terms of the Executive Employment Agreement shall remain unchanged and in full force and effect.

IN WITNESS WHEREOF, the undersigned have duly executed and delivered this Agreement, or have caused this Agreement to be duly executed and delivered on their behalf.

LABORATORY CORPORATION OF AMERICA HOLDINGS

By: /s/ Sandra D. van der Vaart Name: Sandra D. van der Vaart Title: Executive Vice President, Chief Legal Officer, and Secretary

LABCORP HOLDINGS INC.

By: /s/ Sandra D. van der Vaart Name: Sandra D. van der Vaart Title: Executive Vice President, Chief Legal Officer, and

Secretary

EXECUTIVE By: /s/ Adam H. Schechter

Name: Adam H. Schechter Title: President and Chief Executive Officer

ASSIGNMENT AND ASSUMPTION AGREEMENT

THIS ASSIGNMENT AND ASSUMPTION AGREEMENT (this "<u>Agreement</u>") dated as of May 17, 2024 is between Laboratory Corporation of America Holdings, a Delaware corporation ("<u>Labcorp</u>"), Labcorp Holdings Inc., a Delaware corporation ("<u>LHI</u>"), and Radiance Merger Sub Inc., a Delaware corporation ("<u>Merger Sub</u>"). All capitalized terms used in this Agreement and not defined herein have the respective meanings ascribed to them in the Agreement and Plan of Merger, dated as of May 17, 2024 (the "<u>Merger Agreement</u>"), by and among Labcorp, LHI and Merger Sub.

RECITALS

WHEREAS, pursuant to the Merger Agreement, Merger Sub will merge with and into Labcorp with Labcorp surviving as a direct, wholly owned subsidiary of LHI (the "<u>Merger</u>" and together with the other transactions contemplated by the Merger Agreement, the "<u>Reorganization</u>") and each outstanding share of common stock of Labcorp, par value \$0.10 per share ("<u>Labcorp Stock</u>") will be converted into one share of common stock of LHI, par value \$0.10 per share ("<u>LHI Stock</u>") of the same class and with the same rights and privileges relative to LHI that such share had relative to Labcorp prior to the Merger;

WHEREAS, once the conditions set forth in the Merger Agreement have been satisfied or waived, as the case may be, Labcorp intends to file a certificate of merger executed in accordance with the relevant provisions of the General Corporation Law of the State of Delaware (the "<u>DGCL</u>") (the "<u>Certificate of Merger</u>"), with the Secretary of State of the State of Delaware (the "<u>Secretary of State</u>") and shall make all other filings or recordings required under the DGCL to effectuate the Merger, which shall become effective at such time as the Certificate of Merger is duly filed with the Secretary of State or at such later date and time as is specified in the Certificate of Merger (the date and time the Merger becomes effective being referred to herein as the "<u>Effective Time</u>").

WHEREAS, in connection with the Reorganization, Labcorp will transfer and assign (including sponsorship of) to LHI, and LHI will assume (including sponsorship of), Labcorp's equity compensation plans listed on <u>Exhibit A</u> hereto and any subplans, appendices or addendums thereto, and any and all agreements evidencing Awards (as defined below) (the "Labcorp Equity Compensation Plans");

WHEREAS, in connection with the Reorganization and as a consequence of the transfers and assignments of the Labcorp Equity Compensation Plans, at the Effective Time, Labcorp will transfer and assign to LHI, and LHI will assume and perform all obligations of Labcorp pursuant to (i) each stock option to purchase a share of Labcorp Stock (each, a "<u>Stock Option</u>"), (ii) each right to acquire or vest in a share of Labcorp Stock pursuant to awards of restricted stock units (each, an "<u>RSU</u>") and pursuant to awards of performance shares (each, a "<u>Performance Share</u>"), and (iii) each right to purchase a share of Labcorp Stock under the Laboratory Corporation of America Holdings 2016 Employee Stock Purchase Plan (the "<u>Labcorp ESPP</u>") (each, an "<u>ESPP Right</u>") (the Stock Options, RSUs, Performance Shares and ESPP Rights, collectively, the "<u>Awards</u>") issued under the Labcorp Equity Compensation Plans that is outstanding and unexercised, unvested and not yet paid immediately prior to the Effective Time, which Awards shall be converted into a Stock Option or ESPP Right or an RSU or a Performance Share pursuant to which the holder may acquire a share of LHI Stock with the same rights and privileges relative to LHI that such Stock Option, ESPP Right, RSU or Performance Share had relative to Labcorp immediately prior to the Effective Time on otherwise the same terms and conditions as were applicable immediately prior to the Effective Time, including, for Stock Options, at an exercise price per share equal

to the exercise price per share for the applicable share of Labcorp Stock and for ESPP Rights at a purchase price determined in accordance with the terms of the Labcorp ESPP.

WHEREAS, in connection with the Reorganization, Labcorp will transfer and assign to LHI, and LHI will assume and perform all obligations of Labcorp pursuant to each of the other agreements listed on <u>Exhibit B</u> hereto (the "<u>Other Agreements</u>");

WHEREAS, the Board of Directors of Labcorp has determined that it is in the best interests of Labcorp and its stockholders for Labcorp to enter into this Agreement;

WHEREAS, the Board of Directors of LHI has determined that it is in the best interests of LHI and its stockholders for LHI to enter into this Agreement;

NOW, THEREFORE, for good and valuable consideration, the receipt, adequacy and sufficiency of which are hereby acknowledged, Labcorp and LHI hereby agree as follows:

I. EQUITY PLANS AND AWARDS

1. Subject to and as of the Effective Time, Labcorp transfers and assigns to LHI and LHI assumes and agrees to perform, from and after the Effective Time, all obligations of Labcorp pursuant to the Labcorp Equity Compensation Plans.

2. Subject to and as of the Effective Time, Labcorp transfers and assigns to LHI and LHI assumes and agrees to perform all obligations of Labcorp pursuant to each Award that is outstanding and unexercised or unvested and not yet paid or payable immediately prior to the Effective Time issued under the Labcorp Equity Compensation Plans and each such Award shall be converted into an identical equity award of LHI as follows:

- (a) each RSU and Performance Share shall be converted into a right to acquire or vest in, on otherwise the same terms and conditions as were applicable under the applicable Labcorp Equity Compensation Plan and/or agreement evidencing an Award, a share of LHI Stock with the same rights and privileges applicable to the share of Labcorp Stock subject to such RSU or Performance Share immediately prior to the Effective Time;
- (b) each Stock Option shall be converted into an option to purchase, on otherwise the same terms and conditions as were applicable under the applicable Labcorp Equity Compensation Plan and/or agreement evidencing an Award, a share of LHI Stock with the same rights and privileges applicable to the share of Labcorp Stock subject to such Stock Option immediately prior to the Effective Time, at an exercise price per share equal to the exercise price per share of Labcorp Stock subject to such Stock Option immediately prior to the Effective Time, provided that all Stock Options shall be adjusted and converted in accordance with the requirements of Section 424 of the United States Internal Revenue Code of 1986, as amended, and regulations thereunder; and

(c) each ESPP Right shall be converted into a right to purchase, on otherwise the same terms and conditions as were applicable under the Labcorp ESPP a number of shares of LHI Stock as determined in accordance with the terms of the Labcorp ESPP, provided that for purposes of the Offering Period (as such term is defined in the Labcorp ESPP) in effect as of the Effective Time (the "<u>Current Offering Period</u>"), the purchase price per share of LHI Stock under the Labcorp ESPP will be 85% of the lesser of the average of the high and low sales price of (i) a share of Labcorp Stock on the first trading day of the Current Offering Period or (ii) a share of LHI Stock on the last trading day of Current Offering Period.

3. At the Effective Time, the Awards, the Labcorp Equity Compensation Plans and any provision of any other compensatory plan, agreement or arrangement providing for the grant or issuance of Labcorp Stock shall each be automatically deemed to be amended, to the extent necessary or appropriate, to provide that references to Labcorp in such awards, documents and provisions shall be read to refer to LHI and references to Labcorp Stock in such awards, documents and provisions shall be read to refer to LHI and Labcorp agree to (i) prepare and execute all amendments to the Labcorp Equity Compensation Plans, Awards and other documents necessary to effectuate LHI's assumption of the Labcorp Equity Compensation Plans and outstanding Awards, (ii) provide notice of the assumption to holders of such Awards, and (iii) submit any required filings with the Securities and Exchange Commission in connection with the same.

4. On or prior to the Effective Time, LHI shall reserve sufficient shares of LHI Stock to provide for the issuance of LHI Stock to satisfy LHI's obligations under this Agreement with respect to the Labcorp Equity Compensation Plans and Awards assumed by LHI.

5. Labcorp and LHI agree that the Reorganization does not constitute a "Change in Control" under the Laboratory Corporation of America Holdings 2016 Omnibus Incentive Plan, as such term is defined therein.

II. OTHER AGREEMENTS

1. Subject to and as of the Effective Time, Labcorp transfers and assigns and LHI assumes and agrees to perform, from and after the Effective Time, all of the obligations of Labcorp pursuant to the Other Agreements.

2. At the Effective Time, the Other Agreements shall each be automatically deemed to be amended, to the extent necessary or appropriate, to provide that references to Labcorp in such agreements, documents and provisions shall be read to refer to LHI and references to Labcorp Stock in such agreements, documents and provisions shall be read to refer to LHI and Labcorp agree to) prepare and execute all amendments to the Other Agreements necessary to effectuate LHI's assumption of the Other Agreements.

III. MISCELLANEOUS

1. Each of Labcorp and LHI will, from time to time and at all times hereafter, upon every reasonable request to do so by any other party hereto, make, do, execute and deliver, or cause to be made, done, executed and delivered, all such further acts, deeds, assurances and things as may be reasonably required or necessary in order to further implement and carry out the intent and purpose of this Agreement.

IN WITNESS WHEREOF, the undersigned have duly executed and delivered this Agreement, or have caused this Agreement to be duly executed and delivered on their behalf.

LABORATORY CORPORATION OF AMERICA HOLDINGS

By: /s/ Sandra D. van der Vaart

Name: Sandra D. van der Vaart

Title: Executive Vice President, Chief Legal Officer, Chief Compliance Officer and Secretary

LABCORP HOLDINGS INC.

By: <u>/s/ Sandra D. van der Vaart</u> Name: Sandra D. van der Vaart Title: Executive Vice President, Chief Legal Officer and Secretary

RADIANCE MERGER SUB INC.

By: <u>/s/ Sandra D. van der Vaart</u> Name: Sandra D. van der Vaart Title: President and Secretary

Exhibit A - Labcorp Equity Compensation Plans

Laboratory Corporation of America Holdings 2016 Omnibus Incentive Plan Laboratory Corporation of America Holdings Amended and Restated 2016 Employee Stock Purchase Plan

A -1

Exhibit B - Other Agreements

Executive Employment Agreement, dated June 4, 2019, by and between Laboratory Corporation of America Holdings and Adam H. Schechter.

B -1

Exhibit 10.4

LABCORP HOLDINGS INC. AMENDED AND RESTATED 2016 OMNIBUS INCENTIVE PLAN

TABLE OF CONTENTS

				<u>Page</u>	
1.	PURP	OSE	1		
2.	DEFI	NITIONS		1	
3.	ADM	8			
	3.1	Comm	nittee	8	
		3.1.1	Powers and Authorities	8	
		3.1.2	Composition of the Committee	9	
		3.1.3	Other Committees	9	
		3.1.4	Delegation by the Committee.	9	
	3.2	Board		10	
	3.3	Terms	10		
		3.3.1	Committee Authority	10	
		3.3.2	Forfeiture; Recoupment	10	
	3.4	No Re	pricing	11	
	3.5	Deferr	ral Arrangement	11	
	3.6	Regist	tration; Share Certificates	11	
4.	STOC	12			
	4.1	Numbe	er of Shares of Stock Available for Awards	12	
	4.2	Adjust	tments in Authorized Shares of Stock	12	
	4.3	Share	Usage	12	
5.	TERM	13			
	5.1	Term		13	
	5.2	13			
6.	AWAI	14			
	6.1	Eligibl	le Grantees	14	
	6.2	Limita	ation on Shares of Stock Subject to Awards and Cash Awards	14	
	6.3	Stand-	Alone, Additional, Tandem, and Substitute Awards	14	
7.	AWARD AGREEMENT				
8.	TERMS AND CONDITIONS OF OPTIONS				
	8.1	Option	n Price	15	
	8.2	Vesting	g and Exercisability	15	
	8.3	Term		16	
	8.4	Termir	nation of Service	16	
	8.5	Limita	ations on Exercise of Option	16	
	8.6	Metho	od of Exercise	16	
	8.7	Rights	s of Holders of Options	16	
	8.8	Delive	ery of Stock	17	
	8.9	Transf	ferability of Options	17	
	8.10	Family	y Transfers	17	
	8.11 Limitations on Incentive Stock Options.			17	
	8.12	Notice	e of Disqualifying Disposition.	18	

9.	TERM	IS AND CONDITIONS OF STOCK APPRECIATION RIGHTS	18
	9.1	Right to Payment and SAR Price	18
	9.2	Other Terms	18
	9.3	Term	19
	9.4	Rights of Holders of SARs.	19
	9.5	Transferability of SARs	19
	9.6	Family Transfers	19
10	TERM	IS AND CONDITIONS OF RESTRICTED STOCK, RESTRICTED STOCK UNITS, AND DEFERRED	10
10.		K UNITS Creat of Destricted Stock, Destricted Stock Units, and Deformed Stock Units	19
	10.1 10.2	Grant of Restricted Stock, Restricted Stock Units, and Deferred Stock Units Restrictions	19 20
	10.2		20 20
	10.3	Registration; Restricted Stock Certificates Rights of Holders of Restricted Stock	20 20
	10.4	•	
	10.5	Rights of Holders of Restricted Stock Units and Deferred Stock Units10.5.1Voting and Dividend Rights	21 21
		· · ·	
	10.6	10.5.2 Creditor's Rights Termination of Service	21 21
	10.0	Purchase of Restricted Stock and Shares of Stock Subject to Restricted Stock Units and Deferred Stock Units	21
	10.7	Delivery of Shares of Stock	21
11.		IS AND CONDITIONS OF UNRESTRICTED STOCK AWARDS AND OTHER EQUITY-BASED AWARDS	21
11.	11.1	Unrestricted Stock Awards	22
	11.1	Other Equity-Based Awards	22
12.		IS AND CONDITIONS OF DIVIDEND EQUIVALENT RIGHTS	22
	12.1	Dividend Equivalent Rights	22
	12.2	Termination of Service	23
13.		IS AND CONDITIONS OF PERFORMANCE-BASED AWARDS	23
	13.1	Grant of Performance-Based Awards	23
	13.2	Value of Performance-Based Awards	23
	13.3	Earning of Performance-Based Awards	23
	13.4	Form and Timing of Payment of Performance-Based Awards	23
	13.5	Performance Conditions	24
	13.6	Performance-Based Awards Granted to Designated Covered Employees	24
		13.6.1 Performance Goals Generally	24
		13.6.2 Timing For Establishing Performance Goals	24
		13.6.3 Payment of Awards; Other Terms	24
		13.6.4 Performance Measures	25
		13.6.5 Evaluation of Performance	26
		13.6.6 Adjustment of Performance-Based Compensation	27
		13.6.7 Committee Discretion	27
		13.6.8 Status of Awards Under Code Section 162(m)	27
14.	FORM	IS OF PAYMENT	27
	14.1	General Rule	27
	14.2	Surrender of Shares of Stock	27

ii

	14.3	Cashless Exercise	28	
	14.4	Other Forms of Payment	28	
15.	REQUIREMENTS OF LAW			
	15.1	General	28	
	15.2	Rule 16b-3	29	
16.	EFFECT OF CHANGES IN CAPITALIZATION			
	16.1	Changes in Stock	29	
	16.2	Reorganization in Which the Company Is the Surviving Entity Which Does Not Constitute a Change in Control	30	
	16.3	Change in Control in Which Awards Are Not Assumed	30	
	16.4	Change in Control in Which Awards Are Assumed	31	
	16.5	Adjustments.	32	
	16.6	No Limitations on Company	32	
17.	PARA	CHUTE LIMITATIONS	32	
18.	GENERAL PROVISIONS			
	18.1	Disclaimer of Rights	33	
	18.2	Nonexclusivity of the Plan	33	
	18.3	Withholding Taxes	33	
	18.4	Captions	34	
	18.5	Construction	34	
	18.6	Other Provisions	34	
	18.7	Number and Gender	34	
	18.8	Severability	34	
	18.9	Governing Law	34	
	18.10	Foreign Jurisdictions	35	
	18.11	Section 409A of the Code	35	
	18.12	Limitation on Liability.	36	

iii

LABCORP HOLDINGS INC. AMENDED AND RESTATED 2016 OMNIBUS INCENTIVE PLAN

1. PURPOSE

The Plan is intended to (a) provide eligible individuals with an incentive to contribute to the success of the Company and to operate and manage the Company's business in a manner that will provide for the Company's long-term growth and profitability and that will benefit its shareholders and other important stakeholders, including its employees and customers, and (b) provide a means of recruiting, rewarding, and retaining key personnel. To this end, the Plan provides for the grant of Awards of Options, Stock Appreciation Rights, Restricted Stock, Restricted Stock Units, Deferred Stock Units, Unrestricted Stock, Dividend Equivalent Rights, Performance Shares and other Performance-Based Awards, Other Equity-Based Awards, and cash bonus awards. Any of these Awards may, but need not, be made as performance incentives to reward the holders of such Awards for the achievement of performance goals in accordance with the terms of the Plan. Options granted under the Plan may be Non-qualified Stock Options or Incentive Stock Options, as provided herein.

2. **DEFINITIONS**

For purposes of interpreting the Plan documents, including the Plan and Award Agreements, the following capitalized terms shall have the meanings specified below, unless the context clearly indicates otherwise:

2.1 "Affiliate" shall mean any Person that controls, is controlled by, or is under common control with the Company within the meaning of Rule 405 of Regulation C under the Securities Act, including any Subsidiary. For purposes of grants of Options or Stock Appreciation Rights, an entity may not be considered an Affiliate unless the Company holds a Controlling Interest in such entity.

2.2 "Applicable Laws" shall mean the legal requirements relating to the Plan and the Awards under (a) applicable provisions of the Code, the Securities Act, the Exchange Act, any rules or regulations thereunder, and any other laws, rules, regulations, and government orders of any jurisdiction applicable to the Company or its Affiliates, (b) applicable provisions of the corporate, securities, tax, and other laws, rules, regulations, and government orders of any jurisdiction applicable to Awards granted to residents thereof, and (c) the rules of any Stock Exchange or Securities Market on which the Stock is listed or publicly traded.

2.3 "Award" shall mean a grant under the Plan of an Option, a Stock Appreciation Right, Restricted Stock, a Restricted Stock Unit, a Deferred Stock Unit, Unrestricted Stock, a Dividend Equivalent Right, a Performance Share or other Performance-Based Award, an Other Equity-Based Award, or cash.

2.4 "Award Agreement" shall mean the written agreement, in such written, electronic, or other form as determined by the Committee, between the Company and a Grantee that evidences and sets forth the terms and conditions of an Award.

2.5 "Beneficial Owner" shall have the meaning set forth in Rule 13d-3 under the Exchange Act.

2.6 "Benefit Arrangement" shall mean any formal or informal plan or other arrangement for the direct or indirect provision of compensation to a Grantee (including groups or classes of Grantees or beneficiaries of which the Grantee is a member), whether or not such compensation is deferred, is in cash, or is in the form of a benefit to or for the Grantee.

2.7 "Board" shall mean the Board of Directors of the Company.

2.8 "**Cause**" shall have the meaning set forth in an applicable agreement between a Grantee and the Company or an Affiliate, and in the absence of any such agreement, shall mean, with respect to any Grantee and as determined by the Committee, (a) gross negligence or willful misconduct in connection with the performance of duties; (b) conviction of, or pleading guilty or *nolo contendere* to, a criminal offense (other than minor traffic offenses); or (c) material breach of any term of any employment, consulting or other services, confidentiality, intellectual property, or non-competition agreements, if any, between such Grantee and the Company or an Affiliate. Any determination by the Committee regarding whether an event constituting Cause shall have occurred shall be final, binding, and conclusive.

2.9 "Capital Stock" shall mean, with respect to any Person, any and all shares, interests, participations, or other equivalents (however designated, whether voting or non-voting) in equity of such Person, whether outstanding on the Effective Date or issued thereafter, including, without limitation, all shares of Stock.

2.10 "Change in Control" shall mean, subject to Section 18.11, the occurrence of any of the following:

(a) Any transaction (including without limitation a merger or reorganization in which the Company is the surviving entity) which results in any Person (other than any employee benefit plan (or related trust) of the Company) owning thirty percent (30%) or more of the combined voting power of all classes of Voting Stock; excluding, however, the following: (1) any acquisition directly from the Company, other than as a result of the exercise of a conversion privilege, (2) any repurchase by the Company, (3) any acquisition by any employee benefit plan (or related trust) of the Company or any entity controlled by the Company, or (4) any acquisition pursuant to a transaction described in Section 2.10(c)(i), (ii) or (iii);

(b) Individuals who constitute the Board on the Effective Date (the "**Incumbent Board**") cease for any reason to constitute at least a majority thereof, provided that any Person becoming a director subsequent to the date hereof whose election was approved by a vote of at least three-quarters of the directors comprising the Incumbent Board, or whose nomination for election by the Company's shareholders was approved by the Incumbent Board, shall be for purposes of this clause (b), considered as though he or she were a member of the Incumbent Board;

(c) The Company consolidates with, or merges with or into, any Person, or any Person consolidates with, or merges with or into, the Company (regardless of whether the Company is the surviving Person), other than any such transaction in which (i) all or substantially all of the Prior Shareholders own directly or indirectly at least a majority of the voting power of the Voting Stock of the surviving Person in such reorganization, merger, or consolidation transaction immediately after such transaction, (ii) no Person (other than the Company, any employee benefit plan (or related trust) of the Company or such corporation resulting from such reorganization, merger or consolidation, except to the extent such ownership derives from ownership that existed prior to such reorganization, merger or consolidation, and (iii) individuals who were members of the Board immediately prior to such reorganization, consolidation or merger will constitute at least a majority of the members of the board of directors of the surviving Person resulting from such reorganization, merger or consolidation;

(d) The dissolution or liquidation of the Company; or

(e) Sale of substantially all of the assets of the Company to another Person.

The Board shall have full and final authority, in its sole discretion, to determine conclusively whether a Change in Control has occurred pursuant to the above definition, the date of the occurrence of such Change in Control, and any incidental matters relating thereto.

2.11 "Code" shall mean the Internal Revenue Code of 1986, as amended, as now in effect or as hereafter amended, and any successor thereto. References in the Plan to any Code Section shall be deemed to include, as applicable, regulations and guidance promulgated under such Code Section.

2.12 "Committee" shall mean a committee of, and designated from time to time by resolution of, the Board, which shall be constituted as provided in Section 3.1.2 and Section 3.1.3 (or, if no Committee has been so designated, the Board).

2.13 "Company" shall mean Labcorp Holdings Inc. and any successor thereto.

2.14 "Controlling Interest" shall have the meaning set forth in Treasury Regulation Section 1.414(c)-2(b)(2)(i); provided that (a) except as specified in clause (b) below, an interest of "at least 50 percent" shall be used instead of an interest of "at least 80 percent" in each case where "at least 80 percent" appears in Treasury Regulation Section 1.414(c)-2(b)(2)(i), and (b) where a grant of Options or Stock Appreciation Rights is based upon a legitimate business criterion, an interest of "at least 20 percent" shall be used instead of an interest of "at least 80 percent" in each case where "at least 80 percent" appears in Treasury Regulation Section 1.414(c)-2(b)(2)(i).

2.15 "Covered Employee" shall mean a Grantee who is, or could become, a "covered employee" within the meaning of Code Section 162(m)(3).

2.16 "Deferred Stock Unit" shall mean a Restricted Stock Unit, the terms of which provide for delivery of the underlying shares of Stock, cash, or a combination thereof subsequent to the date of vesting, at a time or times consistent with the requirements of Code Section 409A.

2.17 "Disability" shall mean the inability of a Grantee to perform each of the essential duties of such Grantee's position by reason of a medically determinable physical or mental impairment which is potentially permanent in character or which can be expected to last for a continuous period of not less than twelve (12) months; provided that, with respect to rules regarding the expiration of an Incentive Stock Option following termination of a Grantee's Service, Disability shall mean the inability of such Grantee to engage in any substantial gainful activity by reason of a medically determinable physical or mental impairment which can be expected to result in death or which has lasted or can be expected to last for a continuous period of not less than twelve (12) months.

2.18 "Disqualified Individual" shall have the meaning set forth in Code Section 280G(c).

2.19 "Dividend Equivalent Right" shall mean a right, granted to a Grantee pursuant to Article 12, entitling the Grantee thereof to receive, or to receive credits for the future payment of, cash, Stock, other Awards, or other property equal in value to dividend payments or distributions, or other periodic payments, declared or paid with respect to a number of shares of Stock specified in such Dividend Equivalent Right (or other Award to which such Dividend Equivalent Right relates) as if such shares of Stock had been issued to and held by the Grantee of such Dividend Equivalent Right as of the record date.

2.20 "Effective Date" shall mean May 17, 2024.

2.21 "Employee" shall mean, as of any date of determination, an employee (including an officer) of the Company or an Affiliate.

2.22 "Exchange Act" shall mean the Securities Exchange Act of 1934, as amended, as now in effect or as hereafter amended, and any successor thereto.

2.23 "Fair Market Value" shall mean the fair market value of a share of Stock for purposes of the Plan, which shall be, as of any date of determination:

(a) If on such date the shares of Stock are listed on a Stock Exchange, or are publicly traded on another Securities Market, the Fair Market Value of a share of Stock shall be the closing price of the Stock as reported on such Stock Exchange or such Securities Market (provided that, if there is more than one such Stock Exchange or Securities Market, the Committee shall designate the appropriate Stock Exchange or Securities Market for purposes of the Fair Market Value determination). If there is no such reported closing price on such date, the Fair Market Value of a share of Stock shall be the closing price of the Stock on the next preceding day on which any sale of Stock shall have been reported on such Stock Exchange or such Securities Market.

(b) If on such date the shares of Stock are not listed on a Stock Exchange or publicly traded on a Securities Market, the Fair Market Value of a share of Stock shall be the value of the Stock as determined by the Committee by the reasonable application of a reasonable valuation method, in a manner consistent with Code Section 409A.

Notwithstanding this Section 2.23 or Section 18.3, for purposes of determining taxable income and the amount of the related tax withholding obligation pursuant to Section 18.3, the Fair Market Value shall be determined by the Committee in good faith using any reasonable method as it deems appropriate, to be applied consistently with respect to Grantees; provided, further, that the Committee shall determine the Fair Market Value of shares of Stock for tax withholding obligations due in connection with sales, by or on behalf of a Grantee, of such shares of Stock subject to an Award to pay the Option Price, SAR Price, and/or any tax withholding obligation on the same date on which such shares may first be sold pursuant to the terms of the applicable Award Agreement (including broker-assisted cashless exercises of Options and Stock Appreciation Rights, as described in Section 14.3, and sell-to-cover transactions) in any manner consistent with applicable provisions of the Code, including but not limited to using the sale price of such shares on such date (or if sales of such shares are effectuated at more than one sale price, the weighted average sale price of such shares on such date) as the Fair Market Value of such shares, so long as such Grantee has provided the Company, or its designee or agent, with advance written notice of such sale.

2.24 "Family Member" shall mean, with respect to any Grantee as of any date of determination, (a) a Person who is a spouse, former spouse, child, stepchild, grandchild, parent, stepparent, grandparent, niece, nephew, mother-in-law, father-in-law, son-in-law, daughter-in-law, brother, sister, brother-in-law, or sister-in-law, including adoptive relationships, of such Grantee, (b) any Person sharing such Grantee's household (other than a tenant or employee), (c) a trust in which any one or more of the Persons specified in clauses (a) and (b) above (and such Grantee) own more than fifty percent (50%) of the beneficial interest, (d) a foundation in which any one or more of the Persons specified in clauses (a) and (b) above (and such Grantee) own more than fifty percent (50%) of the management of assets, and (e) any other entity in which one or more of the Persons specified in clauses (a) and (b) above (and such Grantee) own more than fifty percent (50%) of the voting interests.

"Grant Date" shall mean, as determined by the Committee, the latest to occur of (a) the date as of which the Committee approves 2.25 the Award, (b) the date on which the recipient of an Award first becomes eligible to receive an Award under Article 6 hereof (e.g., in the case of a new hire, the first date on which such new hire performs any Service), or (c) such subsequent date specified by the Committee in the corporate action approving the Award.

2.26 "Grantee" shall mean a Person who receives or holds an Award under the Plan.

2.27 "Incentive Stock Option" shall mean an "incentive stock option" within the meaning of Code Section 422.

2.28 "Non-qualified Stock Option" shall mean an Option that is not an Incentive Stock Option.

2.29 "Non-Employee Director" shall have the meaning set forth in Rule 16b-3 under the Exchange Act.

2.30 "Officer" shall have the meaning set forth in Rule 16a-1(f) under the Exchange Act.

2.31 "Option" shall mean an option to purchase one or more shares of Stock at a specified Option Price awarded to a Grantee pursuant to Article 8.

2.32 "Option Price" shall mean the per share exercise price for shares of Stock subject to an Option.

"Original Effective Date" shall mean May 11, 2016. 2.33

2.34 "Other Agreement" shall mean any agreement, contract, or understanding heretofore or hereafter entered into by a Grantee with the Company or an Affiliate, except an agreement, contract, or understanding that expressly addresses Code Section 280G and/or Code Section 4999.

2.35 "Other Equity-Based Award" shall mean an Award representing a right or other interest that may be denominated or payable in, valued in whole or in part by reference to, or otherwise based on or related to Stock, other than an Option, a Stock Appreciation Right, Restricted Stock, a Restricted Stock Unit, a Deferred Stock Unit, Unrestricted Stock, a Dividend Equivalent Right, or a Performance Share.

2.36 "Outside Director" shall have the meaning set forth in Code Section 162(m)(4)(C)(i).

2.37 "Parachute Payment" shall mean a "parachute payment" within the meaning of Code Section 280G(b)(2).

2.38 "Performance-Based Award" shall mean an Award of Options, Stock Appreciation Rights, Restricted Stock, Restricted Stock Units, Deferred Stock Units, Performance Shares, Other Equity-Based Awards, or cash made subject to the achievement of performance goals (as provided in Article 13) over a Performance Period specified by the Committee.

2.39 "Performance-Based Compensation" shall mean compensation under an Award that is intended to satisfy the requirements of Code Section 162(m) for Qualified Performance-Based Compensation paid to Covered Employees. Notwithstanding the foregoing, nothing in the Plan shall be construed to mean that an Award which does not satisfy the requirements for Qualified Performance-Based Compensation does not constitute performance-based compensation for other purposes, including the purposes of Code Section 409A.

2.40 "Performance Measures" shall mean measures as specified in Section 13.6.4 on which the performance goal or goals under Performance-Based Awards are based and which are approved by the Company's shareholders pursuant to, and to the extent required by, the Plan in order to qualify such Performance-Based Awards as Performance-Based Compensation.

2.41 "**Performance Period**" shall mean the period of time, up to ten (10) years, during or over which the performance goals under Performance-Based Awards must be met in order to determine the degree of payout and/or vesting with respect to any such Performance-Based Awards.

2.42 "Performance Shares" shall mean a Performance-Based Award representing a right or other interest that may be denominated or payable in, valued in whole or in part by reference to, or otherwise based on or related to Stock, made subject to the achievement of performance goals (as provided in Article 13) over a Performance Period.

2.43 "Person" shall mean an individual, a corporation, a partnership, a limited liability company, an association, a trust, or any other entity or organization, including a government or political subdivision or an agency or instrumentality thereof; provided that, for purposes of Section 2.10(a), Section 2.10(c) and Section 2.10(d), Person shall have the meaning set forth in Sections 13(d) and 14(d)(2) of the Exchange Act.

2.44 "Plan" shall mean this Labcorp Holdings Inc. Amended and Restated 2016 Omnibus Incentive Plan, as it may be further amended from time to time.

2.45 "Prior Plans" shall mean the Laboratory Corporation of America Holdings 2012 Omnibus Incentive Plan, as amended, the Laboratory Corporation of America Holdings 2008 Stock Incentive Plan, as amended, the Laboratory Corporation of America Holdings 2000 Stock Incentive Plan, as amended and restated, the Laboratory Corporation of America Holdings 1994 Stock Option Plan.

2.46 "**Prior Shareholders**" shall mean the holders of equity securities that represented one hundred percent (100%) of the Voting Stock of the Company immediately prior to a reorganization, merger, or consolidation involving the Company (or other equity securities into which such equity securities are converted as part of such reorganization, merger, or consolidation transaction).

2.47 "Qualified Performance-Based Compensation" shall have the meaning set forth in Code Section 162(m).

2.48 "**Restricted Period**" shall mean a period of time established by the Committee during which an Award of Restricted Stock, Restricted Stock Units, or Deferred Stock Units is subject to restrictions.

2.49 "Restricted Stock" shall mean shares of Stock awarded to a Grantee pursuant to Article 10.

2.50 "**Restricted Stock Unit**" shall mean a bookkeeping entry representing the equivalent of one (1) share of Stock awarded to a Grantee pursuant to **Article 10** that may be settled, subject to the terms and conditions of the applicable Award Agreement, in shares of Stock, cash, or a combination thereof.

2.51 "SAR Price" shall mean the per share exercise price of a SAR.

2.52 "Securities Act" shall mean the Securities Act of 1933, as amended, as now in effect or as hereafter amended, and any successor thereto.

2.53 "Securities Market" shall mean an established securities market.

2.54 "Separation from Service" shall have the meaning set forth in Code Section 409A.

2.55 "Service" shall mean service qualifying a Grantee as a Service Provider to the Company or an Affiliate. Unless otherwise provided in the applicable Award Agreement, a Grantee's change in position or duties shall not result in interrupted or terminated Service, so long as such Grantee continues to be a Service Provider to the Company or an Affiliate. Subject to the preceding sentence, any determination by the Committee whether a termination of Service shall have occurred for purposes of the Plan shall be final, binding, and conclusive. If a Service Provider's employment or other Service relationship is with an Affiliate and the applicable entity ceases to be an Affiliate, a termination of Service shall be deemed to have occurred when such entity ceases to be an Affiliate unless the Service Provider transfers his or her employment or other Service relationship to the Company or any other Affiliate.

2.56 "Service Provider" shall mean (a) an Employee or director of the Company or an Affiliate, or (b) a consultant or adviser to the Company or an Affiliate (i) who is a natural person, (ii) who is currently providing bona fide services to the Company or an Affiliate, and (iii) whose services are not in connection with the Company's sale of securities in a capital-raising transaction and do not directly or indirectly promote or maintain a market for the Company's Capital Stock.

2.57 "Service Recipient Stock" shall have the meaning set forth in Code Section 409A.

2.58 "Share Limit" shall have the meaning set forth in Section 4.1.

2.59 "Short-Term Deferral Period" shall have the meaning set forth in Code Section 409A.

2.60 "Stock" shall mean the common stock, par value \$0.10 per share, of the Company, or any security into which shares of Stock may be changed or for which shares of Stock may be exchanged as provided in Section 16.1.

2.61 "Stock Appreciation Right" or "SAR" shall mean a right granted to a Grantee pursuant to Article 9.

2.62 "Stock Exchange" shall mean the New York Stock Exchange, the NASDAQ Capital Market, the NASDAQ Global Market, the NASDAQ Global Select Market, or another established national or regional stock exchange.

2.63 "Subsidiary" shall mean any corporation (other than the Company) or non-corporate entity with respect to which the Company owns, directly or indirectly, fifty percent (50%) or more of the total combined voting power of all classes of Voting Stock. In addition, any other entity may be designated by the Committee as a Subsidiary, provided that (a) such entity could be considered as a subsidiary according to generally accepted accounting principles in the United States of America and (b) in the case of an Award of Options or Stock Appreciation Rights, such Award would be considered to be granted in respect of Service Recipient Stock under Code Section 409A.

2.64 "Substitute Award" shall mean an Award granted upon assumption of, or in substitution for, outstanding awards previously granted under a compensatory plan of the Company, an Affiliate, or a business entity acquired or to be acquired by the Company or an Affiliate or with which the Company or an Affiliate has combined or will combine.

2.65 "**Ten Percent Shareholder**" shall mean a natural Person who owns more than ten percent (10%) of the total combined voting power of all classes of Voting Stock of the Company, the Company's parent (if any), or any of the Company's Subsidiaries. In determining stock ownership, the attribution rules of Code Section 424(d) shall be applied.

2.66 "Unrestricted Stock" shall mean Stock that is free of any restrictions.

2.67 "Voting Stock" shall mean, with respect to any Person, Capital Stock of any class or kind ordinarily having the power to vote for the election of directors, managers, or other voting members of the governing body of such Person.

3. ADMINISTRATION OF THE PLAN

3.1 Committee.

3.1.1 Powers and Authorities.

The Committee shall administer the Plan and shall have such powers and authorities related to the administration of the Plan as are consistent with the Company's certificate of incorporation and bylaws and Applicable Laws. Without limiting the generality of the foregoing, the Committee shall have full power and authority to take all actions and to make all determinations required or provided for under the Plan, any Award, or any Award Agreement and shall have full power and authority to take all such other actions and to make all such other determinations not inconsistent with the specific terms and provisions of the Plan which the Committee deems to be necessary or appropriate to the administration of the Plan, any Award, or any Award Agreement. For the avoidance of doubt, the Committee reserves the right to accelerate the vesting of all Awards granted under the Plan. All such actions and determinations shall be made by (a) the affirmative vote of a majority of the members of the Committee present at a meeting at which a quorum is present, or (b) the unanimous consent of the members of the Committee executed in writing or evidenced by electronic transmission in accordance with the Company's certificate of incorporation and bylaws and Applicable Laws. Unless otherwise expressly determined by the Board, the Committee shall have the authority to interpret and construe all provisions of the Plan, any Award Agreement, and any such interpretation or construction, and any other determination contemplated to be made under the Plan or any Award Agreement, by the Committee shall be final, binding, and conclusive on all Persons, whether or not expressly provided for in any provision of the Plan, such Award, or such Award Agreement.

In the event that the Plan, any Award, or any Award Agreement provides for any action to be taken by the Board or any determination to be made by the Board, such action may be taken or such determination may be made by the Committee constituted in accordance with this **Section 3.1** if the Board has delegated the power and authority to do so to such Committee.

3.1.2 Composition of the Committee.

The Committee shall be a committee composed of not fewer than two (2) directors of the Company designated by the Board to administer the Plan. Each member of the Committee shall be (a) a Non-Employee Director, (b) an Outside Director, and (c) an independent director in accordance with the rules of any Stock Exchange on which the Stock is listed; provided that any action taken by the Committee shall be valid and effective whether or not members of the Committee at the time of such action are later determined not to have satisfied the requirements for membership set forth in this **Section 3.1.2** or otherwise provided in any charter of the Committee. Without limiting the generality of the foregoing, the Committee may be the Compensation

Committee of the Board or a subcommittee thereof if the Compensation Committee of the Board or such subcommittee satisfies the foregoing requirements.

3.1.3 Other Committees.

The Board also may appoint one or more committees of the Board, each composed of one or more directors of the Company who need not be Outside Directors, which (a) may administer the Plan with respect to Grantees who are not Officers or directors of the Company, (b) may grant Awards under the Plan to such Grantees, and (c) may determine all terms of such Awards, in each case, excluding (for the avoidance of doubt) Performance-Based Awards intending to constitute Qualified Performance-Based Compensation and subject, if applicable, to the requirements of Rule 16b-3 under the Exchange Act and the rules of any Stock Exchange or Securities Market on which the Stock is listed or publicly traded.

3.1.4 Delegation by the Committee.

To the extent permitted by Applicable Laws, the Committee may, by resolution, delegate some or all of its authority with respect to the Plan and Awards to the Chief Executive Officer of the Company and/or any other officer of the Company designated by the Committee, provided that the Committee may not delegate its authority hereunder (a) to make Awards to directors of the Company, (b) to make Awards to Employees who are (i) Officers, (ii) Covered Employees, or (iii) officers of the Company who are delegated authority by the Committee pursuant to this **Section 3.1.4**, or (c) to interpret the Plan, any Award, or any Award Agreement. Any delegation hereunder shall be subject to the restrictions and limits that the Committee specifies at the time of such delegation or thereafter. Nothing in the Plan shall be construed as obligating the Committee to delegate authority to any officer of the Company, and the Committee may at any time rescind the authority delegated to an officer of the Company appointed hereunder and delegate authority to one or more other officers of the Company. At all times, an officer of the Company delegated authority pursuant to this **Section 3.1.4** shall serve in such capacity at the pleasure of the Committee. Any action undertaken by any such officer of the Company in accordance with the Committee's delegation of authority shall have the same force and effect as if undertaken directly by the Committee, and any reference in the Plan to the "Committee" shall, to the extent consistent with the terms and limitations of such delegation, be deemed to include a reference to each such officer.

3.2 Board.

The Board, from time to time, may exercise any or all of the powers and authorities related to the administration and implementation of the Plan, as set forth in **Section 3.1** and other applicable provisions of the Plan, as the Board shall determine, consistent with the Company's certificate of incorporation and bylaws and Applicable Laws.

3.3 Terms of Awards.

3.3.1 Committee Authority.

Subject to the other terms and conditions of the Plan, the Committee shall have full and final authority to:

- (a) designate Grantees;
- (b) determine the type or types of Awards to be made to a Grantee;

(c) determine the number of shares of Stock to be subject to an Award or to which an Award relates;

(d) establish the terms and conditions of each Award (including the Option Price, the SAR Price, and the purchase price for applicable Awards; the nature and duration of any restriction or condition (or provision for lapse thereof) relating to the vesting, exercise, transfer, or forfeiture of an Award or the shares of Stock subject thereto; the treatment of an Award in the event of a Change in Control (subject to applicable agreements); and any terms or conditions that may be necessary to qualify Options as Incentive Stock Options);

(e) prescribe the form of each Award Agreement evidencing an Award;

(f) subject to the limitation on repricing in Section 3.4, amend, modify, or supplement the terms of any outstanding Award, which authority shall include the authority, in order to effectuate the purposes of the Plan but without amending the Plan, to make Awards or to modify outstanding Awards made to eligible natural Persons who are foreign nationals or are natural Persons who are employed outside the United States to reflect differences in local law, tax policy, or custom; provided that, notwithstanding the foregoing, no amendment, modification, or supplement of the terms of any outstanding Award shall, without the consent of the Grantee thereof, materially impair such Grantee's rights under such Award; and

(g) make Substitute Awards.

3.3.2 Forfeiture; Recoupment.

The Committee may reserve the right in an Award Agreement to cause a forfeiture of the gain realized by a Grantee with respect to an Award thereunder on account of actions taken by, or failed to be taken by, such Grantee in violation or breach of, or in conflict with, any (a) employment agreement, (b) non-competition agreement, (c) agreement prohibiting solicitation of Employees or clients of the Company or an Affiliate, (d) confidentiality obligation with respect to the Company or an Affiliate, (e) Company or Affiliate policy or procedure, (f) other agreement, or (g) other obligation of such Grantee to the Company or an Affiliate, as and to the extent specified in such Award Agreement. If the Grantee of an outstanding Award is an Employee of the Company or an Affiliate and such Grantee's Service is terminated for Cause, the Committee may annul such Grantee's outstanding Award as of the date of the Grantee's termination of Service for Cause.

Any Award granted pursuant to the Plan shall be subject to mandatory repayment by the Grantee to the Company (x) to the extent set forth in this Plan or an Award Agreement or (y) to the extent the Grantee is, or in the future becomes, subject to (1) any Company or Affiliate "clawback" or recoupment policy that is adopted to comply with the requirements of any Applicable Laws, or (2) any Applicable Laws which impose mandatory recoupment, under circumstances set forth in such Applicable Laws.

3.4 No Repricing.

Except in connection with a corporate transaction involving the Company (including, without limitation, any stock dividend, distribution (whether in the form of cash, shares of Stock, other securities, or other property), stock split, extraordinary dividend, recapitalization, Change in Control, reorganization, merger, consolidation, split-up, spin-off, combination, repurchase or exchange of shares of Stock, or other securities or similar transaction), the Company may not: (a) amend the terms of outstanding Options or SARs to reduce the Option Price or SAR Price, as applicable, of such outstanding Options or SARs; (b) cancel outstanding Options or SARs in exchange for or substitution of Options or SARs with an Option Price or SAR Price, as applicable, that is less than the Option Price or SAR Price, as applicable, of the original Options or SARs; or (c) cancel outstanding

Options or SARs with an Option Price or SAR Price, as applicable, above the current Fair Market Value in exchange for cash or other securities, in each case, unless such action is subject to and approved by the Company's shareholders.

3.5 Deferral Arrangement.

The Committee may permit or require the deferral of any payment pursuant to any Award into a deferred compensation arrangement, subject to such rules and procedures as it may establish, which may include provisions for the payment or crediting of interest or Dividend Equivalent Rights and, in connection therewith, provisions for converting such credits into Deferred Stock Units and for restricting deferrals to comply with hardship distribution rules affecting tax-qualified retirement plans subject to Code Section 401(k)(2)(B)(IV); provided that no Dividend Equivalent Rights may be granted in connection with, or related to, an Award of Options or SARs. Any such deferrals shall be made in a manner that complies with Code Section 409A, including, if applicable, with respect to when a Separation from Service occurs.

3.6 Registration; Share Certificates.

Notwithstanding any provision of the Plan to the contrary, the ownership of the shares of Stock issued under the Plan may be evidenced in such a manner as the Committee, in its sole discretion, deems appropriate, including by book-entry or direct registration (including transaction advices) or the issuance of one or more share certificates.

4. STOCK SUBJECT TO THE PLAN

4.1 Number of Shares of Stock Available for Awards.

Subject to adjustment pursuant to Article 16, the maximum number of shares of Stock reserved for issuance under the Plan shall be equal to eleven million two hundred thousand (11,200,000), less one (1) share for every one (1) share that was subject to an option or stock appreciation right granted after December 31, 2015 under any Prior Plan and 2.62 shares for every one (1) share that was subject to an award other than an option or stock appreciation right granted after December 31, 2015, in each case using the share usage provisions set forth in Section 4.3 (the "Share Limit"). Such shares of Stock may be authorized and unissued shares of Stock, treasury shares of Stock, or any combination of the foregoing, as may be determined from time to time by the Board or by the Committee. Any of the shares of Stock reserved and available for issuance under the Plan may be used for any type of Award under the Plan, and any or all of the shares of Stock reserved for issuance under the Plan shall be available for issuance pursuant to Incentive Stock Options.

4.2 Adjustments in Authorized Shares of Stock.

In connection with mergers, reorganizations, separations, or other transactions to which Code Section 424(a) applies, the Committee shall have the right to cause the Company to assume awards previously granted under a compensatory plan of another business entity that is a party to such transaction and to grant Substitute Awards under the Plan for such awards. The Share Limit shall neither be increased nor decreased by the number of shares of Stock subject to any such assumed awards and Substitute Awards. Shares available for issuance under a shareholder-approved plan of a business entity that is a party to such transaction (as appropriately adjusted, if necessary, to reflect such transaction) may be used for Awards under the Plan and shall not reduce the number of shares of Stock otherwise available for issuance under the Plan, subject to applicable rules of any Stock Exchange or Securities Market on which the Stock is listed or publicly traded.

4.3 Share Usage.

(a) Shares of Stock subject to an Award shall be counted as used as of the Grant Date for purposes of calculating the number of shares of Stock available for issuance under **Section 4.1**.

(b) Any shares of Stock that are subject to Awards, including shares of Stock acquired through dividend reinvestment pursuant to **Article 10**, shall be counted against the Share Limit as: (i) 2.62 shares of Stock for every one (1) share of Stock subject to an Award other than Options or SARs and (ii) one (1) share of Stock for every one (1) share of Stock subject to an Award of Stock subject to an Award of stock-settled SARs shall be counted against the Share Limit as one (1) share of Stock for every one (1) share of Stock subject to such Award regardless of the number of shares of Stock actually issued to settle such SARs upon the exercise of the SARs. A number of shares of Stock at least equal to the target number of shares issuable under Performance Shares shall be counted against the Share Limit as of the Grant Date, but such number shall be adjusted to equal the actual number of shares issued upon settlement of the Performance Shares to the extent different from such number of shares.

(c) If any shares of Stock subject to an Award or to awards outstanding under the Prior Plans after December 31, 2015, are not purchased or are forfeited or expire or otherwise terminate without delivery of any Stock subject thereto or are settled in cash in lieu of shares, then the number of shares of Stock counted against the Share Limit with respect to such Award shall, to the extent of any such forfeiture, termination, expiration, or settlement, again be available for making Awards under the Plan, provided that any shares subject to an award granted under the Prior Plans shall be available for making Awards under the Plan in the same number as Awards under **Section 4.3(b)** (that is, as if such awards had been originally granted under the Plan). Shares of Stock tendered or withheld or subject to an Award (other than an Option or SAR), or to awards (other than options or stock appreciation rights) outstanding under the Prior Plans after December 31, 2015, surrendered in connection with the purchase of shares of Stock or deducted or delivered from payment of such Award in connection with the Company's tax withholding obligations shall again be available for making Awards under the Plan in the same number as such shares were counted against the Share Limit, provided that any such shares subject to an award granted under the Prior Plans shall be available for making Awards under the Prior Plans shall be available for making Awards under the Plan in the same number as such shares were counted against the Share Limit, provided that any such shares subject to an award granted under the Prior Plans shall be available for making Awards under the Plan in the same number as Awards under the Plan in the same number as such shares were counted against the Share Limit, provided that any such shares subject to an award granted under the Prior Plans shall be available for making Awards under the Plan in the same number as Awards under the Plan in the same number as Awards under the Plan in the same number as available for m

(d) The number of shares of Stock available for issuance under the Plan shall not be increased by the number of shares of Stock (i) tendered, withheld, or subject to an Option granted under the Plan surrendered in connection with the payment of the Option Price upon exercise of an Option or in connection with the Company's tax withholding obligations with respect to Options or stock-settled SARs, (ii) that were not issued upon the net settlement or net exercise of a Stock-settled SAR granted under the Plan, or (iii) purchased by the Company with proceeds from Option exercises.

5. TERM; AMENDMENT, SUSPENSION, AND TERMINATION

5.1 Term.

The Plan originally became effective as of the Original Effective Date. Following the Original Effective Date, no awards shall be made under the Prior Plans. Notwithstanding the foregoing, shares of Stock reserved under the Prior Plans to settle awards, including performance-based awards, which are made under the Prior Plans prior to the Original Effective Date may be issued and delivered following the Original Effective Date to settle such awards. The Plan shall terminate on the first to occur of (a) the day before the tenth (10th) anniversary of the Original Effective Date, (b) the date determined in accordance with **Section 5.2**, and (c) the date determined in accordance with **Section 16.3**; provided, however, that Incentive Stock Options

may not be granted under the Plan after March 25, 2026. Upon such termination of the Plan, all outstanding Awards shall continue to have full force and effect in accordance with the provisions of the terminated Plan and the applicable Award Agreement (or other documents evidencing such Awards).

5.2 Amendment, Suspension, and Termination.

The Board may, at any time and from time to time, amend, suspend, or terminate the Plan; provided that, with respect to Awards theretofore granted under the Plan, no amendment, suspension, or termination of the Plan shall, without the consent of the Grantee, materially impair the rights or obligations under any such Award. The effectiveness of any amendment to the Plan shall be contingent on approval of such amendment by the Company's shareholders to the extent provided by the Board or required by Applicable Laws; provided that no amendment shall be made to the norepricing provisions of **Section 3.4**, the Option Pricing provisions of **Section 8.1**, or the SAR Pricing provisions of **Section 9.1** without the approval of the Company's shareholders.

6. AWARD ELIGIBILITY AND LIMITATIONS

6.1 Eligible Grantees.

Subject to this Article 6, Awards may be made under the Plan to (a) any Service Provider, as the Committee shall determine and designate from time to time, and (b) any other individual whose participation in the Plan is determined to be in the best interests of the Company by the Committee.

6.2 Limitation on Shares of Stock Subject to Awards and Cash Awards.

During any time when the Company has any class of common equity securities registered under Section 12 of the Exchange Act, but subject to adjustment as provided in Article 16:

(a) The maximum number of shares of Stock that may be granted under the Plan, pursuant to Options or SARs, in a calendar year to any Person eligible for an Award under **Section 6.1**, other than a non-employee director of the Company, is three hundred thousand (300,000) shares;

(b) The maximum number of shares of Stock that may be granted under the Plan, pursuant to Awards other than Options or SARs that are intended to be Qualified Performance-Based Compensation and are Stock-denominated and are either Stock- or cash-settled, in a calendar year to any Person eligible for an Award under **Section 6.1** who is a Covered Employee is two hundred thousand (200,000) shares;

(c) The maximum number of shares of Stock subject to Awards granted during a single fiscal year to any non-employee director, taken together with any cash fees paid to such non-employee director during the fiscal year, shall not exceed \$600,000 in total value (calculating the value of any such Awards based on the grant date fair value of such Awards for financial reporting purposes); provided, that the Board may make exceptions to this limit for individual non-employee directors in extraordinary circumstances as the Board may determine in its sole discretion, so long as (x) the aggregate limit does not exceed \$750,000 in total value during a fiscal year and (y) the non-employee director receiving such additional compensation does not participate in the decision to award such compensation or in other contemporaneous compensation decisions involving non-employee directors; and

(d) The maximum amount that may be paid as a cash-denominated Performance-Based Award (whether or not cash-settled) that is intended to qualify as Performance-Based Compensation for a Performance Period of twelve (12) months or less to any Person eligible for an Award under **Section 6.1** who

is a Covered Employee shall be ten million dollars (\$10,000,000), and the maximum amount that may be paid as a cash-denominated Performance-Based Award (whether or not cash-settled) that is intended to qualify as Performance-Based Compensation for a Performance Period of greater than twelve (12) months to any Person eligible for an Award under **Section 6.1** who is a Covered Employee shall be ten million dollars (\$10,000,000) times the number of years in the Performance Period.

6.3 Stand-Alone, Additional, Tandem, and Substitute Awards.

Subject to **Section 3.4**, Awards granted under the Plan may, in the discretion of the Committee, be granted either alone or in addition to, in tandem with, or in substitution or exchange for, (a) any other Award, (b) any award granted under another plan of the Company, an Affiliate, or any business entity that has been a party to a transaction with the Company or an Affiliate, or (c) any other right of a Grantee to receive payment from the Company or an Affiliate. Such additional, tandem, exchange, or Substitute Awards may be granted at any time. If an Award is granted in substitution or exchange for another Award, or for an award granted under another plan of the Company, an Affiliate, or any business entity that has been a party to a transaction with the Company or an Affiliate, the Committee shall require the surrender of such other Award or award under such other plan in consideration for the grant of such exchange or Substitute Award. In addition, Awards may be granted in lieu of cash compensation, including in lieu of cash payments under other plans of the Company or an Affiliate. Notwithstanding **Section 8.1** and **Section 9.1**, but subject to **Section 3.4**, the Option Price of an Option or the SAR Price of a SAR that is a Substitute Award may be less than one hundred percent (100%) of the Fair Market Value of a share of Stock on the original Grant Date; provided that such Option Price or SAR Price is determined in accordance with the principles of Code Section 424 for any Incentive Stock Option and consistent with Code Section 409A for any other Option or SAR.

7. AWARD AGREEMENT

Each Award granted pursuant to the Plan shall be evidenced by an Award Agreement, which shall be in such form or forms as the Committee shall from time to time determine. Award Agreements utilized under the Plan from time to time or at the same time need not contain similar provisions but shall be consistent with the terms of the Plan. Each Award Agreement evidencing an Award of Options shall specify whether such Options are intended to be Non-qualified Stock Options or Incentive Stock Options, and, in the absence of such specification, such Options shall be deemed to constitute Non-qualified Stock Options. In the event of any inconsistency between the Plan and an Award Agreement, the provisions of the Plan shall control.

8. TERMS AND CONDITIONS OF OPTIONS

8.1 Option Price.

The Option Price of each Option shall be fixed by the Committee and stated in the Award Agreement evidencing such Option. Except in the case of Substitute Awards, the Option Price of each Option shall be at least the Fair Market Value of one (1) share of Stock on the Grant Date; provided that, in the event that a Grantee is a Ten Percent Shareholder, the Option Price of an Option granted to such Grantee that is intended to be an Incentive Stock Option shall be not less than one hundred ten percent (110%) of the Fair Market Value of one (1) share of Stock on the Grant Date. In no case shall the Option Price of any Option be less than the par value of one (1) share of Stock.

8.2 Vesting and Exercisability.

(a) Subject to Sections 8.3 and 16.3, each Option granted under the Plan shall become vested and/or exercisable at such times and under such conditions as shall be determined by the Committee and stated in the Award Agreement, in another agreement with the Grantee, or otherwise in writing; provided that no Option shall be granted to Grantees who are entitled to overtime under Applicable Laws that will vest or be exercisable within a six (6)-month period starting on the Grant Date.

(b) Notwithstanding Section 8.2(a), for Options (other than Substitute Awards) granted on or after the Original Effective Date, (i) the grant, issuance, retention, vesting, and/or exercise, of an Option that is based solely upon performance criteria and level of achievement versus such criteria shall be subject to a performance period of not less than twelve (12) months, and (ii) the grant, issuance, retention, vesting, and/or exercise, of an Option that is based solely upon continued Service and/or the passage of time may not vest or be exercised in full or in part prior to the satisfaction of a one (1) year minimum vesting period. Notwithstanding the foregoing, the Committee may provide for the earlier vesting and/or exercisability under any such Option (i) in the event of the Grantee's death or Disability, or (ii) in connection with a Change in Control.

(c) Notwithstanding the limitations in Section 8.2(b) and Section 9.2(b), up to five percent (5%) of the Share Limit may be granted pursuant to the Plan without being subject to the limitations in Section 8.2(b) and Section 9.2(b). The foregoing five percent (5%) share issuance limit shall be subject to adjustment consistent with the adjustment provisions of Section 16 and the share usage rules of Section 4.3.

8.3 Term.

Each Option granted under the Plan shall terminate, and all rights to purchase shares of Stock thereunder shall cease, on the day before the tenth (10th) anniversary of the Grant Date of such Option, or under such circumstances and on such date prior thereto as is set forth in the Plan or as may be fixed by the Committee and stated in the Award Agreement relating to such Option; provided that, in the event that the Grantee is a Ten Percent Shareholder, an Option granted to such Grantee that is intended to be an Incentive Stock Option shall not be exercisable after the day before the fifth (5th) anniversary of the Grant Date of such Option.

8.4 Termination of Service.

Each Award Agreement with respect to the grant of an Option shall set forth the extent to which the Grantee thereof, if at all, shall have the right to exercise such Option following termination of such Grantee's Service. Such provisions shall be determined in the sole discretion of the Committee, need not be uniform among all Options issued pursuant to the Plan, and may reflect distinctions based on the reasons for termination of Service.

8.5 Limitations on Exercise of Option.

Notwithstanding any provision of the Plan to the contrary, in no event may any Option be exercised, in whole or in part, after the occurrence of an event referred to in **Article 16** which results in the termination of such Option.

8.6 Method of Exercise.

Subject to the terms of Article 14 and Section 18.3, an Option that is exercisable may be exercised by the Grantee's delivery to the Company or its designee or agent of notice of exercise on any business day, at the

Company's principal office or the office of such designee or agent, on the form specified by the Company and in accordance with any additional procedures specified by the Committee. Such notice shall specify the number of shares of Stock with respect to which such Option is being exercised and shall be accompanied by payment in full of the Option Price of the shares of Stock for which such Option is being exercised, plus the amount (if any) of federal and/or other taxes which the Company may, in its judgment, be required to withhold with respect to the exercise of such Option.

8.7 Rights of Holders of Options.

Unless otherwise stated in the applicable Award Agreement, a Grantee or other Person holding or exercising an Option shall have none of the rights of a shareholder of the Company (for example, the right to receive cash or dividend payments or distributions attributable to the shares of Stock subject to such Option, to direct the voting of the shares of Stock subject to such Option, or to receive notice of any meeting of the Company's shareholders) until the shares of Stock subject thereto are fully paid and issued to such Grantee or other Person. Except as provided in **Article 16**, no adjustment shall be made for dividends, distributions, or other rights with respect to any shares of Stock subject to an Option for which the record date is prior to the date of issuance of such shares of Stock.

8.8 Delivery of Stock.

Promptly after the exercise of an Option by a Grantee and the payment in full of the Option Price with respect thereto, such Grantee shall be entitled to receive such evidence of such Grantee's ownership of the shares of Stock subject to such Option as shall be consistent with **Section 3.6**.

8.9 Transferability of Options.

Except as provided in **Section 8.10**, during the lifetime of a Grantee of an Option, only such Grantee (or, in the event of such Grantee's legal incapacity or incompetency, such Grantee's guardian or legal representative) may exercise such Option. Except as provided in **Section 8.10**, no Option shall be assignable or transferable by the Grantee to whom it is granted, other than by will or the laws of descent and distribution.

8.10 Family Transfers.

If authorized in the applicable Award Agreement or by the Committee, in its sole discretion, a Grantee may transfer, not for value, all or part of an Option which is not an Incentive Stock Option to any Family Member. For the purpose of this **Section 8.10**, a transfer "not for value" is a transfer which is (a) a gift, (b) a transfer under a domestic relations order in settlement of marital property rights, or (c) unless Applicable Laws do not permit such transfer, a transfer to an entity in which more than fifty percent (50%) of the voting interests are owned by Family Members (and/or the Grantee) in exchange for an interest in such entity. Following a transfer under this **Section 8.10**, any such Option shall continue to be subject to the same terms and conditions as were applicable immediately prior to such transfer. Subsequent transfers of transferred Options shall be prohibited except to Family Members of the original Grantee in accordance with this **Section 8.10** or by will or the laws of descent and distribution. The provisions of **Section 8.4** relating to termination of Service shall continue to be applied with respect to the original Grantee of the Option, following which such Option shall be exercisable by the transfere only to the extent, and for the periods specified, in **Section 8.4**.



8.11 Limitations on Incentive Stock Options.

An Option shall constitute an Incentive Stock Option only (a) if the Grantee of such Option is an Employee of the Company or any corporate Subsidiary, (b) to the extent specifically provided in the related Award Agreement, and (c) to the extent that the aggregate Fair Market Value (determined at the time such Option is granted) of the shares of Stock with respect to which all Incentive Stock Options held by such Grantee become exercisable for the first time during any calendar year (under the Plan and all other plans of the Company and its Affiliates) does not exceed one hundred thousand dollars (\$100,000). Except to the extent provided in the regulations under Code Section 422, this limitation shall be applied by taking Options into account in the order in which they were granted.

8.12 Notice of Disqualifying Disposition.

If any Grantee shall make any disposition of shares of Stock issued pursuant to the exercise of an Incentive Stock Option under the circumstances provided in Code Section 421(b) (relating to certain disqualifying dispositions), such Grantee shall notify the Company of such disposition immediately but in no event later than ten (10) days thereafter.

9. TERMS AND CONDITIONS OF STOCK APPRECIATION RIGHTS

9.1 Right to Payment and SAR Price.

A SAR shall confer on the Grantee to whom it is granted a right to receive, upon exercise thereof, the excess of (a) the Fair Market Value of one (1) share of Stock on the date of exercise, over (b) the SAR Price as determined by the Committee. The Award Agreement for a SAR shall specify the SAR Price, which shall be no less than the Fair Market Value of one (1) share of Stock on the Grant Date of such SAR. SARs may be granted in tandem with all or part of an Option granted under the Plan or at any subsequent time during the term of such Option, in combination with all or any part of any other Award, or without regard to any Option or other Award; provided that a SAR that is granted in tandem with all or part of an Option shall have the same term, and expire at the same time, as the related Option.

9.2 Other Terms.

(a) The Committee shall determine, on the Grant Date or thereafter, the time or times at which, and the circumstances under which, a SAR may be exercised in whole or in part (including based on achievement of performance goals and/or future Service requirements); the time or times at which SARs shall cease to be or become exercisable following termination of Service or upon other conditions; the method of exercise, method of settlement, form of consideration payable in settlement, method by or forms in which shares of Stock shall be delivered or deemed to be delivered to Grantees, whether or not a SAR shall be granted in tandem or in combination with any other Award; and any and all other terms and conditions of any SAR; provided that no SARs shall be granted to Grantees who are entitled to overtime under Applicable Laws that will vest or be exercisable within a six (6)-month period starting on the Grant Date.

(b) Notwithstanding Section 9.2(a), for SARs (other than Substitute Awards) granted on or after the Original Effective Date, (i) the grant, issuance, retention, vesting, and/or exercise, of a SAR that is based solely upon performance criteria and level of achievement versus such criteria shall be subject to a performance period of not less than twelve (12) months, and (ii) the grant, issuance, retention, vesting, and/or exercise, of a SAR that is based solely upon continued Service and/or the passage of time may not vest or be exercised in full or in part prior to the satisfaction of a one (1) year minimum vesting period. Notwithstanding

the foregoing, the Committee may provide for the earlier vesting and/or exercisability under any such SAR (i) in the event of the Grantee's death or Disability, or (ii) in connection with a Change in Control.

(c) Notwithstanding the limitations in Section 8.2(b) and Section 9.2(b), up to five percent (5%) of the Share Limit may be granted pursuant to the Plan without being subject to the limitations in Section 8.2(b) and Section 9.2(b). The foregoing five percent (5%) share issuance limit shall be subject to adjustment consistent with the adjustment provisions of Section 16 and the share usage rules of Section 4.3.

9.3 Term.

Each SAR granted under the Plan shall terminate, and all rights thereunder shall cease, on the day before the tenth (10th) anniversary of the Grant Date of such SAR or under such circumstances and on such date prior thereto as is set forth in the Plan or as may be fixed by the Committee and stated in the Award Agreement relating to such SAR.

9.4 Rights of Holders of SARs.

Unless otherwise stated in the applicable Award Agreement, a Grantee or other Person holding or exercising a SAR shall have none of the rights of a shareholder of the Company (for example, the right to receive cash or dividend payments or distributions attributable to the shares of Stock underlying such SAR, to direct the voting of the shares of Stock underlying such SAR, or to receive notice of any meeting of the Company's shareholders) until the shares of Stock underlying such SAR, if any, are issued to such Grantee or other Person. Except as provided in **Article 16**, no adjustment shall be made for dividends, distributions, or other rights with respect to any shares of Stock underlying a SAR for which the record date is prior to the date of issuance of such shares of Stock, if any.

9.5 Transferability of SARs.

Except as provided in **Section 9.6**, during the lifetime of a Grantee of a SAR, only the Grantee (or, in the event of such Grantee's legal incapacity or incompetency, such Grantee's guardian or legal representative) may exercise such SAR. Except as provided in **Section 9.6**, no SAR shall be assignable or transferable by the Grantee to whom it is granted, other than by will or the laws of descent and distribution.

9.6 Family Transfers.

If authorized in the applicable Award Agreement or by the Committee, in its sole discretion, a Grantee may transfer, not for value, all or part of a SAR to any Family Member. For the purpose of this **Section 9.6**, a transfer "not for value" is a transfer which is (a) a gift, (b) a transfer under a domestic relations order in settlement of marital property rights, or (c) unless Applicable Laws do not permit such transfer, a transfer to an entity in which more than fifty percent (50%) of the voting interests are owned by Family Members (and/or the Grantee) in exchange for an interest in such entity. Following a transfer under this **Section 9.6**, any such SAR shall continue to be subject to the same terms and conditions as were in effect immediately prior to such transfer. Subsequent transfers of transferred SARs shall be prohibited except to Family Members of the original Grantee in accordance with this **Section 9.6** or by will or the laws of descent and distribution.

10. TERMS AND CONDITIONS OF RESTRICTED STOCK, RESTRICTED STOCK UNITS, AND DEFERRED STOCK UNITS

10.1 Grant of Restricted Stock, Restricted Stock Units, and Deferred Stock Units.

Awards of Restricted Stock, Restricted Stock Units, and Deferred Stock Units may be made for consideration or for no consideration, other than the par value of the shares of Stock, which shall be deemed paid by past Service or, if so provided in the related Award Agreement or a separate agreement, the promise by the Grantee to perform future Service to the Company or an Affiliate.

10.2 Restrictions.

At the time a grant of Restricted Stock, Restricted Stock Units, or Deferred Stock Units is made, the Committee may, in its sole discretion, (a) establish a Restricted Period applicable to such Restricted Stock, Restricted Stock Units, or Deferred Stock Units and (b) prescribe restrictions in addition to or other than the expiration of the Restricted Period, including the achievement of corporate or individual performance goals, which may be applicable to all or any portion of such Restricted Stock, Restricted Stock Units, or Deferred Stock Units as provided in **Article 13**. Awards of Restricted Stock, Restricted Stock Units may not be sold, transferred, assigned, pledged, or otherwise encumbered or disposed of during the Restricted Period or prior to the satisfaction of any other restrictions prescribed by the Committee with respect to such Awards.

10.3 Registration; Restricted Stock Certificates.

Pursuant to Section 3.6, to the extent that ownership of Restricted Stock is evidenced by a book-entry registration or direct registration (including transaction advices), such registration shall be notated to evidence the restrictions imposed on such Award of Restricted Stock under the Plan and the applicable Award Agreement. Subject to Section 3.6 and the immediately following sentence, the Company may issue, in the name of each Grantee to whom Restricted Stock has been granted, certificates representing the total number of shares of Restricted Stock granted to the Grantee, as soon as reasonably practicable after the Grant Date of such Restricted Stock. The Committee may provide in an Award Agreement with respect to an Award of Restricted Stock that either (a) the Secretary of the Company shall hold such certificates for such Grantee's benefit until such time as such shares of Restricted Stock are forfeited to the Company or the restrictions applicable thereto lapse and such Grantee shall deliver a stock power to the Company with respect to each certificate, or (b) such certificates shall be delivered to such Grantee, provided that such certificates shall bear legends that comply with Applicable Laws and make appropriate reference to the restrictions imposed on such Award of Restricted Stock under the Plan and such Award Agreement.

10.4 Rights of Holders of Restricted Stock.

Unless the Committee provides otherwise in an Award Agreement and subject to the restrictions set forth in the Plan, any applicable Company program, and the applicable Award Agreement, holders of Restricted Stock shall have the right to vote such shares of Restricted Stock and the right to receive any dividend payments or distributions declared or paid with respect to such shares of Restricted Stock. The Committee may provide in an Award Agreement evidencing a grant of Restricted Stock that (a) any cash dividend payments or distributions paid on Restricted Stock shall be reinvested in shares of Stock, which may or may not be subject to the same vesting conditions and restrictions as applicable to such underlying shares of Restricted Stock or (b) any dividend payments or distributions declared or paid on shares of Restricted Stock

shall only be made or paid upon satisfaction of the vesting conditions and restrictions applicable to such shares of Restricted Stock. Dividend payments or distributions declared or paid on shares of Restricted Stock which vest or are earned based upon the achievement of performance goals shall not vest unless such performance goals for such shares of Restricted Stock are achieved, and if such performance goals are not achieved, the Grantee of such shares of Restricted Stock shall promptly forfeit and, to the extent already paid or distributed, repay to the Company such dividend payments or distributions. All stock dividend payments or distributions, if any, received by a Grantee with respect to shares of Restricted Stock as a result of any stock split, stock dividend, combination of stock, or other similar transaction shall be subject to the same vesting conditions and restrictions as applicable to such underlying shares of Restricted Stock.

10.5 Rights of Holders of Restricted Stock Units and Deferred Stock Units.

10.5.1 Voting and Dividend Rights.

Holders of Restricted Stock Units and Deferred Stock Units shall have no rights as shareholders of the Company (for example, the right to receive dividend payments or distributions attributable to the shares of Stock underlying such Restricted Stock Units and Deferred Stock Units, to direct the voting of the shares of Stock underlying such Restricted Stock Units and Deferred Stock Units, or to receive notice of any meeting of the Company's shareholders). The Committee may provide in an Award Agreement evidencing an Award of Restricted Stock Units or Deferred Stock Units that the holder thereof shall be entitled to receive Dividend Equivalent Rights with respect to each Restricted Stock Unit or Deferred Stock Unit.

10.5.2 Creditor's Rights.

A holder of Restricted Stock Units or Deferred Stock Units shall have no rights other than those of a general unsecured creditor of the Company. Restricted Stock Units and Deferred Stock Units represent unfunded and unsecured obligations of the Company, subject to the terms and conditions of the applicable Award Agreement.

10.6 Termination of Service.

Unless the Committee provides otherwise in an Award Agreement, in another agreement with the Grantee, or otherwise in writing after such Award Agreement is issued, but prior to termination of Grantee's Service, upon the termination of such Grantee's Service, any Restricted Stock, Restricted Stock Units, or Deferred Stock Units held by such Grantee that have not vested, or with respect to which all applicable restrictions and conditions have not lapsed, shall immediately be deemed forfeited. Upon forfeiture of such Restricted Stock, Restricted Stock Units, or Deferred Stock Units, the Grantee thereof shall have no further rights with respect thereto, including any right to vote such Restricted Stock or any right to receive dividends or Dividend Equivalent Rights, as applicable, with respect to such Restricted Stock, Restricted Stock Units, or Deferred Stock Units.

10.7 Purchase of Restricted Stock and Shares of Stock Subject to Restricted Stock Units and Deferred Stock Units.

The Grantee of an Award of Restricted Stock, vested Restricted Stock Units, or vested Deferred Stock Units shall be required, to the extent required by Applicable Laws, to purchase such Restricted Stock or the shares of Stock subject to such vested Restricted Stock Units or Deferred Stock Units from the Company at a purchase price equal to the greater of (x) the aggregate par value of the shares of Stock represented by such Restricted Stock or such vested Restricted Stock Units or Deferred Stock Units or (y) the purchase price, if any, specified in the Award Agreement relating to such Restricted Stock or such vested Restricted Stock Units

or Deferred Stock Units. Such purchase price shall be payable in a form provided in Article 14 or, in the sole discretion of the Committee, in consideration for Service rendered or to be rendered by the Grantee to the Company or an Affiliate.

10.8 Delivery of Shares of Stock.

Upon the expiration or termination of any Restricted Period and the satisfaction of any other conditions prescribed by the Committee, including, without limitation, any performance goals or delayed delivery period, the restrictions applicable to Restricted Stock, Restricted Stock Units, or Deferred Stock Units settled in shares of Stock shall lapse, and, unless otherwise provided in the applicable Award Agreement, a bookentry or direct registration (including transaction advices) or a certificate evidencing ownership of such shares of Stock shall, consistent with **Section 3.6**, be issued, free of all such restrictions, to the Grantee thereof or such Grantee's beneficiary or estate, as the case may be. Neither the Grantee, nor the Grantee's beneficiary or estate, shall have any further rights with regard to a Restricted Stock Unit or Deferred Stock Unit once the shares of Stock represented by such Restricted Stock Unit or Deferred Stock Unit have been delivered in accordance with this **Section 10.8**.

11. TERMS AND CONDITIONS OF UNRESTRICTED STOCK AWARDS AND OTHER EQUITY-BASED AWARDS

11.1 Unrestricted Stock Awards.

The Committee may, in its sole discretion, grant (or sell at the par value of a share of Stock or at such other higher purchase price as shall be determined by the Committee) an Award to any Grantee pursuant to which such Grantee may receive shares of Unrestricted Stock under the Plan. Awards of Unrestricted Stock may be granted or sold to any Grantee as provided in the immediately preceding sentence in respect of Service rendered or, if so provided in the related Award Agreement or a separate agreement, to be rendered by the Grantee to the Company or an Affiliate or other valid consideration, in lieu of or in addition to any cash compensation due to such Grantee.

11.2 Other Equity-Based Awards.

The Committee may, in its sole discretion, grant Awards in the form of Other Equity-Based Awards, as deemed by the Committee to be consistent with the purposes of the Plan. Awards granted pursuant to this **Section 11.2** may be granted with vesting, value, and/or payment contingent upon the achievement of one or more performance goals. The Committee shall determine the terms and conditions of Other Equity-Based Awards on the Grant Date or thereafter. Unless the Committee provides otherwise in an Award Agreement, in another agreement with the Grantee, or otherwise in writing after such Award Agreement is issued, but prior to termination of Grantee's Service, upon the termination of a Grantee's Service, any Other Equity-Based Awards held by such Grantee that have not vested, or with respect to which all applicable restrictions and conditions have not lapsed, shall immediately be deemed forfeited. Upon forfeiture of any Other Equity-Based Award, the Grantee thereof shall have no further rights with respect to such Other Equity-Based Award.

12. TERMS AND CONDITIONS OF DIVIDEND EQUIVALENT RIGHTS

12.1 Dividend Equivalent Rights.

A Dividend Equivalent Right may be granted hereunder, provided that no Dividend Equivalent Rights may be granted in connection with, or related to, an Award of Options or SARs. The terms and conditions of Dividend Equivalent Rights shall be specified in the Award Agreement therefor. Dividend equivalents

credited to the holder of a Dividend Equivalent Right may be paid currently (with or without being subject to forfeiture or a repayment obligation) or may be deemed to be reinvested in additional shares of Stock or Awards, which may thereafter accrue additional Dividend Equivalent Rights (with or without being subject to forfeiture or a repayment obligation). Dividend Equivalent Rights may be settled in cash, shares of Stock, or a combination thereof, in a single installment or in multiple installments, all as determined in the sole discretion of the Committee. A Dividend Equivalent Right granted as a component of another Award may (a) provide that such Dividend Equivalent Right shall be settled upon exercise, settlement, or payment of, or lapse of restrictions on, such other Award and that such Dividend Equivalent Right shall expire or be forfeited or annulled under the same conditions as such other Award or (b) contain terms and conditions which are different from the terms and conditions of such other Award, provided that Dividend Equivalent Rights credited pursuant to a Dividend Equivalent Right granted as a component of another Award or (b) contain terms and conditions which are different from the terms and conditions of such other Award, provided that Dividend Equivalent Rights credited pursuant to a Dividend Equivalent Right granted as a component of another Award are achieved, and if such performance goals are not achieved, the Grantee of such Dividend Equivalent Rights shall promptly forfeit and, to the extent already paid or distributed, repay to the Company payments or distributions made in connection with such Dividend Equivalent Rights.

12.2 Termination of Service.

Unless the Committee provides otherwise in an Award Agreement, in another agreement with the Grantee, or otherwise in writing after such Award Agreement is issued, a Grantee's rights in all Dividend Equivalent Rights shall automatically terminate upon such Grantee's termination of Service for any reason.

13. TERMS AND CONDITIONS OF PERFORMANCE-BASED AWARDS

13.1 Grant of Performance-Based Awards.

Subject to the terms and provisions of the Plan, the Committee, at any time and from time to time, may grant Performance-Based Awards in such amounts and upon such terms as the Committee shall determine.

13.2 Value of Performance-Based Awards.

Each grant of a Performance-Based Award shall have an initial cash value or an actual or target number of shares of Stock that is established by the Committee as of the Grant Date. The Committee shall set performance goals in its discretion which, depending on the extent to which they are achieved, shall determine the value and/or number of shares of Stock subject to a Performance-Based Award that shall be paid out to the Grantee thereof.

13.3 Earning of Performance-Based Awards.

Subject to the terms of the Plan, in particular Section 13.6.3, after the applicable Performance Period has ended, the Grantee of a Performance-Based Award shall be entitled to receive a payout of the value earned under such Performance-Based Award by such Grantee over such Performance Period.

13.4 Form and Timing of Payment of Performance-Based Awards.

Payment of the value earned under Performance-Based Awards shall be made, as determined by the Committee, in the form, at the time, and in the manner described in the applicable Award Agreement. Subject to the terms of the Plan, the Committee, in its sole discretion, (i) may pay the value earned under Performance-Based Awards in the form of cash, shares of Stock, other Awards, or a combination thereof, including shares of Stock and/or Awards that are subject to any restrictions deemed appropriate by the Committee, and (ii) shall

pay the value earned under Performance-Based Awards at the close of the applicable Performance Period, or as soon as reasonably practicable after the Committee has determined that the performance goal or goals relating thereto have been achieved; provided that, unless specifically provided in the Award Agreement for such Performance-Based Awards, such payment shall occur no later than the fifteenth (15th) day of the third (3rd) month following the end of the calendar year in which such Performance Period ends. The determination of the Committee with respect to the form of payout of such Performance-Based Awards shall be set forth in the Award Agreement therefor.

13.5 Performance Conditions.

The right of a Grantee to exercise or to receive a grant or settlement of any Performance-Based Award, and the timing thereof, may be subject to such performance conditions as may be specified by the Committee. The Committee may use such business criteria and other measures of performance as it may deem appropriate in establishing any performance conditions. If and to the extent required under Code Section 162(m), any power or authority relating to an Award intended to qualify under Code Section 162(m) shall be exercised by the Committee and not by the Board.

13.6 Performance-Based Awards Granted to Designated Covered Employees.

If and to the extent that the Committee determines that a Performance-Based Award to be granted to a Grantee should constitute Qualified Performance-Based Compensation for purposes of Code Section 162(m), the grant, exercise, and/or settlement of such Performance-Based Award shall be contingent upon achievement of pre-established performance goals and other terms set forth in this **Section 13.6**.

13.6.1 Performance Goals Generally.

The performance goals for Performance-Based Awards shall consist of one or more business criteria and a targeted level or levels of performance with respect to each of such criteria, as specified by the Committee consistent with this **Section 13.6**. Performance goals shall be objective and shall otherwise meet the requirements of Code Section 162(m), including the requirement that the level or levels of performance targeted by the Committee result in the achievement of performance goals being "substantially uncertain." The Committee may determine that such Awards shall be granted, exercised, and/or settled upon achievement of any single performance goal or of two (2) or more performance goals. Performance goals may differ for Performance-Based Awards granted to any one Grantee or to different Grantees.

13.6.2 Timing For Establishing Performance Goals.

Performance goals for any Performance-Based Award shall be established not later than the earlier of (a) ninety (90) days after the beginning of any Performance Period applicable to such Performance-Based Award, and (b) the date on which twenty-five percent (25%) of any Performance Period applicable to such Performance-Based Award has expired, or at such other date as may be required or permitted for compensation payable to a Covered Employee to constitute Performance-Based Compensation.

13.6.3 Payment of Awards; Other Terms.

Payment of Performance-Based Awards shall be in cash, shares of Stock, other Awards, or a combination thereof, including shares of Stock and/or Awards that are subject to any restrictions deemed appropriate by the Committee, in each case as determined in the sole discretion of the Committee. The Committee may, in its sole discretion, reduce the amount of a payment otherwise to be made in connection

with such Performance-Based Awards. The Committee shall specify the circumstances in which such Performance-Based Awards shall be paid or forfeited in the event of termination of Service by the Grantee prior to the end of a Performance Period or settlement of such Performance-Based Awards. In the event payment of the Performance-Based Award is made in the form of another Award subject to Service-based vesting, the Committee shall specify the circumstances in which the payment Award shall be paid or forfeited in the event of a termination of Service.

13.6.4 Performance Measures.

The performance goals upon which the vesting or payment of a Performance-Based Award to a Covered Employee that is intended to qualify as Performance-Based Compensation may be conditioned shall be limited to the following Performance Measures, with or without adjustment (including pro forma adjustments):

(a) earnings before interest, taxes, depreciation, and/or amortization;

(b) earnings before interest, taxes, depreciation, and/or amortization as adjusted to exclude any one or more of the following: (i) stockbased compensation expense, (ii) income from discontinued operations, (iii) gain on cancellation of debt, (iv) debt extinguishment and related costs, (v) restructuring, separation, and/or integration charges and costs, (vi) reorganization and/or recapitalization charges and costs, (vii) impairment charges, (viii) merger-related events, (ix) gain or loss related to investments, (x) sales and use tax settlements, and (xi) gain on non-monetary transactions;

- (c) operating income, earnings, or profits;
- (d) return measures, including return on equity, assets, revenue, capital, capital employed, or investment;
- (e) pre-tax or after-tax operating income, earnings, or profits;
- (f) net income;
- (g) earnings or book value per share;

(h) cash flow(s), including (i) operating cash flow, (ii) free cash flow, (iii) levered cash flow, (iv) cash flow return on equity, and (v) cash flow return on investment;

- (i) total sales or revenues growth or targets or sales or revenues per employee, product, service, or customer;
- (j) Stock price, including growth measures and total shareholder return;
- (k) dividends;

(1) strategic business objectives, consisting of one or more objectives based on meeting specified cost targets, business expansion goals, and goals relating to acquisitions or divestitures or any combination thereof. Each goal may be expressed on an absolute and/or relative basis, may be based on or otherwise employ comparisons based on internal targets, the past performance of the Company and/or the past or current performance of other companies, and in the case of earnings-based measures, may use or employ comparisons relating to capital, shareholders' equity and/or shares outstanding, or to assets or net assets;

- (m) gross or operating margins;
- (n) productivity ratios;
- (o) costs, reductions in cost, and cost control measures;
- (p) expense targets;
- (q) market or market segment share or penetration;
- (r) financial ratios as provided in credit agreements of the Company and its subsidiaries;
- (s) working capital targets;
- (t) regulatory achievements or compliance;
- (u) customer satisfaction measurements;
- (v) execution of contractual arrangements or satisfaction of contractual requirements or milestones;
- (w) product development achievements; and
- (x) any combination of the foregoing business criteria.

Performance under any of the foregoing Performance Measures (a) may be used to measure the performance of (i) the Company, its Subsidiaries, and other Affiliates as a whole, (ii) the Company, any Subsidiary, any other Affiliate, or any combination thereof, or (iii) any one or more business units or operating segments of the Company, any Subsidiary, and/or any other Affiliate, in each case as the Committee, in its sole discretion, deems appropriate and (b) may be compared to the performance of one or more other companies or one or more published or special indices designated or approved by the Committee for such comparison, as the Committee, in its sole discretion, deems appropriate. In addition, the Committee, in its sole discretion, may select performance under the Performance Measure specified in clause (j) above for comparison to performance under one or more stock market indices designated or approved by the Committee for approved by the Committee for approved by the Committee for approved by the Committee or approved by the Committee or approved by the Committee and the performance under the Performance Measure specified in clause (j) above for comparison to performance under one or more stock market indices designated or approved by the Committee. The Committee shall also have the authority to provide for accelerated vesting of any Performance-Based Award based on the achievement of performance goals pursuant to the Performance Measures specified in this Article 13.

13.6.5 Evaluation of Performance.

The Committee may provide in any Performance-Based Award that any evaluation of performance may include or exclude any of the following events that occur during a Performance Period: (a) asset write-downs; (b) litigation or claims, judgments, or settlements; (c) the effect of changes in tax laws, accounting principles, or other laws or provisions affecting reported results; (d) any reorganization or restructuring events or programs; (e) extraordinary, non-core, non-operating, or non-recurring items and items that are either of an unusual nature or of a type that indicates infrequency of occurrence as a separate component of income from continuing operations; (f) acquisitions or divestitures; (g) foreign exchange gains and losses; (h) impact of shares of Stock purchased through share repurchase programs; (i) tax valuation allowance reversals; (j) impairment expense; and (k) environmental expense. To the extent such inclusions or exclusions affect Awards to Covered Employees that are intended to qualify as Performance-Based Compensation, such inclusions or exclusions shall be prescribed in a form that meets the requirements of Code Section 162(m) for deductibility.

13.6.6 Adjustment of Performance-Based Compensation.

The Committee shall have the sole discretion to adjust Awards that are intended to qualify as Performance-Based Compensation, either on a formula or discretionary basis, or on any combination thereof, as the Committee determines consistent with the requirements of Code Section 162(m) for deductibility.

13.6.7 Committee Discretion.

In the event that Applicable Laws change to permit Committee discretion to alter the governing Performance Measures without obtaining shareholder approval of such changes, the Committee shall have sole discretion to make such changes without obtaining shareholder approval, provided that the exercise of such discretion shall not be inconsistent with the requirements of Code Section 162(m). In addition, in the event that the Committee determines that it is advisable to grant Awards that shall not qualify as Performance-Based Compensation, the Committee may make such grants without satisfying the requirements of Code Section 162(m) and base vesting on Performance Measures other than those set forth in **Section 13.6.4**.

13.6.8 Status of Awards Under Code Section 162(m).

It is the intent of the Company that Performance-Based Awards under **Section 13.6** granted to Grantees who are designated by the Committee as likely to be Covered Employees shall, if so designated by the Committee, constitute Qualified Performance-Based Compensation. Accordingly, the terms of **Section 13.6**, including the definitions of Covered Employee and other terms used therein, shall be interpreted in a manner consistent with Code Section 162(m). If any provision of the Plan, the applicable Award Agreement, or any other agreement relating to any such Performance-Based Award does not comply or is inconsistent with the requirements of Code Section 162(m), such provision shall be construed or deemed amended to the extent necessary to conform to such requirements.

14. FORMS OF PAYMENT

14.1 General Rule.

Payment of the Option Price for the shares of Stock purchased pursuant to the exercise of an Option or the purchase price, if any, for Restricted Stock, vested Restricted Stock Units, and/or vested Deferred Stock Units shall be made in cash or in cash equivalents acceptable to the Company.

14.2 Surrender of Shares of Stock.

To the extent that the applicable Award Agreement so provides, payment of the Option Price for shares of Stock purchased pursuant to the exercise of an Option or the purchase price, if any, for Restricted Stock, vested Restricted Stock Units, and/or vested Deferred Stock Units may be made all or in part through the tender or attestation to the Company of shares of Stock, which shall be valued, for purposes of determining the extent to which such Option Price or purchase price has been paid thereby, at their Fair Market Value on the date of such tender or attestation.

14.3 Cashless Exercise.

To the extent permitted by Applicable Laws and to the extent the Award Agreement so provides, payment of the Option Price for shares of Stock purchased pursuant to the exercise of an Option may be made all or in part by delivery (on a form acceptable to the Committee) of an irrevocable direction to a licensed securities broker acceptable to the Company to sell shares of Stock and to deliver all or part of the proceeds of such sale to the Company in payment of such Option Price and/or any withholding taxes described in Section 18.3.

14.4 Other Forms of Payment.

To the extent that the applicable Award Agreement so provides and/or unless otherwise specified in an Award Agreement, payment of the Option Price for shares of Stock purchased pursuant to exercise of an Option or the purchase price, if any, for Restricted Stock, vested Restricted Stock Units, and/or vested Deferred Stock Units may be made in any other form that is consistent with Applicable Laws, including (a) with respect to Restricted Stock, vested Restricted Stock Units, and/or vested Deferred Stock Units, and/or vested Deferred Stock Units, and/or vested Deferred Stock Units only, Service rendered or to be rendered by the Grantee thereof to the Company or an Affiliate and (b) with the consent of the Company, by withholding the number of shares of Stock that would otherwise vest or be issuable in an amount equal in value to the Option Price or purchase price and/or the required tax withholding amount.

15. REQUIREMENTS OF LAW

15.1 General.

The Company shall not be required to offer, sell, or issue any shares of Stock under any Award, whether pursuant to the exercise of an Option, a SAR, or otherwise, if the offer, sale, or issuance of such shares of Stock would constitute a violation by the Grantee, the Company, an Affiliate, or any other Person of any provision of the Company's certificate of incorporation or bylaws or of Applicable Laws, including any federal or state securities laws or regulations. If at any time the Company shall determine, in its discretion, that the listing, registration, or qualification of any shares of Stock subject to an Award upon any Stock Exchange or Securities Market or under any governmental regulatory body is necessary or desirable as a condition of, or in connection with, the offering, sale, issuance, or purchase of shares of Stock in connection with any Award, no shares of Stock may be offered, sold, or issued to the Grantee or any other Person under such Award, whether pursuant to the exercise of an Option, a SAR, or otherwise, unless such listing, registration, or qualification shall have been effected or obtained free of any conditions not acceptable to the Company, and any delay caused thereby shall in no way affect the date of termination of such Award. Without limiting the generality of the foregoing, upon the exercise of any Option or any SAR that may be settled in shares of Stock or the delivery of any shares of Stock underlying an Award, unless a registration statement under the Securities Act is in effect with respect to the shares of Stock subject to such Award, the Company shall not be required to offer, sell, or issue such shares of Stock unless the Committee shall have received evidence satisfactory to it that the Grantee or any other Person exercising such Option or SAR or accepting delivery of such shares may acquire such shares of Stock pursuant to an exemption from registration under the Securities Act. Any determination by the Committee in connection with the foregoing shall be final, binding, and conclusive. The Company may register, but shall in no event be obligated to register, any shares of Stock or other securities issuable pursuant to the Plan pursuant to the Securities Act. The Company shall not be obligated to take any affirmative action in order to cause the exercise of an Option or a SAR or the issuance of shares of Stock or other securities issuable pursuant to the Plan or any Award to comply with any Applicable Laws. As to any jurisdiction that expressly imposes the requirement that an Option or SAR that may be settled in shares of Stock shall not be exercisable until the shares of Stock subject to such Option or SAR are registered under the securities laws thereof or are exempt from such registration, the exercise of such Option or SAR under circumstances in which the laws of such jurisdiction apply shall be deemed conditioned upon the effectiveness of such registration or the availability of such an exemption.

15.2 Rule 16b-3.

During any time when the Company has any class of common equity securities registered under Section 12 of the Exchange Act, it is the intention of the Company that Awards pursuant to the Plan and the exercise of Options and SARs granted hereunder that would otherwise be subject to Section 16(b) of the Exchange Act shall qualify for the exemption provided by Rule 16b-3 under the Exchange Act. To the extent that any provision of the Plan or action by the Committee does not comply with the requirements of such Rule 16b-3, such provision or action shall be deemed inoperative with respect to such Awards to the extent permitted by Applicable Laws and deemed advisable by the Committee and shall not affect the validity of the Plan. In the event that such Rule 16b-3 is revised or replaced, the Committee may exercise its discretion to modify the Plan in any respect necessary or advisable in its judgment to satisfy the requirements of, or to permit the Company to avail itself of the benefits of, the revised exemption or its replacement.

16. EFFECT OF CHANGES IN CAPITALIZATION

16.1 Changes in Stock.

If the number of outstanding shares of Stock is increased or decreased or the shares of Stock are changed into or exchanged for a different number of shares or kind of Capital Stock or other securities of the Company on account of any merger, reorganization, recapitalization, reclassification, stock split, reverse stock split, spin-off, combination of stock, exchange of stock, stock dividend or other distribution payable in capital stock, or other increase or decrease in shares of Stock effected without receipt of consideration by the Company occurring after the Effective Date, the number and kinds of shares of Capital Stock for which grants of Options and other Awards may be made under the Plan, including the Share Limit set forth in Section 4.1, the individual share limitations set forth in Section 6.2, and the share limitations in Section 8.2(c), shall be adjusted proportionately and accordingly by the Committee. In addition, the number and kind of shares of Capital Stock for which Awards are outstanding shall be adjusted proportionately and accordingly by the Committee so that the proportionate interest of the Grantee therein immediately following such event shall, to the extent practicable, be the same as immediately before such event. Any such adjustment in outstanding Options or SARs shall not change the aggregate Option Price or SAR Price payable with respect to shares that are subject to the unexercised portion of such outstanding Options or SARs, as applicable, but shall include a corresponding proportionate adjustment in the per share Option Price or SAR Price, as the case may be. The conversion of any convertible securities of the Company shall not be treated as an increase in shares effected without receipt of consideration. Notwithstanding the foregoing, in the event of any distribution to the Company's shareholders of securities of any other entity or other assets (including an extraordinary dividend, but excluding a non-extraordinary dividend, declared and paid by the Company) without receipt of consideration by the Company, the Board or the Committee constituted pursuant to Section 3.1.2 shall, in such manner as the Board or the Committee deems appropriate, adjust (a) the Share limit set forth in Section 4.1, the individual share limitations set forth in Section 6.2, and the share limitations in Section 8.2(c), (b) the number and kind of shares of Capital Stock subject to outstanding Awards and/or (c) the aggregate and per share Option Price of outstanding Options and the aggregate and per share SAR Price of outstanding SARs as required to reflect such distribution.

16.2 Reorganization in Which the Company Is the Surviving Entity Which Does Not Constitute a Change in Control.

Subject to Section 16.3, if the Company shall undergo any reorganization, merger, or consolidation of the Company with one or more other entities which does not constitute a Change in Control, any Award theretofore granted pursuant to the Plan shall pertain to and apply to the Capital Stock to which a holder of the number of shares of Stock subject to such Award would have been entitled immediately following such

reorganization, merger, or consolidation, with a corresponding proportionate adjustment of the per share Option Price or SAR Price of any outstanding Option or SAR so that the aggregate Option Price or SAR Price thereafter shall be the same as the aggregate Option Price or SAR Price of the shares of Stock remaining subject to the Option or SAR as in effect immediately prior to such reorganization, merger, or consolidation. Subject to any contrary language in an Award Agreement, in another agreement with the Grantee, or as otherwise set forth in writing, any restrictions applicable to such Award shall apply as well to any replacement shares of Capital Stock subject to such Award, or received by the Grantee, as a result of such reorganization, merger, or consolidation. In the event of any reorganization, merger, or consolidation of the Company referred to in this Section 16.2, Performance-Based Awards shall be adjusted (including any adjustment to the Performance Measures applicable to such Awards would have been entitled to receive immediately following such reorganization, merger, or consolidation.

16.3 Change in Control in Which Awards Are Not Assumed.

Except as otherwise provided in the applicable Award Agreement, in another agreement with the Grantee, or as otherwise set forth in writing, upon the occurrence of a Change in Control in which outstanding Awards are not being assumed or continued, the following provisions shall apply to such Award, to the extent not assumed or continued:

(a) Immediately prior to the occurrence of such Change in Control, in each case with the exception of Performance-Based Awards, all outstanding shares of Restricted Stock, and all Restricted Stock Units, Deferred Stock Units, and Dividend Equivalent Rights shall be deemed to have vested, and all shares of Stock and/or cash subject to such Awards shall be delivered; and either or both of the following two (2) actions shall be taken:

(i) At least fifteen (15) days prior to the scheduled consummation of such Change in Control, all Options and SARs outstanding hereunder shall become immediately exercisable and shall remain exercisable for a period of fifteen (15) days. Any exercise of an Option or SAR during this fifteen (15)-day period shall be conditioned upon the consummation of the applicable Change in Control and shall be effective only immediately before the consummation thereof, and upon consummation of such Change in Control, the Plan and all outstanding but unexercised Options and SARs shall terminate, with or without consideration (including, without limitation, consideration in accordance with clause (ii) below) as determined by the Committee in its sole discretion. The Committee shall send notice of an event that shall result in such a termination to all Persons who hold Options and SARs not later than the time at which the Company gives notice thereof to its shareholders.

and/or

(ii) The Committee may elect, in its sole discretion, to cancel any outstanding Awards of Options, SARs, Restricted Stock, Restricted Stock Units, Deferred Stock Units, and/or Dividend Equivalent Rights and pay or deliver, or cause to be paid or delivered, to the holder thereof an amount in cash or Capital Stock having a value (as determined by the Committee acting in good faith), in the case of Restricted Stock, Restricted Stock Units, Deferred Stock Units, and Dividend Equivalent Rights (for shares of Stock subject thereto), equal to the formula or fixed price per share paid to holders of shares of Stock pursuant to such Change in Control and, in the case of Options or SARs, equal to the product of the number of shares of Stock subject to such Options or SARs multiplied by the amount, if any, by which (x) the formula or fixed price per share paid to holders of shares of Stock pursuant to such transaction exceeds (y) the Option Price or SAR Price applicable to such Options or SARs.

(b) For Performance-Based Awards, if less than half of the Performance Period has lapsed, such Performance-Based Awards shall be treated as though target performance has been achieved. If at least half of the Performance Period has lapsed, actual performance to date shall be determined as of a date reasonably proximal to the date of consummation of the Change in Control as determined by the Committee, in its sole discretion, and that level of performance thus determined shall be treated as achieved immediately prior to occurrence of the Change in Control. For purposes of the preceding sentence, if, based on the discretion of the Committee, actual performance is not determinable, the Performance-Based Awards shall be treated as though target performance has been achieved. After application of this **Section 16.3(b)**, if any Awards arise from application of this **Article 16**, such Awards shall be settled under the applicable provision of **Section 16.3(a)**.

(c) Other Equity-Based Awards shall be governed by the terms of the applicable Award Agreement.

16.4 Change in Control in Which Awards Are Assumed.

Except as otherwise provided in the applicable Award Agreement, in another agreement with the Grantee, or as otherwise set forth in writing, upon the occurrence of a Change in Control in which outstanding Awards are being assumed or continued, the following provisions shall apply to such Award, to the extent assumed or continued:

The Plan and the Options, SARs, Restricted Stock, Restricted Stock Units, Deferred Stock Units, Dividend Equivalent Rights, and Other Equity-Based Awards granted under the Plan shall continue in the manner and under the terms so provided in the event of any Change in Control to the extent that provision is made in writing in connection with such Change in Control for the assumption or continuation of such Options, SARs, Restricted Stock, Restricted Stock Units, Deferred Stock Units, Dividend Equivalent Rights, and Other Equity-Based Awards, or for the substitution for such Options, SARs, Restricted Stock, Restricted Stock, Restricted Stock Units, Deferred Stock Units, Deferred Stock Units, Dividend Equivalent Rights, and Other Equity-Based Awards of new stock options, stock appreciation rights, restricted stock, restricted stock units, deferred stock units, dividend equivalent rights, and other equity-based awards relating to the Capital Stock of a successor entity, or a parent or subsidiary thereof, with appropriate adjustments as to the number of shares (disregarding any consideration that is not common stock) and exercise prices of options and stock appreciation rights.

In the event an Award is assumed, continued, or substituted upon the consummation of any Change in Control and the Service of such Grantee with the Company or an Affiliate is terminated without Cause within one (1) year (or such longer or shorter period as may be determined by the Committee) following the consummation of such Change in Control, such Award shall be fully vested and may be exercised in full, to the extent applicable, beginning on the date of such termination and for the one (1)-year period (or such longer or shorter period as may be determined by the Committee) immediately following such termination or for such longer period as the Committee shall determine.

16.5 Adjustments.

Adjustments under this **Article 16** related to shares of Stock or other Capital Stock of the Company shall be made by the Committee, whose determination in that respect shall be final, binding, and conclusive. No fractional shares or other securities shall be issued pursuant to any such adjustment, and any fractions resulting from any such adjustment shall be eliminated in each case by rounding downward to the nearest whole share. The Committee may provide in the applicable Award Agreement as of the Grant Date, in another agreement with the Grantee, or otherwise in writing at any time thereafter with the consent of the Grantee, for different provisions to apply to an Award in place of those provided in **Sections 16.1, 16.2, 16.3**, and **16.4**.

This Article 16 shall not limit the Committee's ability to provide for alternative treatment of Awards outstanding under the Plan in the event of an internal reorganization change in control event involving the Company that is not a Change in Control.

16.6 No Limitations on Company.

The making of Awards pursuant to the Plan shall not affect or limit in any way the right or power of the Company to make adjustments, reclassifications, reorganizations, or changes of its capital or business structure or to merge, consolidate, dissolve, or liquidate, or to sell or transfer all or any part of its business or assets (including all or any part of the business or assets of any Subsidiary or other Affiliate) or to engage in any other transaction or activity.

17. PARACHUTE LIMITATIONS

If any Grantee is a Disqualified Individual, then, notwithstanding any other provision of the Plan or of any Other Agreement to the contrary and notwithstanding any Benefit Arrangement, any right of the Grantee to any exercise, vesting, payment, or benefit under the Plan shall be reduced or eliminated:

(a) to the extent that such right to exercise, vesting, payment, or benefit, taking into account all other rights, payments, or benefits to or for the Grantee under the Plan, all Other Agreements, and all Benefit Arrangements, would cause any exercise, vesting, payment, or benefit to the Grantee under the Plan to be considered a Parachute Payment; and

(b) if, as a result of receiving such Parachute Payment, the aggregate after-tax amounts received by the Grantee from the Company under the Plan, all Other Agreements, and all Benefit Arrangements would be less than the maximum after-tax amount that could be received by the Grantee without causing any such payment or benefit to be considered a Parachute Payment.

Except as required by Code Section 409A or to the extent that Code Section 409A permits discretion, the Committee shall have the right, in the Committee's sole discretion, to designate those rights, payments, or benefits under the Plan, all Other Agreements, and all Benefit Arrangements that should be reduced or eliminated so as to avoid having such rights, payments, or benefits be considered a Parachute Payment; provided, however, to the extent any payment or benefit constitutes deferred compensation under Code Section 409A, in order to comply with Code Section 409A, the Company shall instead accomplish such reduction by first reducing or eliminating any cash payments (with the payments to be made furthest in the future being reduced first), then by reducing or eliminating any accelerated vesting of Performance-Based Awards, then by reducing or eliminating any accelerated vesting of Restricted Stock, Restricted Stock Units, or Deferred Stock Units, then by reducing or eliminating any other remaining Parachute Payments.

18. GENERAL PROVISIONS

18.1 Disclaimer of Rights.

No provision in the Plan, any Award, or any Award Agreement shall be construed (a) to confer upon any individual the right to remain in the Service of the Company or an Affiliate, (b) to interfere in any way with any contractual or other right or authority of the Company or an Affiliate either to increase or decrease the compensation or other payments to any Person at any time, or (c) to terminate any Service or other relationship between any Person and the Company or an Affiliate. In addition, notwithstanding any provision of the Plan to

the contrary, unless otherwise stated in the applicable Award Agreement, in another agreement with the Grantee, or otherwise in writing, no Award granted under the Plan shall be affected by any change of duties or position of the Grantee thereof, so long as such Grantee continues to provide Service. The obligation of the Company to pay any benefits pursuant to the Plan shall be interpreted as a contractual obligation to pay only those amounts provided herein, in the manner and under the conditions prescribed herein. The Plan and Awards shall in no way be interpreted to require the Company to transfer any amounts to a third-party trustee or otherwise hold any amounts in trust or escrow for payment to any Grantee or beneficiary under the terms of the Plan.

18.2 Nonexclusivity of the Plan.

Neither the adoption of the Plan nor the submission of the Plan to the shareholders of the Company for approval shall be construed as creating any limitations upon the right and authority of the Board or the Committee to adopt such other incentive compensation arrangements (which arrangements may be applicable either generally to a class or classes of individuals or specifically to a particular individual or particular individuals) as the Board or the Committee in their discretion determine desirable.

18.3 Withholding Taxes.

The Company or an Affiliate, as the case may be, shall have the right to deduct from payments of any kind otherwise due to a Grantee any federal, state, or local taxes of any kind required by Applicable Laws to be withheld with respect to the vesting of or other lapse of restrictions applicable to an Award or upon the issuance of any shares of Stock upon the exercise of an Option or pursuant to any other Award. At the time of such vesting, lapse, or exercise, the Grantee shall pay in cash to the Company or an Affiliate, as the case may be, any amount that the Company or such Affiliate may reasonably determine to be necessary to satisfy such withholding obligation; provided that if there is a same-day sale of shares of Stock subject to an Award, the Grantee shall pay such withholding obligation on the day on which such same-day sale is completed. Subject to the prior approval of the Company or an Affiliate, which may be withheld by the Company or such Affiliate, as the case may be, in its sole discretion, the Grantee may elect to satisfy such withholding obligation, in whole or in part, (a) by causing the Company or such Affiliate to withhold shares of Stock otherwise issuable to the Grantee or (b) by delivering to the Company or such Affiliate shares of Stock already owned by the Grantee. The shares of Stock so withheld or delivered shall have an aggregate Fair Market Value equal to such withholding obligation. The Fair Market Value of the shares of Stock used to satisfy such withholding obligation shall be determined by the Company or such Affiliate as of the date on which the amount of tax to be withheld is to be determined. A Grantee who has made an election pursuant to this Section 18.3 may satisfy such Grantee's withholding obligation only with shares of Stock that are not subject to any repurchase, forfeiture, unfulfilled vesting, or other similar requirements. The maximum number of shares of Stock that may be withheld from any Award to satisfy any federal, state, or local tax withholding requirements upon the exercise, vesting, or lapse of restrictions applicable to any Award or payment of shares of Stock pursuant to such Award, as applicable, may not exceed such number of shares of Stock having a Fair Market Value equal to the minimum statutory amount (or, if permitted by the Company, such other rate as will not cause adverse accounting consequences and is permitted under applicable IRS withholding rules) required by the Company or the applicable Affiliate to be withheld and paid to any such federal, state, or local taxing authority with respect to such exercise, vesting, lapse of restrictions, or payment of shares of Stock.

18.4 Captions.

The use of captions in the Plan or any Award Agreement is for convenience of reference only and shall not affect the meaning of any provision of the Plan or such Award Agreement.

18.5 Construction.

Unless the context otherwise requires, all references in the Plan to "including" shall mean "including without limitation."

18.6 Other Provisions.

Each Award granted under the Plan may contain such other terms and conditions not inconsistent with the Plan as may be determined by the Committee, in its sole discretion.

18.7 Number and Gender.

With respect to words used in the Plan, the singular form shall include the plural form, and the masculine gender shall include the feminine gender, as the context requires.

18.8 Severability.

If any provision of the Plan or any Award Agreement shall be determined to be illegal or unenforceable by any court of law in any jurisdiction, the remaining provisions hereof and thereof shall be severable and enforceable in accordance with their terms, and all provisions shall remain enforceable in any other jurisdiction.

18.9 Governing Law.

The validity and construction of the Plan and the instruments evidencing the Awards hereunder shall be governed by, and construed and interpreted in accordance with, the laws of the State of Delaware, other than any conflicts or choice of law rule or principle that might otherwise refer construction or interpretation of the Plan and the instruments evidencing the Awards granted hereunder to the substantive laws of any other jurisdiction.

18.10 Foreign Jurisdictions.

To the extent the Committee determines that the material terms set by the Committee imposed by the Plan preclude the achievement of the material purposes of the Plan in jurisdictions outside the United States, the Committee shall have the authority and discretion to modify those terms and provide for such additional terms and conditions as the Committee determines to be necessary, appropriate, or desirable to accommodate differences in local law, policy, or custom or to facilitate administration of the Plan. The Committee may adopt or approve sub-plans, appendices, or supplements to, or amendments, restatements, or alternative versions of the Plan as in effect for any other purposes. The special terms and any appendices, supplements, amendments, restatements, or alternative versions, however, shall not include any provisions that are inconsistent with the terms of the Plan as in effect, unless the Plan could have been amended to eliminate such inconsistency without further approval by the Company's shareholders.

18.11 Section 409A of the Code.

The Plan is intended to comply with Code Section 409A to the extent subject thereto, and, accordingly, to the maximum extent permitted, the Plan shall be interpreted and administered to be in compliance with Code Section 409A. Any payments described in the Plan that are due within the Short-Term Deferral Period shall not be treated as deferred compensation unless Applicable Laws require otherwise. Notwithstanding any provision of the Plan to the contrary, to the extent required to avoid accelerated taxation and tax penalties under Code Section 409A, amounts that would otherwise be payable and benefits that would otherwise be provided pursuant to the Plan during the six (6)-month period immediately following the

Grantee's Separation from Service shall instead be paid on the first payroll date after the six (6)-month anniversary of the Grantee's Separation from Service (or the Grantee's death, if earlier).

Furthermore, notwithstanding anything in the Plan to the contrary, in the case of an Award that is characterized as deferred compensation under Code Section 409A, and pursuant to which settlement and delivery of the cash or shares of Stock subject to the Award is triggered based on a Change in Control, in no event shall a Change in Control be deemed to have occurred for purposes of such settlement and delivery of cash or shares of Stock if the transaction is not also a "change in the ownership or effective control of" the Company or "a change in the ownership of a substantial portion of the assets of" the Company as determined under Treasury Regulation Section 1.409A-3(i)(5) (without regard to any alternative definition thereunder). If an Award characterized as deferred compensation under Code Section 409A is not settled and delivery triggering event that is a permissible triggering event under Code Section 409A. No provision of this paragraph shall in any way affect the determination of a Change in Control for purposes of vesting in an Award that is characterized as deferred compensation under Code Section 409A.

Notwithstanding the foregoing, neither the Company nor the Committee shall have any obligation to take any action to prevent the assessment of any excise tax or penalty on any Grantee under Code Section 409A, and neither the Company or an Affiliate nor the Board or the Committee shall have any liability to any Grantee for such tax or penalty.

18.12 Limitation on Liability.

No member of the Board or the Committee shall be liable for any action or determination made in good faith with respect to the Plan, any Award, or any Award Agreement. Notwithstanding any provision of the Plan to the contrary, neither the Company, an Affiliate, the Board, the Committee, nor any person acting on behalf of the Company, an Affiliate, the Board, or the Committee shall be liable to any Grantee or to the estate or beneficiary of any Grantee or to any other holder of an Award under the Plan by reason of any acceleration of income, or any additional tax (including any interest and penalties), asserted by reason of the failure of an Award to satisfy the requirements of Code Section 422 or Code Section 409A or by reason of Code Section 4999, or otherwise asserted with respect to the Award; provided, that this **Section 18.12** shall not affect any of the rights or obligations set forth in an applicable agreement between the Grantee and the Company or an Affiliate.

LABCORP HOLDINGS INC. AMENDED AND RESTATED 2016 EMPLOYEE STOCK PURCHASE PLAN As Amended and Restated, Effective May 17, 2024

1. PURPOSE AND INTERPRETATION

(a) The purpose of the Plan is to encourage and to enable Eligible Employees of the Company and its Participating Affiliates, through after-tax payroll deductions or periodic cash contributions, to acquire proprietary interests in the Company through the purchase and ownership of shares of Stock. The Plan is intended to benefit the Company and its shareholders (a) by incentivizing Participants to contribute to the success of the Company and to operate and manage the Company's business in a manner that will provide for the Company's long-term growth and profitability and that will benefit its shareholders and other important stakeholders and (b) by encouraging Participants to remain in the employ of the Company or its Participating Affiliates.

(b) The Plan and the Options granted under the Plan are intended to satisfy the requirements for an "employee stock purchase plan" under Code Section 423. Notwithstanding the foregoing, the Company makes no undertaking to, nor representation that it will, maintain the qualified status of the Plan or any Options granted under the Plan. In addition, Options that do not satisfy the requirements for an "employee stock purchase plan" under Code Section 423 may be granted under the Plan pursuant to the rules, procedures, or sub-plans adopted by the Administrator, in its sole discretion, for certain Eligible Employees.

2. DEFINITIONS

(a) "Account" shall mean a bookkeeping account established and maintained to record the amount of funds accumulated pursuant to the Plan with respect to a Participant for the purpose of purchasing shares of Stock under the Plan.

(b) "Administrator" shall mean the Board, the Compensation Committee of the Board, or any other committee of the Board designated by the Board.

(c) "Board" shall mean the Board of Directors of the Company.

(d) "**Code**" shall mean the Internal Revenue Code of 1986, as amended, as now in effect or as hereafter amended, and any successor thereto. References in the Plan to any Code Section shall be deemed to include, as applicable, regulations and guidance promulgated under such Code Section.

(e) "Company" shall mean Labcorp Holdings Inc., a Delaware corporation, and any successor thereto.

(f) "Custodian" shall mean the third-party administrator designated by the Administrator from time to time.

(g) "Effective Date" shall mean May 11, 2016.

(h) "Eligible Compensation" shall mean, unless otherwise established by the Administrator prior to the start of an Offering Period, regular base pay (including any shift differentials) but excludes any bonus, overtime payment, sales commission, contribution to any Code Section 125 or 401(k) plan or to the extent applicable, its equivalence under the laws of a foreign jurisdiction, or other form of extra compensation.

(i) "Eligible Employee" shall mean a natural person who has been an employee (including an officer) of the Company or a Participating Affiliate for at least six (6) months as of an Offering Date, except the following, who shall not be eligible to participate under the Plan: (i) an employee whose customary

employment is twenty (20) hours or less per week, (ii) an employee whose customary employment is for not more than five (5) months in any calendar year, (iii) an employee who, after exercising his or her rights to purchase shares of Stock under the Plan, would own (directly or by attribution pursuant to Code Section 424(d)) shares of Stock (including shares that may be acquired under any outstanding Options) representing five percent (5%) or more of the total combined voting power of all classes of stock of the Company, (iv) an employee who is a citizen or resident of a foreign jurisdiction (without regard to whether such employee is also a U.S. citizen or resident alien), if the grant of an Option under the Plan or an Offering Period to such employee is prohibited under the laws of such foreign jurisdiction or compliance with the laws of such foreign jurisdiction would cause the Plan or an Offering Period to violate the requirements of Code Section 423 and (v) any other natural person whom the Administrator determine to exclude from an offering designed to satisfy the requirements of Code Section 423 provided such exclusion is permitted by Code Section 423 and the guidance issued thereunder as well as, to the extent applicable, the laws of a foreign jurisdiction. Notwithstanding the foregoing, for purposes of a Non-423(b) Offering under the Plan, if any, the Administrator shall have the authority, in its sole discretion, to establish a different definition of Eligible Employee as it may deem advisable or necessary.

(j) "Enrollment Form" shall mean the agreement(s) between the Company and an Eligible Employee, in such written, electronic, or other format and/or pursuant to such written, electronic, or other process as may be established by the Administrator from time to time, pursuant to which an Eligible Employee elects to participate in the Plan or to which a Participant elects to make changes with respect to the Participant's participation as permitted by the Plan.

(k) "Enrollment Period" shall mean that period of time prescribed by the Administrator, which period shall conclude prior to the Offering Date, during which Eligible Employees may elect to participate in an Offering Period. The duration and timing of Enrollment Periods may be changed or modified by the Administrator from time to time.

(I) "Fair Market Value" shall mean the value of each share of Stock subject to the Plan on a given date determined as follows: (i) if on such date the shares of Stock are listed on an established national or regional stock exchange or are publicly traded on an established securities market, the Fair Market Value of a share of Stock shall be the average of the high and low sales price of the Stock on such exchange or in such market (the exchange or market selected by the Administrator if there is more than one such exchange or market) on such date or, if such date is not a Trading Day, on the Trading Day immediately preceding such date, or, if no sale of the shares of Stock is reported for such trading day, on the next preceding day on which any sale shall have been reported; or (ii) if the shares of Stock are not listed on such an exchange or traded on such a market, the Fair Market Value of the shares of Stock shall be determined by the Board in good faith.

(m) "Holding Period" shall have the meaning set forth in Section 10(c)(i).

(n) "**Non-423(b) Offering**" shall mean the rules, procedures, or sub-plans, if any, adopted by the Administrator, in its sole discretion, as a part of the Plan, pursuant to which Options that do not satisfy the requirements for "employee stock purchase plans" that are set forth under Code Section 423 may be granted to Eligible Employees as a separate offering under the Plan.

(o) "Offering Date" shall mean the first day of any Offering Period under the Plan.

(p) "Offering Period" shall mean the period determined by the Administrator pursuant to <u>Section 7</u>, which period shall not exceed twenty-seven (27) months, during which payroll deductions or periodic cash contributions are accumulated for the purpose of purchasing Stock under the Plan.

(q) "Option" shall mean the right granted to Participants to purchase shares of Stock pursuant to an offering under the Plan.

(r) "**Outstanding Election**" shall mean a Participant's then-current election to purchase shares of Stock in an Offering Period, or that part of such an election which has not been cancelled (including any voluntary cancellation under <u>Section 5</u> and deemed cancellation under <u>Section 11</u>) prior to the close of business on the last Trading Day of the Offering Period (or if an Offering Period has multiple Purchase Periods, the last Trading Day of the Purchase Period) or such other date as determined by the Administrator.

(s) "Participating Affiliate" shall mean any Subsidiary designated by the Administrator from time to time, in its sole discretion, whose employees may participate in the Plan or in a specific Offering Period under the Plan, if such employees otherwise qualify as Eligible Employees.

(t) "Participant" shall mean an Eligible Employee who has elected to participate in the Plan pursuant to Section 5.

(u) "Plan" shall mean this Labcorp Holdings, Inc. Amended and Restated 2016 Employee Stock Purchase Plan, as it may be further amended from time to time.

(v) "Prior Plan" shall mean the Laboratory Corporation of America Holdings 1997 Employee Stock Purchase Plan, as amended.

(w) "**Purchase Period**" shall mean the period during an Offering Period designated by the Administrator on the last Trading Day of which purchases of Stock are made under the Plan. An Offering Period may have one or more Purchase Periods.

(x) "**Purchase Price**" shall mean the purchase price of each share of Stock purchased under the Plan; *provided, however*, that the Purchase Price shall not be less than the lesser of eighty-five percent (85%) of the average of the high and low sales price of the Common Stock on the New York Stock Exchange on the Offering Date or the last Trading Day of the Offering Period (or if an Offering Period has multiple Purchase Periods, on the last Trading Day of the Purchase Period).

(y) "Stock" shall mean the common stock, par value \$0.10 per share, of the Company, or any security into which shares of Stock may be changed or for which shares of Stock may be exchanged as provided in <u>Section 12</u>.

(z) "**Subsidiary**" shall mean any corporation (other than the Company) in an unbroken chain of corporations beginning with the Company if each of the corporations other than the last corporation in the unbroken chain owns stock possessing fifty percent (50%) or more of the total combined voting power of all classes of stock in one of the other corporations in such chain. A corporation that attains the status of a Subsidiary on a date on or after May 17, 2024 shall be considered a Subsidiary commencing as of such date.

(aa) "**Termination of Employment**" shall mean, with respect to a Participant, a cessation of the employee-employer relationship between the Participant and the Company or a Participating Affiliate for any reason as well as the occurrences included below that would constitute termination of employment at law,

(i) including, without limitation, (A) a voluntary or involuntary termination of employment by resignation, discharge, dismissal, redundancy, expiry of fixed-term contract, death, disability, retirement, (B) or the disaffiliation of a Subsidiary, (C) unless otherwise determined or provided by the Administrator, a transfer of employment to a Subsidiary that is not a Participating Affiliate as of the first day immediately following the three (3)-month period following such transfer, and (D) a termination of employment where the individual continues to provide certain services to the Company or a Subsidiary in a non-employee role, but

(ii) excluding (A) such termination of employment where there is a simultaneous reemployment of the Participant by the Company or a Participating Affiliate, and (B) any bona fide and Company-approved or Participating Affiliate-approved leave of absence, such as family leave, medical leave, personal leave, and military leave, or such other leave that meets the requirements of Treasury Regulations section 1.421-1(h)(2) or, to the extent applicable, its equivalence under the laws of a foreign jurisdiction; *provided, however*, where the period of leave exceeds three (3) months and the employee's right to reemployment is not guaranteed either by statute or by contract, the employee-employer relationship will be deemed to have terminated on the first day immediately following such three (3)-month period.

(bb) "Trading Day" shall mean a day on which the New York Stock Exchange is open for trading.

3. SHARES SUBJECT TO THE PLAN

(a) <u>Share Reserve</u>. Subject to adjustment as provided in <u>Section 12</u>, the maximum number of shares of Stock that may be issued pursuant to Options granted under the Plan (including any Non-423(b) Offering established hereunder) is two million five hundred thousand (2,500,000) shares. The shares of Stock reserved for issuance under the Plan may be authorized but unissued shares, treasury shares, or shares purchased on the open market.

(b) <u>Participation Adjustment as a Result of the Share Reserve</u>. If the Administrator determines that the total number of shares of Stock remaining available under the Plan is insufficient to permit the number of shares of Stock to be purchased by all Participants on the last Trading Day of an Offering Period (or if an Offering Period has multiple Purchase Periods, on the last Trading Day of the Purchase Period) pursuant to <u>Section 9</u>, the Administrator shall make a participation adjustment, where the number of shares of Stock purchasable by all Participants shall be reduced proportionately in as uniform and equitable a manner as is reasonably practicable, as determined in the Administrator's sole discretion. After such adjustment, the Administrator shall refund in cash all affected Participants' Account balances for such Offering Period as soon as practicable thereafter.

(c) <u>Applicable Law Limitations on the Share Reserve</u>. If the Administrator determines that some or all of the shares of Stock to be purchased by Participants on the last Trading Day of an Offering Period (or if an Offering Period has multiple Purchase Periods, the last Trading Day of the Purchase Period) would not be issued in accordance with applicable laws or any approval by any regulatory body as may be required or the shares of Stock would not be issued pursuant to an effective Form S-8 registration statement or that the issuance of some or all of such shares of Stock pursuant to a Form S-8 registration statement is not advisable due to the risk that such issuance will violate applicable laws, the Administrator may, without Participants' consent, terminate any outstanding Offering Period and the Options granted thereunder and refund in cash all affected Participants' Account balances for such Offering Period as soon as practicable thereafter.

4. ADMINISTRATION

(a) <u>Generally</u>. The Plan shall be administered under the direction of the Administrator. Subject to the express provisions of the Plan, the Administrator shall have full authority, in its sole discretion, to take any actions it deems necessary or advisable for the administration of the Plan, including, without limitation:

(i) Interpreting and construing the Plan and Options granted under the Plan; prescribing, adopting, amending, waiving, and rescinding rules and regulations it deems appropriate to implement the Plan, including amending any outstanding Option, as it may deem advisable or necessary to comply with applicable laws; correcting any defect or supplying any omission or reconciling any inconsistency in the Plan or Options granted under the Plan; and making all other decisions relating to the operation of the Plan;

- (ii) Establishing the timing and length of Offering Periods and Purchase Periods;
- (iii) Establishing minimum and maximum contribution rates;

(iv) Establishing new or changing existing limits on the number of shares of Stock a Participant may elect to purchase with respect to any Offering Period, if such limits are announced prior to the first Offering Period to be affected;

(v) Adopting such rules, procedures, or sub-plans as may be deemed advisable or necessary to comply with the laws of countries other than the United States, to allow for tax-preferred treatment of the Options or otherwise to provide for the participation by Eligible Employees who reside outside of the United States, including determining which Eligible Employees are eligible to participate in the Non-423(b) Offering or other sub-plans established by the Administrator;

(vi) Establishing the exchange ratio applicable to amounts withheld in a currency other than U.S. dollars and permitting payroll withholding in excess of the amount designated by a Participant in order to adjust for delays or mistakes in the processing of properly completed Enrollment Forms; and

(vii) Furnishing to the Custodian such information as the Custodian may require.

The Administrator's determinations under the Plan shall be final, binding, and conclusive upon all persons.

(b) <u>Custodian</u>. If the Administrator designates a Custodian for the Plan, the Custodian shall act as custodian under the Plan and shall perform such duties as requested by the Administrator in accordance with any agreement between the Company and the Custodian. The Custodian shall establish and maintain, as agent for each Participant, an Account and any subaccounts as may be necessary or desirable for the administration of the Plan.

(c) <u>No Liability</u>. Neither the Board, the Compensation Committee of the Board, any other committee of the Board, or the Custodian, nor any of their agents or designees, shall be liable to any person (i) for any act, failure to act, or determination made in good faith with respect to the Plan or Options granted under the Plan or (ii) for any tax (including any interest and penalties) by reason of the failure of the Plan, an Option, or an Offering Period to satisfy the requirements of Code Section 423, the failure of the Plan, or shares of Stock purchased or deemed purchased under the Plan.

5. PARTICIPATION IN THE PLAN AND IN AN OFFERING PERIOD

(a) <u>Generally</u>. An Eligible Employee may become a Participant for an Offering Period under the Plan by completing the prescribed Enrollment Form and submitting such Enrollment Form to the Company (or the Company's designee), in the format and pursuant to the process as prescribed by the Administrator, during the Enrollment Period prior to the commencement of the Offering Period to which it relates. If properly completed and timely submitted, the Enrollment Form will become effective for the first Offering Period following submission of the Enrollment Form and all subsequent Offering Periods as provided by <u>Section 5(b)</u> until (i) it is terminated in accordance with <u>Section 11</u>, (ii) it is modified by filing another Enrollment Form in accordance with this <u>Section 5(a)</u> (including an election is made to cease payroll deductions or periodic cash contributions in accordance with <u>Section 6(c)</u>), or (iii) the Participant is otherwise ineligible to participate in the Plan or in a subsequent Offering Period.

(b) <u>Automatic Re-Enrollment</u>. Following the end of each Offering Period, each Participant shall automatically be re-enrolled in the next Offering Period at the applicable rate of payroll deductions or periodic cash contributions in effect on the last Trading Day of the prior Offering Period or otherwise as provided under <u>Section 6</u>, unless (i) the Participant has experienced a Termination of Employment, or (ii) the Participant is otherwise ineligible to participate in the Plan or in the next Offering Period. Notwithstanding the foregoing, the Administrator may require current Participants to complete and submit a new Enrollment Form at any time it deems necessary or desirable to facilitate Plan administration or for any other reason.

5

6. PAYROLL DEDUCTIONS OR PERIODIC CASH CONTRIBUTIONS

(a) <u>Generally</u>. Each Participant's Enrollment Form shall contain a payroll deduction authorization pursuant to which he or she shall elect to have a designated whole percentage of Eligible Compensation between one percent (1%) and ten percent (10%) deducted, on an after-tax basis, on each payday during the Offering Period and credited to the Participant's Account for the purchase of shares of Stock pursuant to the offering. The Administrator shall also have the authority, but not the obligation, to permit a Participant to elect to make periodic cash contributions, in lieu of payroll deductions, for the purchase of shares of Stock pursuant to the offering. Notwithstanding the foregoing, if local law prohibits payroll deductions, a Participant may elect to participate in an Offering Period through contributions to his or her Account in a format and pursuant to a process acceptable to the Administrator. In such event, any such Participant shall be deemed to participate in a separate offering under the Plan, unless the Administrator otherwise expressly provides.

(b) <u>Insufficiency of Contributions</u>. Subject to <u>Section 6(e)</u>, if in any payroll period a Participant has no pay or his or her pay is insufficient (after other authorized deductions) to permit deduction of the full amount of his or her payroll deduction election, then (i) the payroll deduction election for such payroll period shall be reduced to the amount of pay remaining, if any, after all other authorized deductions, and (ii) the percentage or dollar amount of Eligible Compensation shall be deemed to have been reduced by the amount of the reduction in the payroll deduction election for such payroll period. Deductions of the full amount originally elected by the Participant will recommence as soon as his or her pay is sufficient to permit such payroll deductions; *provided*, *however*, no additional amounts shall be deducted to satisfy the Outstanding Election. If the Administrator authorizes a Participant to elect to make periodic cash contributions in lieu of payroll deductions, the failure of a Participant to make any such contributions shall reduce, to the extent of the deficiency in such payments, the number of shares purchasable under the Plan by the Participant.

(c) <u>Cessation after Offering Date</u>. A Participant may cease his or her payroll deductions or periodic cash contributions during an Offering Period by properly completing and timely submitting a new Enrollment Form to the Company (or the Company's designee), in the format and pursuant to the process as prescribed by the Administrator, at any time prior to the last day of such Offering Period (or if an Offering Period has multiple Purchase Periods, the last day of such Purchase Period). Any such cessation in payroll deductions or periodic cash contributions shall be effective as soon as administratively practicable thereafter and shall remain in effect for successive Offering Periods as provided in <u>Section 5(b)</u> unless the Participant submits a new Enrollment Form for a later Offering Period in accordance with <u>Section 5(a)</u>. A Participant may only increase his or her rate of payroll deductions or periodic cash contributions in accordance with <u>Section 6(d)</u>.

(d) <u>Modification Prior to Offering Date</u>. A Participant may increase or decrease his or her rate of payroll deductions or periodic cash contributions, to take effect on the Offering Date of the Offering Period following submission of the Enrollment Form, by properly completing and timely submitting a new Enrollment Form in accordance with <u>Section 5(a)</u>.

(e) <u>Authorized Leave or Disability after Offering Date</u>. Subject to <u>Section 11</u>, if a Participant is absent from work due to an authorized leave of absence or disability (and has not experienced a Termination of Employment), such Participant shall have the right to elect (i) to remain a Participant in the Plan for the then-current Offering Period (or if an Offering Period has multiple Purchase Periods, the then-current Purchase Period) but to cease his or her payroll deductions or periodic cash contributions in accordance with <u>Section 6(c)</u>, or (ii) to remain a Participant in the Plan for the then-current Offering Period (or if an Offering Period has multiple Purchase Periods, the then-current Purchase Period) but to authorize payroll deductions to be made from payments made by the Company or a Participating Affiliate to the Participant during such leave of absence or disability and to undertake to make additional cash payments to the Plan at the end of each payroll period during the Offering Period to the extent that the payroll deductions from payments made by the Company or a Participating Affiliate to such Participant are insufficient to meet such Participant's Outstanding Election. Neither the Company nor a Participating Affiliate shall advance funds to a Participant if the Participant's Account at his or her Outstanding Election.

7. OFFERING PERIODS AND PURCHASE PERIODS; PURCHASE PRICE

(a) The Administrator shall determine from time to time, in its sole discretion, the Offering Periods and Purchase Periods under the Plan. Each Offering Period shall consist of one or more Purchase Periods, as determined by the Administrator. Unless otherwise established by the Administrator prior to the start of an Offering Period, the Plan shall have two (2) Offering Periods (with concurrent Purchase Periods) that commence each calendar year, and each Offering Period shall be of approximately six (6) months' duration, with the first such Offering Period beginning on the first Trading Day of January and ending on the last Trading Day of the immediately following June, and the second such Offering Period beginning on the first Trading Day of July and ending on the last Trading Day of the immediately following December; *provided, however*, that the first Offering Period under the Plan shall commence on the first Trading Day of July following the Effective Date and shall end on the last Trading Day of the immediately following December.

(b) The Administrator shall determine from time to time, in its sole discretion, the Purchase Price of each share of Stock for an Offering Period. Unless otherwise established by the Administrator prior to the start of an Offering Period, the Purchase Price shall be the lesser of eighty-five percent (85%) of the average of the high and low sales price of the Common Stock on the New York Stock Exchange on the Offering Date or the last Trading Day of the Offering Period (or if an Offering Period has multiple Purchase Periods, on the last Trading Day of the Purchase Period).

8. GRANT OF OPTION

(a) <u>Grant of Option</u>. On each Offering Date, each Participant in such Offering Period shall automatically be granted an Option to purchase as many whole or fractional shares of Stock as the Participant will be able to purchase with the payroll deductions or periodic cash contributions credited to the Participant's Account during the applicable Offering Period.

(b) <u>5% Owner Limit</u>. Notwithstanding any provisions of the Plan to the contrary, no Participant shall be granted an Option to purchase shares of Stock under the Plan if such Participant (or any other person whose Stock would be attributed to such Participant pursuant to Code Section 424(d)), immediately after such Option is granted, would own or hold Options to purchase shares of Stock possessing five percent (5%) or more of the total combined voting power or value of all classes of stock of the Company or any of its Subsidiaries.

(c) <u>Other Limitation</u>. The Administrator may determine, as to any Offering Period, that the offering shall not be extended to "highly compensated employees" within the meaning of Code Section 414(q).

9. PURCHASE OF SHARES OF STOCK; PURCHASE LIMITATIONS

(a) <u>Purchase</u>. Unless the Participant's participation in the Plan has otherwise been terminated as provided in <u>Section 11</u>, such Participant will be deemed to have automatically exercised his or her Option to purchase Stock on the last Trading Day of the Offering Period (or if an Offering Period has multiple Purchase Periods, the last Trading Day of the Purchase Period) for the maximum number of shares of Stock that may be purchased at the Purchase Price with the Participant's Account balance at that time; *provided, however*, the number of shares of Stock purchased is subject to adjustment by <u>Section 3</u>, this <u>Section 9</u>, and <u>Section 12</u>. The Administrator shall cause the amount credited to each Participant's Account to be applied to such purchase, and the amount applied to purchase shares of Stock pursuant to an Option shall be deducted from the applicable Participant's Account.

(b) <u>Limit on Number of Shares Purchased</u>. Notwithstanding <u>Section 8(a)</u> or <u>Section 9(a)</u>, in no event may a Participant purchase more than fifty thousand (50,000) shares of Stock in any one Offering Period; *provided*, *however*, that the Administrator may, in its sole discretion, prior to the start of an Offering Period, set a different limit on the number of shares of Stock a Participant may purchase during such Offering Period.

(c) <u>Limit on Value of Shares Purchased</u>. Notwithstanding any provisions of the Plan to the contrary, excluding Options granted pursuant to any Non-423(b) Offering, no Participant shall be granted an Option to purchase shares of Stock under the Plan which permits the Participant's rights to purchase shares under all "employee stock purchase plans" (described in Code Section 423) of the Company and its Subsidiaries to accrue at a rate which exceeds twenty-five thousand dollars (\$25,000) of the Fair Market Value of such shares of Stock (determined at the time such Options are granted) for each calendar year in which such Options are outstanding at any time.

10. STOCK ISSUANCE; SHAREHOLDER RIGHTS; AND SALES OF PLAN SHARES

(a) <u>Stock Issuance and Account Statements</u>. Shares of Stock purchased under the Plan will be held by the Custodian. The Custodian may hold the shares of Stock purchased under the Plan by book entry or in the form of stock certificates in nominee names and may commingle shares held in its custody in a single account without identification as to individual Participants. The Company shall cause the Custodian to deliver to each Participant a statement for each Offering Period during which the Participant purchases Stock under the Plan, which statement shall reflect, for each such Participant, (i) the amount of payroll deductions withheld or periodic cash contributions made during the Offering Period, (ii) the number of shares of Stock purchased, (iii) the Purchase Price of the shares of Stock purchased, and (iv) the total number of shares of Stock held by the Custodian for the Participant as of the end of the Offering Period.

(b) <u>Shareholder Rights</u>. A Participant shall not be a shareholder or have any rights as a shareholder with respect to shares of Stock subject to the Participant's Options under the Plan until the shares of Stock are purchased pursuant to the Options and such shares of Stock are transferred into the Participant's name on the Company's books and records. No adjustment will be made for dividends or other rights for which the record date is prior to such time. Following purchase of shares of Stock under the Plan and transfer of such shares of Stock into the Participant's name on the Company's books and records, a Participant shall become a shareholder with respect to the shares of Stock purchased during such Offering Period (or, if applicable, Purchase Period) and, except as otherwise provided in <u>Section 10(c)</u>, shall thereupon have all dividend, voting, and other ownership rights incident thereto.

(c) <u>Sales of Plan Shares</u>. The Administrator shall have the right to require any or all of the following with respect to shares of Stock purchased under the Plan:

(i) that a Participant may not request that all or part of the shares of Stock be reissued in the Participant's own name and shares be delivered to the Participant until two (2) years (or such shorter period of time as the Administrator may designate) have elapsed since the Offering Date of the Offering Period in which the shares were purchased and one (1) year has elapsed since the day the shares were purchased (the "Holding Period");

(ii) that all sales of shares of Stock during the Holding Period applicable to such purchased shares be performed through a licensed broker acceptable to the Company; and

(iii) that Participants abstain from selling or otherwise transferring shares of Stock purchased pursuant to the Plan for a period lasting up to two (2) years from the date the shares of Stock were purchased pursuant to the Plan.

11. DEEMED CANCELLATION OR TERMINATION OF PARTICIPATION

(a) <u>Termination of Employment Other than Death</u>. In the event a Participant who holds outstanding Options to purchase shares of Stock under the Plan experiences a Termination of Employment for any reason other than death prior to the last Trading Day of the Offering Period, the Participant's outstanding Options to purchase shares of Stock under the Plan shall automatically terminate, and the Administrator shall refund in cash the Participant's Account balance as soon as practicable thereafter.

(b) <u>Death</u>. In the event of the death of a Participant while the Participant holds outstanding Options to purchase shares of Stock under the Plan, the legal representatives of such Participant's estate (or, if the Administrator permits a beneficiary designation, the beneficiary or beneficiaries most recently designated by the Participant prior to his or her death) may, within three (3) months after the Participant's death (but no later than the last Trading Day of the Offering Period (or if an Offering Period has multiple Purchase Periods, the last Trading Day of the then-current Purchase Period)) by written notice to the Company (or the Company's designee), elect one of the following alternatives. In the event the Participant's legal representatives (or, if applicable, beneficiary or beneficiaries) fail to deliver such written notice to the Company (or the Company's designee) within the prescribed period, the alternative in <u>Section 11(b)(ii)</u> shall apply.

(i) The Participant's outstanding Options shall be reduced to the number of shares of Stock that may be purchased, as of the last day of the Offering Period (or if an Offering Period has multiple Purchase Periods, the last Trading Day of the then-current Purchase Period), with the amount then credited to the Participant's Account; or

(ii) The Participant's Options to purchase shares of Stock under the Plan shall automatically terminate, and the Administrator shall refund in cash, to the Participant's legal representatives, the Participant's Account balance as soon as practicable thereafter.

(c) <u>Other Termination of Participation</u>. If a Participant ceases to be eligible to participate in the Plan for any reason, the Administrator shall refund in cash the affected Participant's Account balance as soon as practicable thereafter. Once terminated, participation may not be reinstated for the then-current Offering Period, but, if otherwise eligible, the Eligible Employee may elect to participate in a subsequent Offering Period in accordance with <u>Section 5</u>.

12. CHANGES IN CAPITALIZATION

(a) <u>Changes in Stock</u>. If the number of outstanding shares of Stock is increased or decreased or the shares of Stock are changed into or exchanged for a different number or kind of shares or other securities of the Company by reason of any recapitalization, reclassification, stock split, reverse stock split, spin-off, combination of shares, exchange of shares, stock dividend, or other distribution payable in capital stock, or other increase or decrease in such shares effected without receipt of consideration by the Company occurring after the Effective Date, the number and kinds of shares that may be purchased under the Plan shall be adjusted proportionately and accordingly by the Company. In addition, the number and kind of shares for which Options are outstanding shall be similarly adjusted so that the proportionate interest of a Participant immediately following such event shall, to the extent practicable, be the same as immediately prior to such event. Any such adjustment in outstanding Options shall not change the aggregate Purchase Price payable by a Participant with respect to shares subject to such Options but shall include a corresponding proportionate adjustment in the Purchase Price per share. Notwithstanding the foregoing, in the event of a spin-off that results in no change in the number of outstanding shares of Stock, the Company may, in such manner as the Company deems appropriate, adjust (i) the number and kind of shares for which Options are outstanding under the Plan and (ii) the Purchase Price per share.

(b) <u>Reorganization in Which the Company Is the Surviving Corporation</u>. Subject to <u>Section 12(c)</u>, if the Company shall be the surviving corporation in any reorganization, merger, or consolidation of the Company with one or more other corporations, all outstanding Options under the Plan shall pertain to and apply to the securities to which a holder of the number of shares of Stock subject to such Options would have been entitled immediately following such reorganization, merger, or consolidation, with a corresponding proportionate adjustment of the Purchase Price per share so that the aggregate Purchase Price thereafter shall be the same as the aggregate Purchase Price of the shares subject to such Options immediately prior to such reorganization, merger, or consolidation.

(c) <u>Reorganization in Which the Company Is Not the Surviving Corporation, Sale of Assets or Stock, and Other Corporate Transactions</u>. Upon any dissolution or liquidation of the Company, or upon a merger,

consolidation, or reorganization of the Company with one or more other corporations in which the Company is not the surviving corporation, or upon a sale of all or substantially all of the assets of the Company to another corporation, or upon any transaction (including, without limitation, a merger, consolidation, or reorganization in which the Company is the surviving corporation) approved by the Board that results in any person or entity owning more than fifty percent (50%) of the combined voting power of all classes of stock of the Company, the Plan and all Options outstanding hereunder shall terminate, except to the extent provision is made in writing in connection with such transaction for the continuation of the Plan and/or the assumption of the Options theretofore granted, or for the substitution for such Option of new rights covering the stock of a successor corporation, or a parent or subsidiary thereof, with appropriate adjustments as to the number and kinds of shares and purchase prices, in which event the Plan and rights theretofore granted shall continue in the manner and under the terms so provided. In the event of any such termination of the Plan, the Offering Period shall be deemed to have ended on the last Trading Day prior to such termination, and in accordance with <u>Section 9</u>, the Options of each Participant then outstanding shall be deemed to be automatically exercised on such last Trading Day. The Administrator shall send written notice of an event that will result in such a termination to all Participants at least five (5) days prior to the date upon which the Plan will be terminated.

(d) <u>Adjustments</u>. Adjustments under this <u>Section 12</u> related to stock or securities of the Company shall be made by the Administrator, whose determination in that respect shall be final, binding, and conclusive.

(e) <u>No Limitations on Company</u>. The grant of an Option pursuant to the Plan shall not affect or limit in any way the right or power of the Company to make adjustments, reclassifications, reorganizations, or changes of its capital or business structure or to merge, consolidate, dissolve or liquidate, or to sell or transfer all or any part of its business or assets.

13. TERM; AMENDMENT, SUSPENSION, AND TERMINATION OF THE PLAN

(a) <u>Term</u>. The Plan shall be effective as of the Effective Date. The outstanding offering period under the Prior Plan shall be the last offering period under the Prior Plan which ends on June 30, 2016. The Plan shall terminate on the first to occur of (i) the day before the tenth (10^{th}) anniversary of the date of adoption of the Plan by the Board, (ii) the date on which all shares of Stock reserved for issuance under the Plan pursuant to <u>Section 3</u> have been issued, (iii) the date determined in accordance with <u>Section 12</u>, and (iv) the date determined in accordance with <u>Section 13(b)</u>.

(b) <u>Amendment, Suspension, and Termination of the Plan</u>. The Administrator may, at any time and from time to time, amend, suspend, or terminate the Plan or an Offering Period under the Plan; *provided*, *however*, that no amendment, suspension, or termination shall, without the consent of the Participant, impair any rights of a Participant that have vested at the time of such amendment, suspension, or termination. Without approval of the shareholders of the Company, no amendment shall be made (i) increasing the number of shares reserved for issuance under the Plan pursuant to <u>Section 3</u> (except as provided in <u>Section 12</u>) or (ii) changing the eligibility requirements for participating in the Plan.

14. GENERAL PROVISIONS

(a) <u>Withholding of Taxes</u>. To the extent that a Participant recognizes ordinary income in connection with a sale or other transfer of any shares of Stock purchased under the Plan, the Company may withhold amounts needed to cover such taxes from any payments otherwise due and owing to the Participant or from shares that would otherwise be issued to the Participant under the Plan.

(b) <u>Options Not Transferable or Assignable</u>. A Participant's Options under the Plan may not be sold, pledged, assigned, or transferred in any manner, whether voluntarily, by operation of law, or otherwise. If a Participant sells, pledges, assigns, or transfers his or her Options in violation of this <u>Section 14(b)</u>, such Options shall immediately terminate, and the Participant shall immediately receive a refund of the amount then credited to the Participant's Account. Any payment of cash or issuance of shares of Stock under the

Plan may be made only to the Participant (or, in the event of the Participant's death, to the Participant's estate or, if the Administrator permits a beneficiary designation, the beneficiary or beneficiaries most recently designated by the Participant prior to his or her death). During a Participant's lifetime, only such Participant may exercise his or her Options under the Plan.

(c) <u>No Right to Continued Employment</u>. Neither the Plan nor any Option to purchase Stock under the Plan confers upon any Eligible Employee or Participant any right to continued employment with the Company or any of its Subsidiaries, nor will a Participant's participation in the Plan restrict or interfere in any way with the right of the Company or any of its Subsidiaries to terminate the Participant's employment at any time.

(d) <u>No Interest on Payments</u>. No interest shall be paid on sums withheld from a Participant's pay or otherwise contributed for the purchase of shares of Stock under the Plan unless otherwise determined necessary by the Administrator.

(e) <u>Governmental Regulation</u>. The Company's obligation to issue, sell, and deliver shares of Stock pursuant to the Plan is subject to such approval of any governmental authority and any national securities exchange or other market quotation system as may be required in connection with the authorization, issuance, or sale of such shares.

(f) <u>Rule 16b-3</u>. Transactions under this Plan are intended to comply with all applicable conditions of Rule 16b-3 or any successor provision under the Securities Exchange Act of 1934, as amended. If any provision of the Plan or action by the Administrator fails to so comply, it shall be deemed null and void to the extent permitted by applicable law and deemed advisable by the Board. Moreover, in the event the Plan does not include a provision required by Rule 16b-3 to be stated in the Plan, such provision (other than one relating to eligibility requirements or the price and amount of awards) shall be deemed automatically to be incorporated by reference into the Plan.

(g) Payment of Plan Expenses. The Company shall bear all costs of administering and carrying out the Plan.

(h) <u>Application of Funds</u>. All funds received or held by the Company under the Plan may be used for any corporate purpose until applied to the purchase of Stock and/or refunded to Participants.

(i) <u>Governing Law</u>. The validity and construction of the Plan and the Options granted hereunder shall be governed by, and construed and interpreted in accordance with, the laws of the State of Delaware (other than any conflicts or choice of law rule or principle that might otherwise refer construction or interpretation of the Plan and the Options granted under the Plan to the substantive laws of any other jurisdiction), except to the extent superseded by applicable U.S. federal laws.

* * *

Labcorp Employee Stock Purchase Plan Canadian sub-plan

SUB-PLAN FOR CANADIAN PARTICIPANTS

1. APPLICATION

This <u>Sub-Plan</u> sets forth additional terms and conditions applicable to the rights granted to, and the shares of Stock purchased by, Eligible Employees under the Plan who are, or are deemed to be, resident in Canada for the purpose of payment of income taxes or are otherwise liable to tax under the Income Tax Act (Canada) in respect their participation in the Plan (a "**Canadian Participant**").

The Plan and this Sub-Plan are complimentary to each other and shall be deemed as one. In any case of contradiction between the provisions of this Sub-Plan and the Plan, the provisions set out in the Sub-Plan shall prevail. Any capitalized terms used in this Sub-Plan but not defined shall have the meaning given to those terms in the Plan.

2. GLOBAL PROVISIONS

(a) <u>English Language</u>. By participating in the Plan, each Canadian Participant acknowledges that such Canadian Participant is proficient in the English language, or has consulted with an advisor who is sufficiently proficient in English, so as to allow the Canadian Participant to understand the terms and conditions of the Plan, this Sub-Plan and any other related document under the Plan. If a Canadian Participant has received the Plan, the Sub-Plan or any other document related to the Plan translated into a language other than English and if the meaning of the translated version is different than the English version, the English version will control.

(c) <u>Currency</u>. Each Canadian Participant understands that, if the Canadian Participant's payroll deductions under the Plan are made in any currency other than U.S. dollars, such contributions will be converted to U.S. dollars on or prior to the date shares of Stock are purchased under the Plan using a prevailing exchange rate in effect at the time such conversion is performed, as determined by the Company. Each Canadian Participant understands and agrees that neither the Company or its Participating Affiliates shall be liable for any foreign exchange rate fluctuation between the Canadian Participant's Canadian dollar and the U.S. dollar that may affect the value of the purchase rights granted to the Canadian Participant under the Plan, or of any amounts due to the Canadian Participant under the Plan or as a result of the subsequent sale of any shares of Stock acquired under the Plan.

(d) <u>Acknowledgment of Nature of Plan and Rights</u>. In participating in the Plan and this Sub-Plan, each Canadian Participant acknowledges that:

(i) for employment and labor law purposes, except if required by the applicable employment standards legislation, the rights granted and the shares of Stock purchased under the Plan are an extraordinary item that do not constitute wages or salary of any kind for services of any kind rendered to the Company or its Participating Affiliates, and the award of rights is outside the scope of the Canadian Participant's employment or service contract, if any;

(ii) for employment and labor law purposes, except if required by the applicable employment standards legislation, the rights granted and the shares of Stock purchased under the Plan are not part of integral, normal or expected wages or salary for any purposes, including, but not limited to, calculation of any notice of termination of employment, payment in lieu of any notice of termination of employment, severance, resignation, termination (with or without Cause, as defined below), redundancy, expiry of fixed-term contract, dismissal, end of service payments, bonuses, holiday pay, paid time off,

long-service awards, pension or retirement benefits or similar payments and in no event should be considered as compensation for, or relating in any way to, past services for the Company or its Participating Affiliates;

(iii) the rights and the shares of Stock purchased under the Plan are not intended to be an integral component of compensation or to replace any pension rights or compensation;

(iv) neither the rights nor any provision of Plan or the policies adopted pursuant to the Plan confer upon any Canadian Participant any right with respect to service or continuation of current service and shall not be interpreted to form an employment or a service contract or relationship with the Company or its Participating Affiliates;

(v) the future value of the underlying shares of Stock is unknown and cannot be predicted with certainty;

(vi) if the underlying shares of Stock do not increase in value, the right may have no value;

(vii) if a Canadian Participant acquires shares of Stock, the value of such shares acquired upon purchase may increase or decrease in value, even below the original price paid; and

(viii) participation in the Plan by a Canadian Participant is purely voluntary and the Canadian Participant has not been induced to participate by expectation of engagement, appointment, employment or continued engagement, appointment or employment, as applicable.

(e) <u>Applicable Withholding Taxes</u>. To the extent that a Canadian Participant realizes employment income in connection with the purchase of shares of Stock under the Plan, the Company or its Participating Affiliates may withhold all taxes and other source deductions or other amounts which the Company or its Participating Affiliates are required by law to withhold from any payments otherwise due and owing to the Canadian Participant or from shares that would otherwise be issued to the Canadian Participant under the Plan.

3. LANGUAGE CONSENT

The parties acknowledge that it is their express wish that the Plan, as well as all documents, notices and legal proceedings entered into, given or instituted pursuant hereto or relating directly or indirectly hereto, be drawn up in English.

Consentement relatif à la langue utilisée. Les parties reconnaissent avoir exigé la rédaction en anglais de cette convention, ainsi que de tous documents, avis et procédures judiciaires, exécutés, donnés ou intentés en vertu de, ou liés directement ou indirectement à, la présente convention.

4. TERMINATION OF EMPLOYMENT

In the event of termination of a Canadian Participant's employment, the Canadian Participant's right to purchase shares of Stock under the Plan, if any, will terminate, and he or she shall be deemed to have elected to withdraw from the Plan, effective as of the Canadian Participant's Termination Date.

"Cause" shall mean, in respect of a Canadian Participant: (i) in respect of the termination of an employee employed in Ontario, wilful misconduct, disobedience or wilful neglect of duty by the employee that is not trivial and is not condoned by the employee's employer; and (ii) in respect of an employee employed in another jurisdiction outside of Ontario, such conduct by the employee which permits the

employee's employer to terminate the employee without notice, payment in lieu of notice or severance pay, whether arising under statute, contract or at law.

"Termination Date" shall mean in respect of a Canadian Participant whose employment or term of office with a Participating Affiliate terminates for any reason, including by reason of retirement, expiry of fixed-term contract, resignation, death, disability, termination without Cause, termination for Cause, the last day of the Canadian Participant's actual and active employment or term of office with a Participating Affiliate, which in the event of a termination without Cause or termination due to disability shall include any minimum statutory period of individual notice of termination or pay in lieu to the extent required by the applicable employment standards legislation, but shall exclude any other period of deemed employment as well as additional notice or severance periods or pay in lieu in respect of which the Canadian Participant is in receipt of or may be eligible to receive at statute, common law or civil law, pursuant to a contract, or otherwise. For greater certainty, (a) a Termination Date shall be determined without reference to any statutory severance, notice of mass lay-offs or any contractual or common law or civil law notice of termination or pay in lieu that the Canadian Participant is entitled to or in receipt of; and (b) in no event will the Canadian Participant receive less than that the entitlements required by applicable minimum employment standards legislation.

For absolutely certainty, each Canadian Participant represents, warrants and acknowledges that such Canadian Participant has read and understood the terms and conditions of the Plan which: (i) state that the Canadian Participant shall have no entitlement to damages or other compensation whatsoever arising from, in lieu of, or related to not receiving any compensation, Options or shares of Stock which would have been granted or payable after the Canadian Participant's Termination Date, including but not limited to damages in lieu of notice at common law or civil law; and (ii) have the effect that no period of common law or civil law reasonable notice that exceeds the Canadian Participant's minimum statutory individual notice period under applicable employment standards legislation (if any), shall be used for the purposes of calculating the Canadian Participant's entitlement under this Plan. By participating in this Plan, each Canadian Participant expressly waives any eligibility or entitlement to receive damages or payment in lieu of any forfeited amounts under this Plan that would have vested, accrued or been paid during any common law or civil law reasonable notice period that exceeds such Canadian Participant's minimum statutory individual notice period under the applicable employment standards legislation (if any).

The payroll deductions credited to such Canadian Participant's account during the Offering Period shall be paid to such Canadian Participant or, in the case of his or her death, to the executor, administrator or liquidator of the estate of the Canadian Participant, as soon as reasonably practicable and such Canadian Participant's rights for the Offering Period shall be automatically terminated on the Termination Date.

5. AUTHORIZED LEAVE

"An Authorized Leave of Absence and Disability" in respect of a Canadian Participant shall mean : (i) a paid leave of absence, approved by the Company or its Participating Affiliates and paid through the Company or a Participating Affiliate's payroll, including, for greater certainty, a leave during which the Canadian Participant is in receipt of short-term disability benefits; or (ii) an unpaid leave of absence taken in accordance with applicable employment standards legislation during which the applicable legislation requires that the Canadian Participant be permitted to elect to continue participation in the Plan during the leave.

6. PURCHASE OF SHARES

Notwithstanding anything in the Plan to the contrary, the purchase of Shares for any Canadian Participant shall occur on the first Trading Day following the end of the Offering Period, and the Purchase Price shall be calculated by reference to the sales price of the Common Stock on the New York Stock Exchange on such date.

7. DATA PROTECTION

The Company collects and processes various types of information that is used to administer or support the Plan. "*Personal Information*" means information that can be used to identify or authenticate an individual but does not include business contact information and publicly available information.

In addition to the global provisions of the Sub-Plan, each Canadian Participant hereby authorizes the Company and the Company's representatives to discuss with and obtain all relevant Personal Information from all personnel, professional or not, involved in the administration and operation of the Plan, where necessary or inadvertent, including personal biographical information (including an Eligible Employee's name, address, gender and date of birth), tax reporting information (including a Social Insurance Number and citizenship information), as well as contact information. Each Canadian Participant further authorizes the Company to disclose and discuss the Plan and this Sub-Plan with their advisors, to the extent reasonably necessary to administer the Plan and this Sub-Plan, including in relation to audits and communication of the Plan. Each Canadian Participant further authorizes the Company to record Personal Information and Plan information, and to keep such information in the Eligible Employee's employee file.

The Company affirms its commitment to ensure that all Personal Information of Canadian Participants collected, maintained and used, is kept confidential and used only for the purposes for which it is intended, and assumes responsibility for safeguarding such Personal Information in accordance with the Plan and Sub-Plan requirements and all applicable laws.

In the event of a security breach, the Company will take reasonable steps to comply with all applicable breach notification processes in accordance with applicable law. A security breach occurs when the security or confidentiality of Personal Information is comprised, and includes the unauthorized collection, use, or disclosure of Personal Information.

The measures that the Company will undertake to safeguard the security of Personal Information collected include, but are not necessarily limited to, taking the following steps commensurate with industry standards, as applicable: (i) limiting employee and contractor access to Personal Information; (ii) securing business facilities, data centers, paper files, services back-up systems and computing equipment; (iii) implementing network, device, database, and platform security in accordance with industry standards; (iv) securing information transmission, storage and disposal; (v) implementing appropriate personnel security and integrity procedures sand practices; and (vi) providing appropriate privacy and information security training to employees.

The administration of the Plan might entail storage of Personal Information outside of Canada, including, without limitation, in the following countries: United States of America. Canadian Participants will be clearly informed of such storage outside Canada and any changes thereto, and be provided with the contact information of an individual who can answer questions regarding the collection and use of Personal Information.

8. NOTIFICATIONS

(a) <u>Securities Law Information</u>. There may be securities law implications to a Canadian Participant who sells shares of Stock acquired through the Plan through a broker other than a broker appointed under the Plan or if the sale does not take place through the facilities of a stock exchange outside of

Canada on which the shares of Stock are listed and the Canadian Participant shall be solely responsible for complying with any such securities laws, if applicable.

(b) <u>Foreign Asset/Account Reporting Information</u>. If a Canadian Participant is a Canadian resident, such Canadian Participant may be required to report his or her foreign property on form T1135 (Foreign Income Verification Statement) if the total cost of the foreign property exceeds a certain threshold at any time during the year. Shares of Stock and rights to receive shares of Stock (e.g., Options) under the Plan are property which are included in foreign property that must be reported. Such Options must be reported - generally at a nil cost - if the threshold is exceeded because of other foreign property. When shares of Stock are acquired, their cost generally is the adjusted cost base ("ACB") of the shares. The ACB would ordinarily equal the fair market value of the shares of Stock at the time of purchase, but if other shares of Stock are also owned, this ACB may have to be averaged with the ACB of the other shares of Stock. The form must be filed by April 30th of the following year.

```
A-5
```

Labcorp Employee Stock Purchase Plan

UK sub-plan

Sub-plan for United Kingdom participants

1. Application and interpretation

- 1.1 This <u>Sub-Plan</u> sets forth additional terms and conditions applicable to the rights granted to, and the shares of Stock purchased by, Eligible Employees (as such term shall apply for potential participants in the United Kingdom for the purposes of this Sub-Plan) under the Plan:
 - (a) who are, or are deemed to be, resident in the United Kingdom for the purpose of payment of income taxes;
 - (b) who are otherwise liable to tax in the United Kingdom in respect of their participation in the Plan; and/or
 - (c) to whom the Administrator (in its absolute discretion) designates this Sub-Plan shall apply (including, without limitation, as a result of their employment being subject to the laws of the United Kingdom),

each, a "UK Participant".

- 1.2 The Plan and this Sub-Plan are complimentary to each other and the terms of the Plan shall apply to this Sub-Plan unless otherwise stated. In any case of contradiction between the provisions of this Sub-Plan and the Plan, the provisions set out in the Sub-Plan shall prevail. Any capitalized terms used in this Sub-Plan but not otherwise defined in this Sub-Plan shall have the meaning given to those terms in the Plan.
- 1.3 In this Sub-Plan:
 - (a) "Employee Tax" means income tax, employee's National Insurance contributions and any other employee's social security contributions (whether or not in force at the date of the UK Participant becoming a participant in the Plan and this Sub-Plan, including the employee's portion of any UK Health and Social Care Levy), together with any similar or equivalent taxes, contributions, withholdings or levies in the UK or any other jurisdiction, or any other sums for which the relevant employer is required to account to HM Revenue & Customs or any other tax or social security authority on behalf of the relevant UK Participant, and any interest or penalties relating to any of them;
 - (b) "ITEPA 2003" means the Income Tax (Earnings and Pensions) Act 2003;
 - (c) "Pounds Sterling" shall mean the lawful currency of the United Kingdom.

2. Eligible Employees

For the purposes of this Sub-Plan, and noting that an offer to a UK Participant would be a Non-423(b) Offering, the definition of "Eligible Employee" in paragraph 2(i) of the Plan shall be amended by way of the deletion of sub-paragraphs 2(i)(i) and 2(i)(ii).

B-1

3. Important risk and currency information

- 3.1 Each UK Participant understands and acknowledges that the value of any Shares acquired pursuant to the Plan and this Sub-Plan can go up or down, and that a UK Participant could get back less than the original amount of their payroll deductions (or any other amount invested in Shares) and be exposed to a capital loss; and
- 3.2 Each UK Participant understands and acknowledges that, if the UK Participant's payroll deductions under the Plan and this Sub-Plan are made in any currency other than U.S. dollars, such contributions will be converted to U.S. dollars on or prior to the date shares of Stock are purchased under the Plan and this Sub-Plan using a prevailing exchange rate in effect at the time such conversion is performed, as determined by the Company. Each UK Participant understands and agrees that neither the Company or its Participating Affiliates shall be liable for any foreign exchange rate fluctuation between Pounds Sterling (or any other relevant currency) and the U.S. dollar that may affect the value of the purchase rights granted to the UK Participant under the Plan and this Sub-Plan, or of any amounts due to the UK Participant under the Plan and this Sub-Plan.

4. Relationship to employment

- 4.1 The rights and obligations of an UK Participant in relation to his or her office or employment with the Company or any Participating Affiliate (including but not limited to under any employment contract or service contract) shall not be affected by this Sub-Plan, the Plan, or any Enrollment Form or other documentation or agreements in relation to this Sub-Plan or the Plan. If on termination of employment or engagement (lawfully or otherwise) any UK Participant loses any rights or benefits under this Sub-Plan or the Plan (including rights or benefits that he would not have lost had the employment or engagement not been terminated), the UK Participant is not entitled to any compensation for such loss of rights."
- 4.2 Rule 14(c) of the Plan shall be amended as follows in relation to this Sub-Plan and UK Participants:

None of the Plan, this Sub-Plan or any Option to purchase Stock under the Plan or this Sub-Plan confers upon any Eligible Employee or UK Participant any right to continued employment with the Company, any Subsidiary or any Participating Affiliate, and a UK Participant's participation in the Plan or this Sub-Plan shall not restrict or interfere in any way with the right of the Company, any Subsidiary or any Participating Affiliate to terminate the UK Participant's employment in accordance with applicable law.

5. Tax

- 5.1 As a condition of participation in the Plan and this Sub-Plan, the Administrator may require a UK Participant to enter into any election for tax purposes that the Administrator (in its absolute discretion) deems necessary and/or desirable, including (without limitation) an election under section 431(1) of ITEPA 2003 in circumstances where a Holding Period applies under <u>Section 10(c)(i)</u> of the Plan.
- 5.2 Notwithstanding any other provision of the Plan, a UK Participant's participation in the Plan and this Sub-Plan (including the purchase of Stock under the Plan and this Sub-Plan) shall be conditional upon such UK Participant making adequate provision for payment of an amount equal to any Employee Tax for which the Company, any Subsidiary or any Participating Affiliate is liable to account or pay in connection with the UK Participant's participation in the Plan and this Sub-

B-2

Plan (including the purchase of Stock under the Plan and this Sub-Plan) (all such liabilities, together, the "Relevant Tax Liabilities").

- 5.3 Each UK Participant shall indemnify (on an after-tax basis) the Company, each Subsidiary and each Participating Affiliate (each, an "Indemnified Party") for any Employee Tax which the Indemnified Party may be liable to deduct, withhold, account for or pay by reason of the UK Participant's participation in the Plan and this Sub-Plan (including the UK Participant's purchase of Stock).
- 5.4 Notwithstanding any other provision of the Plan, the Administrator may determine, and each UK Participant agrees that the Administrator may determine, that the UK Participant shall satisfy the Relevant Tax Liabilities in any way determined by the Administrator in the Administrator's absolute discretion (but subject at all times to the requirements of applicable law), including (without limitation) the Company, any Subsidiary or any Participating Affiliate:
 - (a) withholding a number of Stock having an aggregate fair market value equal to such Relevant Tax Liabilities;
 - (b) withholding cash from the UK Participant's aggregate payroll deductions under the Plan and this Sub-Plan equal to such Relevant Tax Liabilities (and, for the avoidance of doubt, such cash need not be applied to the purchase of Stock and the UK Participant shall have no right to receive or be paid such cash);
 - (c) withholding an amount equal to such Relevant Tax Liabilities from the UK Participant's salary and/or any other amounts payable to the UK Participant (including in their capacity as an employee) by the Company, any Subsidiary or any Participating Affiliate; and/or
 - (d) requiring the UK Participant to account directly to the Company, any Subsidiary or any Participating Affiliate for an amount equal to the Relevant Tax Liabilities on a timely basis.

6. Data protection

The Company will process each UK Participant's personal data in connection with the Plan and this Sub-Plan in accordance with the terms of the privacy notice previously provided to the UK Participant.

B-3